

# Integrated Pollution Prevention and Control (IPPC)

# Interim Guidance for the Ferrous Foundries Sector







ENVIRONMENT AGENCY Commissioning Organisation Environment Agency Rio House Waterside Drive Aztec West Almondsbury Bristol BS32 4UD

Tel 01454 624400 Fax 01454 624409

© Environment Agency

First Published 2001

ISBN

### This document is Environment Agency copyright © We specifically allow the following:

• Internal business or personal use. You may use this document for your own private use or for use within your business without restriction.

• Giving copies to others. You may do this without restriction provided that you make no charge.

If you wish to use this document in any way other than as set out above including in particular for commercial gain, for example by way of rental, licence, sale or providing services you should contact:

Liz Greenland Environment Agency Scientific and Technical Information Service 2440 The Quadrant Aztec West Almondsbury Bristol BS32 4AQ

Version	Date	Change
1	May 2001	DRAFT: for internal consultation
1	June 2001	External consultation draft

### **Record of changes**

Note:

Queries about the content of the document should be made to Alison Simmons (0117 914 2778) or any member of the IPPC Project or Technical Guidance Teams.

Written comments or suggested improvements should be sent to Graham Winter at the Environment Agency Technical Guidance Section by email at graham.winter@environment-agency.gov.uk or at:

Environmental Protection National Service Environment Agency Block 1 Government Building Burghill Road Westbury-on-Trym Bristol. BS10 6BF Telephone 0117 914 2868

# **Executive summary**

This guidance has been produced by the Environment Agency for England and Wales in collaboration with the Scottish Environment Protection Agency (SEPA) and the Northern Ireland Environment and Heritage Service (EHS). Together these are referred to as "the Regulator" in this document. Its publication follows consultation with industry, government departments and non-governmental organisations.

*What is IPPC* Integrated Pollution Prevention and Control (IPPC) is a regulatory system that employs an integrated approach to control the environmental impacts of certain industrial activities. It involves determining the appropriate controls for industry to protect the environment through a single permitting process. To gain a Permit, Operators will have to show that they have systematically developed proposals to apply the 'Best Available Techniques' (BAT) and meet certain other requirements, taking account of relevant local factors.

The Regulators intend to implement IPPC to:

- protect the environment as a whole;
- promote the use of "clean technology" to minimise waste at source;
- encourage innovation, by leaving significant responsibility for developing satisfactory solutions to environmental issues with industrial Operators; and
- provide a "one-stop shop" for administering applications for Permits to operate.

Once a Permit has been issued, other parts of IPPC come into play. These include compliance monitoring, periodic Permit reviews, variation of Permit conditions and transfers of Permits between Operators. IPPC also provides for the restoration of industrial sites when the permitted activities cease to operate.

*This Guidance and the BREF based on the BAT Reference document BREF (see Ref. 1) produced by the European Commission once it becomes available. The BREF will be the result of an exchange of information between member states and industry. The BREF for this sector is not available at the time of writing this guidance document and it is planned that this document will be reviewed and revised as appropriate once the BREF becomes available.* 

The aims of this Guidance are to:

The aims of this Guidance

- provide a clear structure and methodology which Operators making an application should follow to
  ensure that all aspects of the PPC Regulations (see Appendix 2 for equivalent legislation in
  Scotland and Northern Ireland) and other relevant Regulations have been addressed (see Section
  1.2), and it should thereby assist the Operator to make a satisfactory application;
- minimise the effort by both Operator and Regulator in the permitting of an installation by use of clear indicative standards and the use of material from previous applications and from accredited Environmental Management Systems (EMSs);
- improve the consistency of applications by ensuring that all relevant issues are addressed;
- increase the transparency of the permitting process by having a structure in which the Operator's
  response to each issue, and any departures from the standards, can be seen clearly;
- improve consistency of regulation across installations and sectors by facilitating the comparison of applications;
- provide a summary of the BAT techniques for pollution control from the BREF and UK experience which are relevant in the UK context expressed, where possible, as clear indicative standards and which need to be addressed by Applicants;
- provide an arrangement of information which allows the reader to find, quickly all of the guidance associated with:
  - -a subject (e.g. accidents, energy or noise) (Sections 2.1 and 2.5 2.11);
  - -the technical areas (e.g. casting or sand reclamation) (Sections 2.3 2.4);
  - -particular emissions (e.g. VOCs or particulates ) (Section 3).

Additionally, to assist Operators in making applications, separate, horizontal guidance is available on a range of topics such as waste minimisation, monitoring, calculating stack heights etc. The majority of this guidance is available free through the Environment Agency, SEPA or EHS (Northern Ireland) web sites (see References).

# CONTENTS

1	INTR	ODUCTIO	ON	1
	1.1	UNDERS	STANDING IPPC AND BAT	1
	1.2	Making	AN APPLICATION	3
	1.3	NSTALL	ATIONS COVERED	4
	1.4	Review	PERIODS	5
	1.5	KEY ISS	UES FOR THIS SECTOR	6
	1.6	SUMMAF	RY OF RELEASES FOR EACH SECTOR	
	1.7	OVERVIE	EW OF THE ACTIVITIES IN THIS SECTOR	
	1.8	ECONON	MIC ASPECTS FOR THIS SECTOR	10
	1.9	SITE-SP	CIERCISSUES	10
2	TECH		FOR POLLUTION CONTROL	11
	21	Manage	EMENT TECHNIQUES	12
	2.2	MATERI	ALS INPUTS	15
		2.2.1	Raw materials selection	15
		2.2.2	Waste minimisation (minimising the use of raw materials)	17
		2.2.3	Water use	18
	2.3	THE MA	IN ACTIVITIES AND ABATEMENT	20
		2.3.1	Storage and handling of raw materials	21
		2.3.2	Launders	22
		2.3.3	Desulphurisation of molten iron in ladies	22
		2.3.4	Proportion of SG Iron in ladies	23
		2.3.5	Casting pouring or moulding	24 27
		2.3.0	Knockout and reclamation	27 29
		2.3.8	Sand reclamation	
		2.3.9	Fettlina, dressina or finishina of castinas	31
		2.3.10	Waste handling and recycling facilities	32
		2.3.11	Control of point source emissions to air	32
		2.3.12	Abatement of point source emissions to surface water and sewer	33
		2.3.13	Control of <u>fugitive</u> emissions to air	36
		2.3.14	Control of <u>fugitive</u> emissions to surface water, sewer and groundwater	37
	<b>0</b> 4	2.3.15	Odour	38
	2.4		INS TO GROUNDWATER	39
	2.5	VVASTE		41
	2.0	FNERCY	/	42 11
	2.1	271	Basic energy requirements (1)	44
		2.7.2	Basic energy requirements (2)	
		2.7.3	Further energy efficiency requirements	47
	2.8	ACCIDE	NTS AND THEIR CONSEQUENCES	49
	2.9	NOISE A	ND VIBRATION	52
	2.10	Μονιτο	RING	54
	2.11	DE-COM	IMISSIONING	58
	2.12	INSTALL	ATION WIDE ISSUES	60
3	EMIS	SION BE	NCHMARKS	61
	3.1	EMISSIO	INVENTORY AND BENCHMARK COMPARISON	61
	3.2	THE EM	ISSION BENCHMARKS	62
		3.2.1	Emissions to air associated with the use of BAT	62
		3.2.2	Emissions to water associated with the use of BA1	62
		3.2.3	Linits for benchmarks and setting limits in Permits	04
		325	Statistical basis for benchmarks and limits in Permits	04
		3.2.6	Reference conditions for releases to air	65
4	BENG	HMARK	RELEASE LEVELS	66
	4.1	BENCH	IARK RELEASES TO AIR	66
	4.2	BENCHN	ARK RELEASES TO SURFACE WATER	67
5	IMPA	ст		68
	5.1	Assess	MENT OF THE IMPACT OF EMISSIONS ON THE ENVIRONMENT	68
	5.2	THE WA	STE MANAGEMENT LICENSING REGULATIONS	69
	5.3	THE HAI	BITATS REGULATIONS	70
REF	EREN	CES		71

# **Preliminary Pages**

DEFINITIONS	73
APPENDIX 1 - SOME COMMON MONITORING AND SAMPLING METHODS	74
APPENDIX 2 - EQUIVALENT LEGISLATION IN SCOTLAND & NORTHERN IRELAND	76

# **TABLE OF FIGURES**

Table 2-1Fumes evolved from current foundry processes	. 26
Table 2-2 Solid waste stream routes	. 43
Table 3-1 Emissions to air associated with the use of BAT	. 63
Table 4-1 Benchmark levels for releases to air <sup>a</sup>	. 66

INTRO	DUCTION	TEC	CHNIQUES	S EN	IISSIONS	IMP	ACT						
IPPC and BAT	Making an application	Installations covered	Review periods	Key issues	Summary of releases	Sector overview	Economic aspects						
	1 INT 1.1 U	RODUC	TION ding IPPC	and BA	т								
IPPC and the Regulations	Integrated approach t the approp To gain a F apply the 'F relevant loo	Pollution Prever o control the enviriate controls for Permit, Operator Best Available To cal factors.	tion and Contro vironmental impa industry to prot s will have to sh echniques' (BAT	I (IPPC) is a r acts of certair ect the envirc ow that they I ) and meet c	egulatory system i industrial activitie nment through a s nave systematicall ertain other require	that employs a s. It involves c ingle permitting y developed pr ements, taking	n integrated letermining g process. oposals to account of						
	The essent appropriate	ce of BAT is that balance betwe	t the selection of en realising env	f techniques t ironmental be	o protect the environ nefits and costs in	onment should curred by Ope	achieve an rators.						
	IPPC opera 3 and Appe (PPC) Act system of I Regulation document	ates under the P endix 2). These 1999 and impler PPC, together w s, can be found IPPC: A Practica	rollution Prevent Regulations have nent the EC Dire vith Government in the Departme al Guide, (see R	ion and Contr ve been made ective 96/61 c policy and m ent of the Env ef. 4).	ol (England and W under the Pollution IPPC. Further in ore detailed advice ronment, Transpo	/ales) Regulati on Prevention a nformation on t e on the interp rt and the Regi	ons, (see Ref. and Control the overall retation of the ions (DETR)						
Installation based, NOT national emission limits	The "BAT" limits (exce instrument level.	The "BAT" approach of IPPC is different from regulatory approaches based on fixed national emission limits (except where General Binding Rules have been issued by the Secretary of State). The legal instrument which ultimately defines BAT is the permit and this can only be issued at the installation level.											
Indicative BAT standards	Indicative E guidance ( action. It sl Section 2 a direction, c installation Notwithsta although B	Indicative BAT standards (essentially for BAT but also covering other aspects) are laid out in national guidance (such as this) and should be applied unless there is strong justification for another course of action. It should be noted that BAT includes both the technical components of the installation given in Section 2 and the benchmark levels identified in Section 3. Departures from those standards, in either direction, can be justified at the local level taking into account the technical characteristics of the installation concerned, its geographical location and the local environmental conditions. Notwithstanding this, if there are any applicable mandatory EU emission limits, they must be met, although BAT may go further than them.											
BAT and EQSs	The "BAT" Environme or, where t prevented environme as a recipie practicable reasonably and only th Guidance a beyond the	although BAT may go further than them. The "BAT" approach is also different from, but complementary to, regulatory approaches based on Environmental Quality Standards (EQS). Essentially BAT requires measures to be taken to <b>prevent</b> or, where this is not practicable, to reduce emissions. That is, if emissions can be reduced further, or prevented altogether, at reasonable cost, then this should be done <b>irrespective</b> of whether any environmental quality standards are already being met. It requires us not to consider the environment as a recipient of pollutants and waste, which can be filled up to a given level, but to do all that is practicable to minimise the impact of industrial activities. The process considers what can be reasonably achieved within the installation first (this is covered by Sections 2 and 3 of this Guidance) and only then checks to ensure that the local environmental conditions are secure, (Section 4 of this Guidance and Ref. 6). The BAT approach is, in this respect, a more precautionary one, which may go											
	Conversely still threate However, t assessmer harm) shou appropriate	/, it is feasible th ned. The Regul his situation sho ht of harm. The uld in most cases to protect the e	at the applicatio ations therefore uld arise very ra BAT assessmer s have come to nvironment.	n of what is E allow for exp arely assumin at, which bala the same con	AT may lead to a enditure beyond B g that the EQS is s nces cost against I clusion about the e	situation in whi AT where nece soundly based benefit (or prev expenditure wh	ch an EQS is essary. on an vention of nich is						
	Advice on given in <i>IP</i> .	the relationship ( PC: A Practical (	of environmenta <i>Guide,</i> ( <mark>see Ref</mark> .	l quality stand 4) and in Se	lards and other sta ction 3.	andards and ob	ligations is						
Assessing BAT at the sector level	The assess BAT refere information flexibility to information At this nation appropriate Secondly, uncompetion	sment of BAT ta nce document ( which member member states contained in the onal level, techn balance of cos the techniques s tive either on a E	kes place at a n BREF) for each states should ta in its application <b>BREF</b> and lays iques which are ts and benefits f hould normally European basis o	umber of leve sector. The like into account in. This UK Se s down the in considered to or a typical, w be affordable or worldwide.	Is. At the Europea BREF is the result nt when determini ector Guidance No dicative standards b be BAT should, f rell-performing inst without making the NOTE: BREF not	an level, the EC of an exchange ng BAT, but wi te takes into ar and expectatio irst of all, repre- callation in that e sector as a w available yet.	C issues a e of hich leaves count the ons in the UK. esent an sector. thole						

INTRO	DUCTION	TEC	CHNIQUES	S	EMIS	SSIONS	IMP	ACT				
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects				
Assessing BAT at the installation level	When asse departures may deper costs and l company p In summar installation grounds of	essing the applic may be justified of upon local factor penefits of the a profitability is <b>no</b> y, departures m concerned, its g individual comp	ability of the sed t in either directi ctors and, where vailable options t considered. ay be justified of geographical loc vany profitability.	ctoral on as the a may n the ation Furl	, indicative described answer is n be needed grounds of and the loo her informa	BAT standards a above. The mo ot self evident, a to establish the the technical ch cal environmenta ation on this can	at the installati st appropriate local assessr best option. In aracteristics o l conditions, b be found in th	on level technique nent of the ndividual f the nut not on e Guide for				
	Applicants	, (see Refs. 4 ar	nd 5). No account at th									
	<ul> <li>where the BAT cost/benefit balance of an improvement only becomes favourable when the relevant item of plant is due for renewal/renovation anyway (e.g. BAT for the sector may be to change to a different design of furnace when a furnace comes up for rebuild). In effect, these are cases where BAT for the sector can be expressed in terms of local investment cycles.</li> </ul>											
	<ul> <li>where a approp (see Ref</li> </ul>	<ul> <li>where a number of expensive improvements are needed, a phasing programme may be appropriate as long as it is not so long as to be seen as rewarding a poor performing installation, (see Ref. 6 for more details).</li> </ul>										
Innovation	The Regula meet the B performand Note descr keep up to cited in an technical c Guidance; valid to cor	The Regulators encourage the development and introduction of new and innovative techniques which meet the BAT criteria and are looking for continuous improvement in the overall environmental performance of the process as a part of progressive sustainable development. This Sector Guidance Note describes the appropriate indicative standards at the time of writing. However, Operators should keep up to date with the best available techniques relevant to the activity and this Note may not be cited in an attempt to delay the introduction of improved, available techniques. Furthermore, the technical characteristics of a particular installation may allow for opportunities not foreseen in the Guidance; as BAT is ultimately determined at the installation level (except in the case of GBRs) it is										
New installations	The indicat departures should nor the require timescales	tive requirement from them in th mally be in place ment is for an a are given for su	s apply to both r e case of new a e before the con udit of ongoing o ich cases.	new a ctivitio nmen opera	nd existing es. For nev cement of c tions, this is	activities but it v w installations, th operations. In so s not feasible an	will be more di ne indicative re ome cases, su d indicative up	fficult to justify equirements ch as where ograding				
Existing installations - standards	For an exis acceptable using differ case it may small decre	sting activity a le where the activ rent plant or pro- y impose a dispr ease in emissior	ss strict proposa ity operates to a cesses from tha oportionate cos is.	al (or a stan t upor t to re	an extende dard that is n which the place the c	ed timescale) ma s very close to ar indicative requine ind plant with the	y, for example n indicative rec rement is base new techniqu	, be quirement, but ed. In such a es for only a				
Existing installations - timescales	small decrease in emissions. The timescales for the major cost improvements for the control of NO <sub>x</sub> , SO <sub>x</sub> , particulate, dioxins and effluent treatment in this sector will depend upon local factors and the results of the cost benefit assessments. The timescales for such improvements should be to the approval of the Regulator.											
	The whole of the Pern principles a	programme of a nit. Any longer f above.	iny other items s imescales will n	should eed to	d be comple o be justifie	eted at the latest d by the Operate	t within 3 years or in accordan	s of the issue ce with the				
	All improve Regulator.	ements should b	e carried out at	the ea	arliest oppo	ortunity and to a	programme ap	proved by the				
	The Appli	cant should inc	lude a propose	ed tim	netable cov	vering all impro	vements.					

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	IPPC and Making an Insta BAT application co		allations overed	Review periods	Key issues		Summary of releases	Sector overview	Economic aspects

# 1.2 Making an Application



- **Note 1** The amount of detail needed to support the application should be sufficient to support the Applicant's contention that either the conditions of the guidance have been met or an alternative measure has been justified. The level of detail should be commensurate with the scale of the operation and its ability to cause pollution. An Applicant is not required to supply detail that could not reasonably be expected to contribute to a decision to issue a Permit.
- **Note 2** For existing IPC or Waste Management Permit holders, your response to each point in Sections 2, 3 or 4 may rely heavily on your previous application. The Regulator does not wish you to duplicate information as long as the previous information adequately addresses the issues. However, the more the information can be reorganised to demonstrate that all the issues have been adequate addressed the better. You will need to send us copies of any information referred to.
- **Note 3** The contents of the outlined BAT boxes in Sections 2, 3 and 4, and additional blank tables etc., are available electronically on the Environment Agency's Website, for the assistance of Applicants.

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co	allations overed	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

# **1.3 Installations Covered**

This Note covers installations, described in Section 2.1 Part A (Part A(1) in England and Wales) of Schedule 1 to the PPC Regulations (see Ref. 3) as follows.

(b) Producing, melting or refining iron or steel or any ferrous alloy, including continuous casting, **except** where the only furnaces used are –

- (i) electric arc furnaces with a designated holding capacity of less than 7 tonnes, or
- (ii) cupola, crucible, reverbatory, rotary, induction or resistance furnaces.

Processes prescribed under Part A(2) are as follows:

- (a) Producing pig iron or steel, including continuous casting, in a plant with a production capacity of more than 2.5 tonnes per hour unless falling within paragraphs (b) of Part A(1) of this Section.
- (d) Casting ferrous metal at a foundry with a production capacity of more than 20 tonnes per day.

In this Section, "ferrous alloy" means an alloy of which iron is the largest constituent, or equal to the largest constituent, by weight, whether or not that alloy also has a non-ferrous metal content greater than any percentage specified in Section 2.2.

The installation includes the main activities as stated above and associated activities which have a technical connection with the main activities and which may have an effect on emissions and pollution.

This guidance note does **not** address melting or refining activities that take place within the electric arc furnace, or within an argon oxygen decarburisation (AOD) converter; nor does it address the issues associated with raw materials, emissions or waste from those operations. All such activities are addressed by the IPPC Guidance note; S2.01 Guidance for the Coke, Iron and Steel Sector.

This guidance note does address the following aspects of the prescribed process, which are considered to be "foundry" operations, namely operations that occur after the melt has been tapped from the furnace:

- storage and handling of raw materials (only those associated with the following foundry operations)
- launders
- desulphurisation of molten iron in ladles
- nodularisation of SG iron in ladles
- preparation of moulds and cores
- casting, pouring or moulding
- the power plant
- a waste to energy plant
- knocking out
- fettling, dressing or finishing of castings
- sand reclamation
- waste handling and recycling facilities
- Advice on the extent of the physical site which is contained within the installation, e.g. split sites, is given in *IPPC Part A(1) Installations: Guide for Applicants* (see Ref. 5). Operators are advised to discuss this issue with the Regulator prior to preparing their application.

Where associated activities are carried out **in conjunction** with the main activities and are not covered in this guidance note (for example combustion activities), reference should be made to:

- other relevant IPPC Guidance Notes and,
- where appropriate, the Secretary of State's Guidance for Local Authority Air Pollution Control. (NB In Northern Ireland this guidance is produced by the Department of the Environment') in particular

PG 1/11(96) reheat and heat treatment furnaces, 20-50 MW net rated thermal input

PG 1/3(95) boilers and furnaces, 20-50MW net rated thermal input

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co\	llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

### **1.4 Review Periods**

Permits can be reviewed or varied at any time. However, the PPC Regulations impose a requirement on Regulators to review Permits in certain specific circumstances such as where the pollution caused by the installation is of such significance that the existing emission limit values need to be revised or new limits set.

In addition, Regulators are required to review the conditions of Permits "periodically". The Government stated in its third consultation paper (England, Wales and Scotland) on the implementation of IPPC, that the new sector-specific IPPC Sector Guidance Notes would provide guidance on appropriate review periods for each sector. These would take into consideration guidance on the relevant criteria, to be provided by the Government. Examples of the likely relevant criteria for setting these review periods are "the risk and level of environmental impacts associated with the sector" and "the cost to the Regulators and regulated industry of undertaking the reviews".

The Regulators consider that at the present time, having regard to those criteria, it is in fact appropriate to set indicative minimum review periods which differ only between those sectors which have been subject to integrated permitting (i.e. IPC or Waste Management Licensing), and those which have not. It is therefore proposed that Permit conditions should normally be reviewed on the following basis:

- for individual activities NOT previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within four years of the issue of the IPPC Permit;
- for individual activities previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within six years of the issue of the IPPC Permit.

This means that activities/installations not currently in IPC or Waste Management Licensing will be initially reviewed within four years and thereafter within six years.

An exception to this is where discharges of List I or List II substances have been permitted or where disposal or tipping for the purposes of disposal, of any matter which might lead to an indirect discharge of any substance on List I or II. In such cases the review must be carried out within four years.

This period will be kept under review and, if any of the above factors change significantly, may be shortened or extended.

INTRODUCTION		TECHNIQUES		EMI	SSIONS	IMPACT			
IPPC and BAT	Making an application	Insta cov	llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

# **1.5 Key Issues for this Sector**

### Nature and consumption of raw and recycled materials

Selection and use of materials to give the best practicable environmental option. This will include making choices that enhance the opportunities for recycling. Choice of binder systems, release agents and carrier solvents will need to be considered.

### Raw material storage and handling

Fugitive dust emissions from unloading, transport, storage and reclamation of stocks, principally of fluxes, refractories, desulphurisation materials (such as lime or limestone, carbon, aluminium), nodularisation materials (such as magnesium or a magnesium alloy with iron or nickel), sand and binders. Transfer and storage facilities for handling must be designed to minimise the risk and consequences of spillage. Both air and groundwater contamination are to be avoided.

### Air quality management

The potential for air pollution is a major problem associated with foundries. Potentially significant pollutants are:

- particulate matter;
- nitrogen oxides;
- carbon oxides;
- iron and its oxides;
- heavy metals;
- ammonia;
- VOCs including formaldehyde, phenols and esters, and
- Dioxins, where dirty scrap is used

Many of these pollutants are malodorous.

Emissions from launders consist of iron and its oxides.

Dust and fume from refining in ladles may include :

a) from **desulphurisation** of molten iron in ladles – magnesium oxide or calcium oxide, carbon monoxide and carbon dioxide

b) from nodularisation of SG iron in ladles - magnesium oxide (in large quantities).

Preparation of **moulds and cores** gives rise to dust from sand handling and gases from any resin, hardener and catalyst used (the binder system) and their reactions during mixing and curing. The different binder systems give rise to different emissions, but the main types emit two or more of the following gases: ammonia, hydrogen sulphide, sulphur dioxide, methyl diisocyanate, phenol, formaldehyde and a range of other VOCs including amines and esters.

Mould and core storage areas may have high concentrations of VOCs such as triethylamine (TEA) and dimethylamine (DMA), which exude from the cores.

**Casting, pouring, moulding and knocking out** give rise to emissions relating to the pyrolisation of the moulds. They include all of those mentioned above for the preparation of moulds and cores, as well as carbon monoxide, carbon dioxide and some PAHs such as cresols and xylenols that are malodorous.

**Fettling, dressing or finishing** give rise to particulate matter and some fume if techniques involving heat are used. In particular, metallic dusts from shot blastting operations are highly aggressive and damaging to paintwork.

**Sand reclamation** gives rise to dust from mechanical reclamation and fume from thermal reclamation. (An aqueous stream is created by wet reclamation techniques.)

More detail relating to the emissions from each part of the process is provided in section 2.3

### Water management

Key issues to be addressed include the following:

- consumption levels;
- monitoring and management of mass flows of individual pollutants;
- management of surface water run-off and treatment facilities;
- security of underground drains; and
- pollution prevention systems and contingency arrangements.

INTRODUCTION		TECHNIQUES			EMI	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co	llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

### Energy efficiency and fuels

Operators in a CCL Agreement are not expected to implement capital improvement projects for energy efficiency under a PPC Permit.

The sector is a major energy consumer. There remain significant opportunities for reduction of emissions caused by energy use and choice of energy source ( $CO_2$ ,  $SO_x$ ,  $NO_x$ , etc. contributing in particular to global warming and acidification). The industry may enter into a Climate Change Agreement or Trading Agreement with the Government. The applicability of techniques and standards for IPPC is explained in Section 2.7.

A Negotiated Agreement has been made between the UK Steel Association and the Government concerning a rebate of the Climate Change Levy (CCL). Signatories will be subject to a reduced level of site specific regulation on energy efficiency matters, in particular capital expenditure is not required on energy efficiency improvements beyond the baseline measures (though a plan has to be provided in case the Operator drops out of the agreement). Some activities covered here may be exempt from the CCL, e.g. The applicability of techniques and standards for PPC is explained in Section 2.7.

Key issues include:

- reclamation of heat from cooling plant;
- minimising use of heavy fuel oil;
- low NO<sub>x</sub> combustion systems; and
- the balance between energy use and environmental protection.

### Recovery, recycling and waste disposal

A strategy and plan are required for the minimisation, recovery and recycling of process materials and for the disposal of waste, in accordance with the Regulator's policies on waste minimisation. Considerations should include collected dusts and slags, coolant water, treated effluents and recovered oils. Wastes sent to landfill are a key issue, particularly slag from the ladles, scrubber liquors and sludge, refractory waste, sand and fettling waste.

#### Noise

The major noise sources associated with foundry processes are related to mould production and handling including pattern cleaning, knocking out and cleaning permanent moulds, fettling and finishing castings, final product handling and sand reclamation. All potential sources of significant noise need to be identified and managed.

### Accident risk and pollution prevention

Apart from the normal process and spillage risks, many older sites (especially those not regulated under IPC) will have drainage systems that will need attention (see Section 2.8).

An assessment is required of the environmental hazards posed by non-routine operation and accidents. This should be co-ordinated with any responsibilities under the COMAH Regulations. Appropriate pollution prevention measures and contingency arrangements are required, protecting all environmental media.

### Long distance and transboundary pollution

The foundries are unlikely to be of sufficient size to have significant transboundary effects.

### Site restoration

**BAT improvement:** Air quality management plan to meet AQS objectives Many ferrous foundries will have been operating on the same site for many years. There may well be ground contamination that could be confused with potential future contamination from the activities as operated under IPPC. In such cases it will be necessary to assess the degree of contamination as a baseline for future operations.

Consideration for remediation would include:

- contamination of stocking and handling areas for raw and recycled materials;
- heavy metals and alkaline materials at slag and metal recovery operations in alloy steel processes;
- alkaline and sulphide contamination at other slag sites; and
- sediments in lagoons.

INTRODUCTION		TECHNIQUES		EMI	SSIONS	IMPACT			
IPPC and BAT	Making an application	Insta cov	llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

# **1.6 Summary of Releases for each Sector**

Releases to air usually result in a subsequent, indirect emission to land and can therefore affect human health, soil and terrestrial ecosystems.

For releases from combustion and incineration plant see the appropriate guidance.

SOURCE       Martin		Ra	La	Dee	No	Prep	Ca	Knoo	Fett casti	Slaç proc
Oxides of sulphurAAAAAAOxides of nitrogenAAAAAAACarbon dioxideAAAAAAACarbon monoxideAAAAAAAHydrogen sulphideAAAAAAAmmoniaAAAAAAOxides of ironAAAAAAAlkali metal compoundsAAAAAAlkaline-earth metal compoundsAAAAAMetal oxide particulatesAAAAANon-metallic particulatesAWAAAAMetallic ironImage: CompoundsAAAAHydrogen cyanideImage: CompoundsAAImage: CompoundsCadmium and cadmium oxideImage: CompoundsImage: CompoundsImage: CompoundsSulphurAAImage: CompoundsImage: CompoundsImage: CompoundsVolatile organic compoundsImage: CompoundsIma	SOURCE	v material storage and handling	Inders	ulphurisation of molten iron	Jularisation of SG iron	aration of cores and moulds	ting, pouring and moulding	ing out, reclamation	ing, dressing or finishing of ngs	and dust from refining asses
Oxides of nitrogenImage: solution of the solution of	Oxides of sulphur			Α		А	А	А		
Carbon dioxideImage: solution of the	Oxides of nitrogen					А	А	А		
Carbon monoxide         Image: solubitity of the solution of the sol	Carbon dioxide			Α	А	А	А	А		
Hydrogen sulphideImage: solution of the solution of t	Carbon monoxide			Α	А	А	А	А		
AmmoniaImage: sector secto	Hydrogen sulphide					Α	А	А		
Oxides of ironImage: A image: A	Ammonia					А	А	А		
Alkali metal compoundsAALLLImage: constraint of the second s	Oxides of iron		A	Α	А		А	AL	AL	
Alkaline-earth metal compoundsAALALAAAMetal oxide particulatesAWAAAAAALALALNon-metallic particulatesAWAAAALALALALALMetallic ironImage: CompoundeImage: Compout	Alkali metal compounds		Α	AL		L				
Metal oxide particulatesAAAAAALALALNon-metallic particulatesAWAAAALAALALALALMetallic ironImage: CompoundeImage: Co	Alkaline-earth metal compounds		Α	AL	Α	L	Α			
Non-metallic particulatesAWAAALALALALALMetallic ironII	Metal oxide particulates		Α	Α	Α		A	AL	AL	L
Metallic ironImage: second	Non-metallic particulates	AW	A	A	Γ	AL	A	AL	AL	L
Hydrogen cyanideImage: state of the state of	Metallic iron								AL	T
Cadmium and cadmium oxideImage: state of the	Hydrogen cyanide				Γ	A	「 <u> </u>			Τ
Zinc, lead and their oxidesImage: second	Cadmium and cadmium oxide									L
SulphurAAAAAAmines/amidesIIIAAIIDioxinsIIIIAAIIVolatile organic compoundsIIIAAIIOils and greasesIIIIIIIIIResinsII	Zinc, lead and their oxides				Γ	「 <u> </u>	「 <u> </u>			L
Amines/amidesImage: second	Sulphur			A	Γ	「 <u> </u>	「 <u> </u>			T
DioxinsAAVolatile organic compoundsAAOils and greasesLLResinsLLAcid vapoursAASlag wasteLLSludgesLLLLLRefractory wasteLLAALLL <td>Amines/amides</td> <td></td> <td></td> <td></td> <td>Γ</td> <td>A</td> <td>A</td> <td></td> <td></td> <td>Τ</td>	Amines/amides				Γ	A	A			Τ
Volatile organic compounds       A       A       A       A         Oils and greases       L       L       L       L       L         Resins       L       L       L       L       L       L         Acid vapours       A       A       A       A       A       L <t< td=""><td>Dioxins</td><td></td><td></td><td></td><td>Γ</td><td></td><td>A</td><td></td><td></td><td>T</td></t<>	Dioxins				Γ		A			T
Oils and greases       Image: Constraint of the second secon	Volatile organic compounds					A	A	A		
ResinsIILLIAcid vapoursIIAAISlag wasteILLIIISludgesIIIIIIUVRefractory wasteLIIALL	Oils and greases					L				
Acid vapours       A       A       A         Slag waste       L       L       L       L         Sludges       L       L       L       LW         Refractory waste       L       L       AL       L	Resins					L		L		
Slag waste     L     L     L     L       Sludges     Image: Sludges     Image: Sludges     Image: Sludges     Image: Sludges       Refractory waste     L     Image: Sludges     Image: Sludges     Image: Sludges	Acid vapours				<u> </u>	A	А			<u> </u>
Sludges     LW       Refractory waste     L	Slag waste			L	L	「 <u> </u>	「 <u> </u>		Γ	Τ
Refractory waste     L     L     AL     L	Sludges				<u> </u>				LW	<u> </u>
A Deless to Ale W. Deless to Weter L. Deless to Lond	Refractory waste		L			L		AL	L	
KEY A – Release to Air, W – Release to Water, L – Release to Land	KEY		A – R	elease to A	Air, W – Re	elease to V	Vater, L –	Release to	Land	

Releases to air may also be released to land or water, depending upon the abatement technology employed, e.g. via collected dusts, sludges or liquors.

Some releases are specific to a particular binder system.

INTRODUCTION			TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta cov	llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects	

# **1.7** Overview of the Activities in this Sector

The processes which are described in this Section are those Part A foundry processes defined in Schedule 1 of the Regulations<sup>(1)</sup> which are related to, or used in conjunction with the melting of iron and steel **in an electric arc furnace with a capacity of more than 7 tonnes**. As a result this Guidance Note applies to a small number of the larger iron and steel founders using electric arc furnace melting to produce castings to their final, or near final shape. The main operations covered by the description in Section 2.1 of the Regulations and which are included in this Note are:-

- Storage and handling of raw materials;
- Launders;
- Desulphurisation of molten iron in ladles;
- Nodularisation of SG iron in ladles;
- Preparation of cores and moulds;
- Casting, pouring or moulding;
- Knocking out;
- Sand reclamation
- Fettling, dressing or finishing of castings; and
- Handling of waste materials.

INTRODUCTION			TEC	CHNIQUES	3	EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application	Insta co\	llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects	

## **1.8 Economic Aspects for this Sector**

### Introduction

In determining an application, the Regulator must translate BAT into conditions to be included in the Permit. These criteria require that judgements are made balancing cost against environmental damage, and also concerning the relative significance of discharges to different media. The operator should have presented the reasons why the proposed option is considered to represent BAT. The advantages over any other feasible techniques should be identified.

The cost of controlling releases should not be disproportionate to the environmental benefits delivered. There may be cost savings resulting from changes made to improve environmental performance. BAT is not concerned with the financial health or resources of a particular operator, and excessive costs are viewed in the context of the process and the industry. While techniques and achievable release levels may vary between industry sectors, the same principles apply to all sectors. For existing processes, the timing of improvement programmes may be a factor in the determination of BAT.

The information contained in this section gives background on the economics of the industry and the ability of operators to raise funds for investment as well as an indication of possible abatement costs.

### **1.9 This sector**

Information concerning the economic position of this industry sector is given in IPC Guidance Note S2 2.01.

THE FOUNDRY BREF NOTE WILL CONTAIN UP TO DATE INFORMATION FOR THE SECTOR ACROSS EUROPE. The BREF note is expected to be published in 2002.

### 1.10 Site-specific issues

Disposing of used foundry sand represents a major item, both environmentally and financially.

Environmental Technology Good Practice Guide GG119 (see Ref 26) provides valuable guidance to assist in optimising use of sand, whilst the sister publication GG104 (see Ref 26) provides similar guidance on the management of chemical binders. Application of the principles developed in these documents, after due regard for any overarching BAT requirements applicable at the site under consideration, will lead to cost effective waste minimisation and pollution prevention at source.

As the process is, by definition, part of a much larger operation, there needs to be careful consideration of the range of operations impacted, and the procedure for allocating costs.

### Cost information for abatement techniques

### Sand recycling

Figures presented in GG119 (see Ref 26)show that there is **significant financial benefit** to be obtained by operating an effective sand recovery system.

A primary reclamation system handling 5520 tonnes of sand per year is described as achieving a net cost saving of £147,108 per year.

The same source describes a thermal reclamation system with a throughput of 1 tonne per hour, yielding a net cost saving of £92,581 per year.

# **2 TECHNIQUES FOR POLLUTION CONTROL**

This section summarises, in the outlined BAT boxes

- BAT Boxes to help in preparing applications
- what is required in the application
- the indicative standards (i.e. what is BAT in most circumstances).

The indicative standards cover the techniques and measures which have been identified as representing BAT in a general sense. They also cover the other requirements of the PPC Regulations and requirements of other Regulations such as the Waste Management Licensing Regulations (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland), and the Groundwater Regulations insofar as they are relevant to an IPPC Permit. For the sake of brevity these boxes simply use the term "BAT".

At the top of each BAT box is the question from the application form (derived from the Regulations) which is being addressed, (see Section 1.2).

In responding to the requirements the Operator should keep the following general principles in mind.

- As a first principle there should be evidence in the application that full consideration has been given to the possibility of **PREVENTING** the release of harmful substances. This may, most commonly, be by:
  - substituting materials or processes (see Section 2.2.1), For example, scope for this in this sector would be by the replacement of binder systems and release agents with less harmful alternatives.
  - -preventing releases of water altogether (see Section 2.2.3),

-preventing waste emissions by reuse or recovery.

- Only where that is not practicable should the second principle be adopted of reducing emissions which may cause harm.
- All available options should be reviewed and it should be demonstrated that the selected combination of primary process and abatement equipment satisfies the Regulations.
- All plant and equipment should be subject to regular preventative maintenance programmes, in line with operational requirements, to ensure continued optimum performance. This should be detailed in response to Section 2.1 and elsewhere as appropriate.

# 2.1 Management Techniques

Within IPPC, an effective system of management is a key technique for ensuring that all appropriate pollution prevention and control techniques are delivered reliably and on an integrated basis. The Regulators strongly support the operation of environmental management systems (EMSs). An Operator with such a system will find it easier to complete not only this section but also the technical/regulatory requirements in the following sections.

The Regulators recommend that the ISO 14001 standard is used as the basis for an environmental management system. Certification to this standard and/or registration under EMAS (EC Eco Management and Audit Scheme) (OJ L168, 10.7.93) are also strongly supported by the Regulator. Both certification and registration provide independent verification that the EMS conforms to an assessable standard. EMAS now incorporates ISO 14001 as the specification for the EMS element. For further details about ISO 14001 and EMAS contact British Standards Institute (BSI) and the Institute of Environmental Management and Assessment (IEMA) respectively.

The steps required in this and subsequent sections may help the Operator to make good any shortfalls in their management system. An effective EMS will help the Operator to maintain compliance with regulatory requirements and to manage other significant environmental impacts. While the requirements below are considered to be BAT for IPPC, they are the same techniques as required in a formal EMS and are also capable of delivering wider environmental benefits. However, it is information on their applicability to IPPC which is primarily required in this application.

Application Form Question 2.1 Provide details of your proposed management techniques.

### With the Application the Operator should:

1. Describe their management system to demonstrate how it meets the *"Requirements for an effective management system"* below. The description should make clear who holds responsibility for each of the requirements. The second column explains where in the application the response to each requirement is best dealt with to avoid duplication. Copies of all procedures are not needed, but examples may be included in your application.

If you are certified to ISO 14001 or registered under EMAS (or both), you may provide a statement derived from certification records/assessments to support your application.

Further specific management procedures are dealt with under the appropriate section on the remainder of the document. It is recommended that you understand all the requirements of the application before completing this section, as many management issues are dealt with in other sections.

2. The type of management system employed will depend upon the scale and complexity of the operations undertaken. The Operator should demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements below, or by justifying departures (as described in Section 1.2 and in the *A1 Guide for Applicants*) or alternative measures.

### Indicative BAT Requirements

The Operator should have a management system in place for the activities which delivers the requirements given in column 1 below. The development of any aspects of the management system not already in place should be completed within the timescale given in Section 1.1.

Re	quirement for an effective management system	How delivered for IPPC
1.	Clear management structure and allocated responsibilities for environmental performance, in particular meeting the aspects of the IPPC Permit	Describe in this section who has allocated responsibilities
2.	Identification, assessment and management of significant environmental impacts	By responding to the requirements in Section 5.1 in the application
3.	Compliance with legal and other requirements applicable to activities impacting on the environment	Compliance with the Permit satisfies this requirement

Cont.

BAT for management techniques

INTRODUC	CIT	N TEC	HNIQ	JES	E	<b>AISSIO</b>	٧S	IMPACT
Management Management	aterials	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring Closure Installation
	p ato							
	4. 5.	Establishing a objectives an requirements a performance Environmenta implement po	an enviro d targets and contin al improve blicy obje	nmental to prever ually imp ement pr ctives an	policy a nt pollutic prove envi rogramm nd target	nd setting In, meet lega ironmental es to s	al C s ir n e T	he applicant should provide a opy of their environmental policy tatement applicable to the installation. The Applicant should nake proposals in response to ach of Sections 2.2 to 2.12.
							ir ir	ncorporated within the Permit nprovement programme
	6.	Establish ope minimise sigr	erational o nificant ei	controls nvironme	to preve ental imp	nt and acts	E ir 2	y responding to the requirements Sections 2.2 to 2.7, 2.11 and .12 in the application
	7.	Preventative plant and equ	t D ws p	escribe system here. List rocedures in Section 2.3				
	8.	Emergency p	B ir	y responding to the requirements Section 2.8 in the application				
	9.	Monitoring an Identify key ind establish and monitor indica	nd measu dicators of maintain a tors to ena	ring perf f environr a program able revie	formance mental pe nme to me w and im	erformance a easure and aprovement o	nd of	escribe in this Section
	10. • •	Monitoring ar to ensure that to detect faults to detect slow	nd contro the install and unin changes i	I system ation fund tended op n plant po	<b>s</b> : ctions as perations erforman	intended; ; ce to trigger	B ir	y responding to the requirements Section 2.10 in the application
	11. • • •	Training Provision of ac relevant staff ( equipment and a clear statem for each job; awareness of the activity and awareness of operation unde prevention of a when accident implementation Expertise requ out. However whom the inst sufficient quali roles. This ma standards or of <b>Communicati</b>	dequate principality of the regula di materials ent of the the regula di their wor all potentia er normal accidental cal emission n and mai uired deper, both tech allation's o ifications, ay be asse codes of pon and re-compliar	rocedure: contracto s), which skills and tory impli k activitie al enviror and abno emission ons occur ntenance ends on th nnical and compliand training a essed aga ractice	s and trai ors and th should in d competer cations o es; mental e ormal circ is and ac ; e of trainir ne activitie d manage ce depen and exper ainst any of incide complair	ning for all ose purchas iclude: encies requir f the Permit ffects from umstances; tion to be tak ng records; es being carrierial staff upo ds need rience for the industry sec nts of actua nts	ried for for tir tor al or D	o be described in this Section onfirming that training for each of he areas covered by Sections 2.2 o 2.3 and 2.5 to 2.10 are covered
		Actions taken to operations	in respons	se, and a	bout prop	oosed chang	es	Cont
								0011.

INTRODU	JCTIO	N TEC	HNIQ	UES	El	VISSIO	٧S	l	MPAC	Т
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	13.	Auditing Reg check that all with these rec be audited at	ular, (pref activities a uirements least once	erably) in are being 5. All of th e per year	depende carried o nese requ	nt, audits to ut in conforn irements sho	nity ould	Describe in thi	is Section	
	14.	Corrective ac recurrence Define respon investigating in any impacts of corrective and Recording, im	ction to an nsibility and non-confor aused and d preventiv vestigating	d authorit mance, ta for initia e action g, taking c	ults and y for han aking act ting and corrective	prevent dling and ion to mitigat completing action and	te	Describe in this dealt with for .2 to 2.3 and ppropriate	is Section or each of 2.5 to 2.1	how this Sections 0 as
		preventing recomplaints ar	currence, i nd incident	in respons s	se to env	ironmental				
	15.	Reviewing an	nd Report	ing Envir	ronment	al Performa	nce			
		Senior manage and ensure and ensure that por remains releve Programmes	gement rev ppropriate olicy comn ant. Revie at least ar	view envir action tal nitments a ew progre nnually.	onmenta ken wher are met a ess of the	l performance e necessary and that polic Managemer	xe D to y nt	escribe in the	is Section	
		Incorporate en aspects of the IPPC, in partic	nvironmen e business cular:	ital issues , insofar a	in all oth as they a	ner relevant re required b	ry C	escribe in the	is Section	
	•	the control of design and re capital project capital approv the allocation planning and	process cl view of ne ts; /al; of resourc scheduling	hange on w facilitie es;	the insta s, engine	llation; ering and ot	her			
	•   •	incorporation operating pro- purchasing po- accounting for	of environ cedures; blicy; r environm	mental as	spects int	o normal	s			
		involved rathe Report on env results of mar audit cycle), fe	er than as o vironmenta nagement or:	overhead al perform reviews (a	s. lance, ba annual ol	sed on the r linked to the	e re	<sup>-</sup> his will becor equirement	me a Perr	nit
		information re effectiveness objectives and	equired by of the mar d targets, a	the Regul nagement and future	lator; and system planned	l against l improvemer	nts.	Describe in thi Describe in thi	is Section is Section	1
		statement	any preter	ару ма р	ublic env	nionmental				
	16.	Managing do	cumentat	tion and i	records					
		List the core e responsibilitie documentatio update docum	elements o s, procedu n in order nentation.	of the EMS ures etc) a to be able	6 (policie and links e to contr	s, to related ol, locate and	d C	Describe in thi	is Section	
		Describe how and reviews a	environm re identifie	ental reco ed, mainta	ords and ained and	results of aud I stored	dits			

INTROD	UCTIO	N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Selection of raw materials	2.2 Materials Inputs
	This section covers the use of <b>raw materials and water</b> and the techniques for both minimising their use and minimising their impact by selection. (The choice of fuels is covered under Section 2.7.3, Energy).
	As a general principle, the Operator will need to demonstrate the measures taken to:
Reduce	• reduce the use of chemicals and other materials (Section 2.2.2);
Substitute	<ul> <li>substitute less harmful materials or those which can be more readily abated and when abated lead to substances which in themselves are more readily dealt with:</li> </ul>
Understand	<ul> <li>understand the fate of by-products and contaminants and their environmental impact (Section 5).</li> </ul>
	2.2.1 Raw materials selection
Summary of materials in	This section looks at the <b>selection and substitution</b> of raw materials used while Section 2.2.2 describes the techniques to <b>minimise</b> their use.

materials in use

Raw materials used in foundry processes consist of:-

- desulphurisation materials; •
- nodularisation materials; ٠
- mould materials; principally sand and binders, also catalysts where required; •
- refractories for launder, ladle and furnace linings;
- pattern release agents and carriers •

The various stages are inter-related, with decisions at each step influencing the degree of freedom for other steps.

The primary consideration when selecting core and mould making techniques is the ability of the foundry to produce castings of the required quality at a competitive price. Quality requirements feed forward to mould and core manufacture, and ultimately to the raw materials to be used. This in turn may place restrictions on the proportion of recycled sand that can be used.

Similarly, the nature of the binders used will affect the fume released during casting, and also the ease with which sand can be removed and recycled.

When assessing an application, the Agency must assess BAT for the operation as a whole.

INTRODUC	TION	TEC	HNIQL	JES	EN	IISSION	IS	S IMPACT				
Management Main	terials Ac puts ab	tivities & atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Selection of raw materials	Applicati Questior	on Form 1 2.2 (part 1)		ldentify water th	the raw a at you p	and auxiliar ropose to u	y matei se.	rials, other s	substanc	es and		
	With the 1. supp inclu • t • t • t • t • a in ( A su Gen norm appr effec 2. justif whic BAT 3. for e certa	Application by a list of the iding: the chemical the quantitie the fate of the environment elevant spe any reasona impact include the substitue itable templ eric informa nally adequa oach to the environment eric informa nally adequa oach to the environment existing activa ain substance	n the Oper ne materia I composit s used, e material al impact of cies). bly practic ding, but n tion princip ate is inclu tion about ate rather i level of de ironment i ne basis of less haza	rator sho Is used, v ion of the (i.e. appl where kno able alter ot be limi ot be limi ot be limi ot be limi ot be limi able alter ot be limi sincluder internals than listin etail shou s includer impact of rdous alter the Oper	which have materials roximate pwn (e.g. mative ra- ted to, and e electro s, and gro g every o ld be use d. Produ- on produc ernative a alls in the rator belie	ve the potent s where relev percentages degradability w materials v y alternative nic version o puping inform commercial a d; ensuring t ct data sheet t quality), the ind that the p	ial for s vant; to each y, bioac which m s descr of this do nation o llternativ hat any ts shoul e contin propose mation, longer t	ignificant en n media and ccumulation p nay have a lo ibed in BAT ocument. f those of a s ve used. A c material cou ld be availab ued use of a d raw materi e.g. the env	vironment to the pro potential, ower envir Requirem similar typ common s uld have a le on-site ny substa ial section ironmenta	tal impact, oduct), toxicity to ronmental nent 5 below be, is sense a significant c. ance for n is therefore al impact of sh.		
BAT for selection	Indicative 1 The • c • c 2 The 3 The mate 4 The mate 5 The Mould m	e BAT Requ Operator sl complete an carry out any approved by Operator sl erials and th Operator sl erials. following ra aw materials	uirements nould: y longer-te y substituti the Regul nould mair nould have nould have nould have	erm studie ons ident ator ntain a de procedu entation c e quality a I substitu • sand • binde • catal • gase	es (Item 3 ified, as i tailed inv tres for th of any suit assurance tions sho er system yst	above), mprovement entory of raw e regular rev table ones w e procedures uld be applie Select	t conditi v materi view of r hich are for the d where tion tec	ions to a time ials used on- new develop e less hazard control of th e appropriate	escale to -site. ments in dous. e content e:	be raw : of raw		
	Pattern i carriers Degreas	release age sers	nts and	<ul> <li>use of Other as point</li> <li>wher chloridad</li> </ul>	of carrier erwise, no ossible. e solvent rinated so	organic solve n-chlorinated based de-gr lvents shoul	ents sho d solver reasing d be us	ould be avoid nts should be is necessary ed.	ded if pos e used as /, then no	sible. sparingly n-		

INTROD	UCTIO	N TEC	TECHNIQUES			EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		

Waste	2.2.2 Waste minimisation (minimising the use of raw materials)
minimisation Principles	The prevention and minimisation of waste and emissions to the environment is a general principle of IPPC. Operators will be expected to consider the application of waste minimisation techniques so that, wherever practicable, all types of wastes and emissions are prevented or reduced to a minimum. The steps below will also help to ensure the prudent use of natural resources.
	Waste minimisation can be defined simply as:
	"a systematic approach to the reduction of waste at source, by understanding and changing processes and activities to prevent and reduce waste".
	A variety of techniques can be classified under the general term of waste minimisation and they range from basic housekeeping techniques through statistical measurement techniques, to the application of clean technologies.
	In the context of waste minimisation and this Guidance, <b>waste</b> relates to the inefficient use of raw materials and other substances at an installation. A consequence of waste minimisation will be the reduction of gaseous, liquid and solid emissions.
	Key operational features of waste minimisation will be:
	<ul> <li>the ongoing identification and implementation of waste prevention opportunities;</li> </ul>
	<ul> <li>the active participation and commitment of staff at all levels including, e.g. staff suggestion schemes;</li> </ul>
	<ul> <li>monitoring of materials usage and reporting against key performance measures.</li> </ul>
	See Ref. 8 for detailed information, guides and case studies on waste minimisation techniques.
	Application Form Question 2.2 (part 2) <i>Identify the <u>raw and auxiliary materials, other substances</u> and water that you propose to use.</i>
	With the Application the Operator should
	1. Identify, from a knowledge of the plant, the main opportunities for waste minimisation and supply information on waste minimisation audits and exercises and the improvements made or planned.
	Indicative BAT Requirements
BAT for waste minimisation	<ol> <li>A regular waste minimisation audit should be carried out. Where one has not been carried out recently, an initial comprehensive audit should be carried out at the earliest opportunity within the improvement programme. New plants will need to have been operating for some time before an audit will be meaningful. Further audits should be at least as frequent as the IPPC Permit reviews. The audit should be carried out as follows:</li> </ol>
	<ul> <li>The Operator should analyse the use of raw materials, assess the opportunities for reductions and provide an action plan for improvements using the following three essential steps:</li> <li>i) process mapping;</li> <li>ii) raw materials mass balance;</li> <li>iii) action plan.</li> </ul>
	The use and fate of raw materials and other materials, including reactants, intermediates, by- products, solvents and other support materials, such as inerting agents, fuels, catalysts and abatement agents, should be mapped onto a process flow diagram (see Ref. 8) using data from the raw materials inventory (see Section 2.2.1) and other company data as appropriate. Data should be incorporated for each principal stage of the operation in order to construct a mass balance for the installation.
	In particular the audit should encompass:
	<ul> <li>the match between cast shape and final product;</li> </ul>
	<ul> <li>process control to maximise the lifetime of refractory material</li> </ul>
	Using this information, opportunities for improved efficiency, changes in process and waste reduction should be generated and assessed, and an action plan prepared for the implementation of improvements to a timescale approved by the Regulator.

2. See Sections 2.5 and 2.6 for identified areas of waste handling, recovery and disposal.

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		

### Water use

Reasons for reducing

water use

### 2.2.3 Water use

Water use should be minimised within the BAT criteria for the prevention or reduction of emissions and be commensurate with the prudent use of water as a natural resource.

Reducing water use may be a valid environmental (or economic) aim in itself, perhaps because of local supply constraints. In addition, from the point of view of reducing polluting emissions, any water passing through an industrial process is degraded by the addition of pollutants, and there are distinct benefits to be gained from reducing the water used, in particular:

- reducing the size of (a new) treatment plant thereby supporting the cost benefit BAT justification of better treatment;
- cost savings where water is purchased or disposed off to another party;
- associated benefits within the process such as reduction of energy requirements for heating and pumping, and reduced dissolution of pollutants into the water leading to reduced sludge generation in the effluent treatment plant.

The use of a simple mass balance for water use will reveal where reductions can be made.

Advice on cost-effective measures for minimising water can be found in ETBPP publications (see Ref. 9).

The amount of water used in foundries is small. Its main uses are: as a coolant; in sand reclamation, if a wet technique is used; as a scrubber medium in abatement plant; and from tumbling drums used for finishing small items.



INTROD	UCTIC	N TEC	HNIQU	JES	E٨	IISSION	IS	IN	ЛРАС	Т
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Water use										
BAT for water efficiency	2. 3. 4. 5.	<ul> <li>The following get</li> <li>water-efficient</li> <li>water should necessary. Which has a</li> <li>in particular, discharged st</li> <li>Measures should (see Section 2.3)</li> <li>To identify the sassociated with waters, should be treatment.</li> <li>Ultimately waster applications, the be usable in the can vary, it can the quality falls treated water from In particular, the individual procest</li> </ul>	eneral prin ht techniqu be recycl Where this lower wate uncontam eparately. d be imple 3.14). cope for s each use be kept se ewater will be best com- process of be recycle below that om the effl cost of miss s streams	ciples sh ues shoul ed within is is not pr er quality inated ro emented f ubstitutin should be parate wh need sol ventional directly or ed selectiv t which th uent trea embrane s or to the	ould be a d be used the proce acticable requirem of and su to minimis g water f e identifience there me form of effluent t when mi vely wher e system tment pla technolo e final effli	pplied in sec d at source v ess from whi , it should be ent; rface water, se contamina rom recycled d. Less con e is scope fo of treatment pro xed with frest the quality can tolerate nt could be u gy continues	quence where p ch it iss recycle which o ation ris I source taminat r reuse, (see Se oduces sh wate is adeq . The ( used an to redue effluer	to reduce em ossible; ues, by treat ed to another cannot be use k of process es, the water possibly after ection 2.3.12) a good water r. While trea uate, revertin Operator shou d justify whe uce. They can the treatment p	hissions t ing it first part of t ed, shoul or surfac quality re ears, e.g er some t . Howev quality v ted efflue ing to disc uld identi re it is no on be app plant. Ul	o water: if he process d be e water equirements g. cooling form of err, in many which may ent quality harge when fy where ot.
BAT (cont)	6. 7. 8. 9.	they could comp remains, however particularly when system could be benefits of provi- Water used in c • vacuuming, s • evaluating th • trigger contro The use of fresh • dilution of rest • wet scrubber Where a propos attention to the Fresh water cor - typically on a c	eletely repl er, a conce re waste h produced ding such leaning an scraping o be scope for ols on all h n water sho sin and oth r make up sal involves arguments asumption daily basis	ace the E entrated e eat is ava l. Where treatmen nd washin r mopping or reusing loses, ha ould be n her additi water wh s a wet p s put forw across th	TP plant effluent si ailable for appropria t. g down s g in prefe g wash wa nd lances ninimised ves nen recyc rocess fo rard by th ne foundr	, leading to a ream but, wi further treat ate, the Oper hould be min rence to hos ater; and washin It should or ed water can r reclaiming e operator fo y should be o	a reduce here thi ment by rator sh nimised sing dov g equip nly be u nnot be sand, th or consid	ed effluent vo s is sufficient y evaporatior ould assess by: wn; ment. sed for: used ne Regulator dering such a measured ar	will pay pa process	here and effluent and and as BAT ed regularly

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

# 2.3 The Main Activities and Abatement

(includes "directly associated activities" in accordance with the PPC Regulations)



INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

# 2.3.1 Storage and handling of raw materials

Summary of the activities	Process: Ray normally stor can be stored	v materials including fluxes in lump and powder form, deoxidants and refractories, are ed under cover. Following delivery, handling is kept to a minimum. Powdered materials I in sealed silos and conveyed pneumatically or kept and handled in sealed bags.					
Storage and	Sand is norm sands arrive	ally delivered in bulk and discharged directly to silo via pneumatic conveyor. Specialist n bags.					
handling of raw materials	Liquid binder are stored in storage. The catalysts and handled in a enclosed and	s, resins and catalysts are delivered in drums, or by bulk container or road tanker. They their delivery containers or, in the case of rail tankers, discharged direct into dedicated containers are connected by pipe directly to the sand/resin/catalyst mixing unit. Some co-reactants are used in a gaseous form but these are also delivered as liquids and similar fashion before being evaporated and mixed with a carrier gas. Evaporation is may be effected by a variety of methods.					
	Refractories,	release agents and other minor deliveries are stored indoors.					
Environmental impact	Water:	Not significant					
mpact	Land:	Spillage, overloading of silos and other containers					
	Air:	Dust and local odour.					
	Waste:	Refer to accidents.					
	Energy:	Not significant					
	Accidents:	Delivery to silos – excessive transfer rates of solids by pneumatic conveyors risks over pressurisation causing filter failure. Overfilling can cause spillage of liquids or powders. Charging materials into the wrong silo or tank can cause waste, spillage or uncontrolled chemical reaction.					
	Noise:	Delivery vehicles may cause nuisance especially if close to the site boundary.					
BAT for raw materials	Application Question 2.3	Form Storage and handling of raw materials					
<ul> <li>storage and handling</li> <li>With the Application the Operator should:</li> <li>1. supply the general application requirements for Section 2.3 listed on page 20 for this the activities;</li> </ul>							

### Indicative BAT Requirements

The main control issues relate to the potential for fugitive emissions. Refer to section 2.3.9

• Deliveries should be carried out in such a way so as to avoid spillage, leaks and dusty emissions. In particular, those arising from accidents during materials transfer.

INTRODUCTION		N TEC	TECHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.3.2 Launders

Summary of the activities	Process: Fou metal runs fro moulds.)	ndry launders are open channels lined with refractory material through which the molten m the furnace into the ladle. (In steelworks they are also used to direct metal into the
	Water:	Not significant
Environmental	Land:	Not significant
impact	Air:	Fume containing iron oxides.
	Waste:	refractory liners – sent to landfill
	Energy:	Not significant
	Accidents:	Not significant
	Noise:	Not significant
BAT for	Application F Question 2.3	Form Launders
launders	With the App	lication the Operator should:
	1. supply t the activ	he general application requirements for Section 2.3 listed on page 20 for this aspect of <i>i</i> ties;
	Indicative BA	AT Requirements
	The main con	trol issues are:
	Cont	aining fugitive emissions
	2.3.3 Des	sulphurisation of molten iron in ladles
Summary of the activities	Process: molt sulphur to pro	en iron can be desulphurised using magnesium or calcium carbide which reacts with the duce a slag which can be separated from the metal.
	A number of c iron and de-su thereby prome	different methods can be used. In the "shaking" process a barrel shaped ladle containing ulphurising agent is shaken to swirl the contents around the inner lining of the ladle and ote mixing.
	In the Continu slag of lime a stirred by inje bottom of the compartment, pouring from steadier emis	uous Divided Ladle process the iron is tapped continuously from the cupola, through a nd calcium carbide, into the first compartment of the divided ladle. The compartment is cting nitrogen into the base of the ladle through a porous plug. Iron which reaches the compartment will be de-sulphurised and passes under the dividing wall into the second from where it flows into a transfer ladle, driven continuously by the head of the hot metal the furnace. Because of its continuous nature the divided ladle process produces a sion of fume.
Environmental	Water:	Not significant
Impact	Land:	Slag
	Air:	Dust and local odour.
	Waste:	Slag
	Energy:	Not significant
	Accidents:	Electrical failure during process operations is likely to generate fume and some scrap.
	Noise:	Not significant

INTRODUCTION		N TEC	TECHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

BAT for desulphurisation of molten iron in ladles	Application Question 2.3 <i>With the App</i> 1. supply the action <i>Indicative B</i> The main con • Containin	Form 3 (cont.) Desulphurisation of molten iron in ladles <b>Desulphurisation of molten iron</b> in ladles <b>Desulphurisation the Operator should:</b> The general application requirements for Section 2.3 listed on page 20 for this aspect of vities; <b>AT Requirements</b> htrol issues are: g the fume emissions.
Nodularisation	2.3.4 No	odularisation of SG iron in ladles
Summary of the activities	Process: Cor accompanieo nodularisatio	nversion of flake grey iron into ductile iron (nodular or spheroidal graphite iron) is I by an improvement in mechanical properties. Such conversion is referred to as n and involves the treatment of the liquid metal with magnesium and rare earth elements.
	There are a r pour over, plu methods resu	number of methods of introducing the nodularising agent into the metal and these include unging, pressure vessel, tundish cover ladle, in-stream and in-mould techniques. Some ult in the evolution of large volumes of fume whilst others produce relatively little.
Environmental	Water:	Not significant
Impact	Land:	Not significant
	Air:	Fume.
	Waste:	Slag
	Energy:	Not significant
	Accidents:	Electrical failure during process operations is likely to generate fume and some scrap.
	Noise:	Not significant
	Application Question 2.3	Form 3 (cont.) Nodularisation of SG iron in ladles
	With the Ap	plication the Operator should:
	1 supply the acti	the general application requirements for Section 2.3 listed on page 20 for this aspect of vities;
	Indicative B	AT Requirements

The main control issues are:

• Containing the emissions of fume

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

### 2.3.5 Preparation of cores and moulds

Summary of the activities

Many different methods of mould manufacture have been developed. Table 3 lists the various techniques together with potential emissions released during production and casting. Some of the most common are described below:-

### The green sand method

A schematic of the process is shown in *Figure 1.* This uses a mixture of sand, an organic component such as dextrose or coal dust (optional), a clay binder and water which is moulded around a pattern within an open box. It is then mechanically compacted. The main emission is dust released during mixing which is captured and filtered. When the organic components are used there is also a release of VOCs and oxides of carbon during casting operations. In common with all sand moulding techniques there is a significant waste disposal problem.

### The shell sand method

This involves immersing a hot metal pattern into a box of sand pre-coated with a novolac resin. The resin cures and binds the sand together. The longer the invest time the thicker the mould. The method is used for mass production of relatively small castings.

Fumes are emitted during production and during cooling. The fumes are normally extracted and discharged to atmosphere uncleaned. The emissions are primarily VOCs.

### Cold setting techniques

These rely on the chemical reaction between a resin and a hardener, possibly in the presence of a catalyst. There are several basic systems although minor variations may be applied:-

- <u>furan</u> a phenolic urethane formaldehyde resin blended with furfuryl alcohol which is hardened by mixing with sulphonic, sulphuric or phosphoric acid. The moulds or cores are self setting.
- <u>alkaline phenolic</u> a phenolic alkaline resin which is hardened by mixing with a liquid ester. It is
  also possible to use a gaseous ester carried in a methyl formate vapour.
- <u>phenolic urethane</u> a blend of phenolic urethane formaldehyde resin with polymeric polyisocyanate. This is also available as a gas hardened system using an amine gas.
- <u>sodium silicate</u> sodium silicate is hardened by mixing with an ester. Gas hardening can be achieved by replacing the ester with CO<sub>2</sub> gas.

Fans are sometimes used to disperse fumes which are evolved during mixing and curing, collection and extraction being rarely employed. However, when using amine gas curing it is necessary to collect and scrub the exhaust gases prior to discharge. Methyl formate is flammable and the releases must either be condensed for reuse or abated, possibly via an afterburner.

### Mould Coating

Iron moulds may be coated with a powdered refractory material to protect the mould surface and to reduce the chilling effect of the iron. The coating is water based and is usually applied by spray under local fume extraction hoods leading to bag filters.

Steel moulds or dies are not coated but are externally water cooled. They are used for pipe casting where the limited thickness of the pipe wall does not have a significant impact on the steel mould temperature and the pipe quickly contracts away from the mould to be ready for withdrawal. Details of components such as the socket ends of pipes or the spigot ends of rolls are formed by sand submoulds within the main mould.

Sand moulds are often coated with a refractory material to protect the mould surface and to improve the surface finish of the casting. The coating may be based on either water or alcohol and may be applied using spray, flow coating, dipping, brushing or swabbing.

### **Oil Sand Process**

This technique makes use of an oil which is polymerised by heating in an oven. Larger installations may require hoods and abatement plant to collect and destroy associated VOC emissions.

### Pattern release agents and carriers

These are used in many mould applications. Typical carrier solvents are 1,1,1 trichloroethane, hydrocarbons and alcohols. Chlorinated solvents tend to produce better results than hydrocarbons and alcohol blends which are inferior in evaporation rates, cleaning ability and flammability. Some non-solvent release agents are available but are difficult to apply and expensive. Release coatings are normally applied under extraction hoods.

			IES				IN		т
INTROD	Materials Activities				1100101		11		/ I
Management	inputs abatemer	t water	Waste	Energy	Accidents	Noise	Monitoring	Closure	issues
Environmental impact	Water: No Land: Not Air: Th - Odou - Dust - Resir - VOC - Emis - Fume	: significant significant e most import r; – the sourc from sand ha handling; emissions ind sions of oxide	tant issue ces of whi indling; cluding ar es of carb	s from mo ch are the nines, ald on;	ould and core gases men lehydes and	e prepa tioned t phenol	ration and str below	orage ar	e as follows:
	- Leak proce - Leak dime from - Emis tetrar - Partie <b>Waste:</b> Cho <b>Energy:</b> Not <b>Accidents:</b> Mix	of gases be sses, such a of gases be hylethylamin he cores in s sions of amm nine which is culate emissio ice of binder significant ng and blenct	ing used a s sulphur ing used e (DMEA) storage; ionia arisi a catalys ons. system a ding proble	as a gass dioxide; as catalys b. Both ga ng from th t used in t ffects pot ems can i	ing agent in sts, such as t ses have un ne thermal do the shell pro- ential for rec mpact upon	certain triethyla pleasar ecompc cess; cycling a recover	mould and c mine (TEA) and todours. Th osition of hexa and recovery. ry	oremakir and ese gase amethyle	ng es exude ene
BAT for preparation of cores and moulds	<ul> <li>Application Form Question 2.3 (constrained)</li> <li>With the Application</li> <li>1. supply the generative set of the activities</li> <li>Indicative BAT R</li> <li>Minimise constructions techniques de Chemical Bind Programme.</li> <li>Control amine odour nuisand mainly requires</li> <li>Where gas-fire paid to good of the set of t</li></ul>	nt.) tion the Ope eneral applic ; equirements umption of bi scribed in Gu ers in Found emissions fro e, using chen d where DME ed heating sys- leaning and r mould coatir scrap moulds organic solve ot possible, r	Prepara moulds prator sho ation requ ation requ idance No ries, prod om gasse nical scru EA is used stems are naintenar ngs should s and core ents for pa	ation of co buld: uirements nicals thro be GG10 uced by the bbing, inc d phenolic bbing, inc d be torch es should ttern rele nated solv	for Section 3 ough good p 4 (see Ref 2 he Environm c urethane s ineration or sin shell or c ner systems and off as so be segregat ase agents c	2.3 liste process 26), Cos hental Tr ystems an effec bil sand) on as it ted from be emp	d on page 20 control, using t Effective M echnology Be where neces ctive odour n b, particular a is safe to do n other waste where should ployed and th	o for this g the who anageme est Pract ssary for eutralise ttention s so after to facilit be avoid	aspect of ole range of ent of ice control of r. This is should be application. ate ed. If nt used

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Releases to the environment commonly associated with the processes described are listed in Table Table 2-1 release routes.

### Table 2-1Fumes evolved from current foundry processes

System Name and Binder Constituents	Setting Method	Fumes during Mixing and Setting	Fumes during Casting
<b>GREEN SAND</b> Clay Coal dust or substitute Water	Pressure	Dust	Carbon oxides Aromatics (inc polycyclics) Nitro aromatics
<b>SHELL SAND</b> Phenol Formaldehyde (Novalak) Resin	Heat	Formaldehyde Ammonia Phenol Aromatics	Carbon oxides Phenols Ammonia Aldehydes Aromatics (inc polycyclics)
ALKALI PHENOLIC Alkaline phenol Formaldehyde resin			
1. Self-setting, eg "Alphaset", "Novaset"	Cold set with esters	Formaldehyde Phenol Esters	Carbon oxides Formaldehyde Phenol Aromatics
2. Gas hardened, eg "Betaset"	Gas hardened with methyl formate vapour	Formaldehyde Phenol Methyl formate	, ionaliss
PHENOLIC URETHANE 1. Gas hardened, eg "Coldbox", "Isocure" 2. Self setting, eg	Amine vapour Self set with substituted	Solvents Isocyanate (MDI) Amine Solvents	Carbon oxides Nitrogen oxides Monoisocyanates Formaldehyde Phenol Aromatics (inc polycyclics)
"Novathane", Pepset"	pyridine	Isocyanates (MDI)	Anilines Naphthalenes Ammonia
<b>FURANE</b> Combination resins of: Phenol Urea Furfuryl alcohol Formaldehyde	Cold set with acids	Formaldehyde Phenol Furfuryl alcohol Hydrogen sulphide Sulphur dioxide Acid mists	Carbon oxides Phenol Formaldehyde Aromatics Sulphur dioxide Ammonia Aniline
HOT BOX Combination resins of: Phenol Urea Furfuryl alcohol Formaldehyde	Heat	Formaldehyde Acids Furfuryl alcohol Phenol	Carbon oxides Nitrogen oxides Formaldehyde Phenol Aromatics Aniline Ammonia
OIL SAND Linseed oil and starch	Heat	Acrolein Complex organics	Carbon oxides Butadiene Ketones Acrolein
CO <sub>2</sub> PROCESS Sodium silicate	Gas hardened with CO2 gas	None	Carbon oxides
SILICATE ESTER "Self set" Sodium silicate	Cold set with esters	Esters	Carbon oxides Alkanes Acetone Acetic acid Acrolein

INTRODUCTION		N TEC	TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Casting,	2.3.6 Casting, pouring or moulding
pouring or moulding	Fume generated during casting arises from two sources. Fine iron oxide is generated at the surface of the molten metal as it is poured, and the organic products are expelled from the mould as the resins

the molten metal as it is poured, and the organic products are expelled from the mould as the resins and binders decompose. The former only occurs as the metal is being poured, whilst smoke and fumes caused by decomposition of the binder will continue to be evolved as the mould cools. Whilst COSHH assessments can be of value in assessing the environmental impact of fugitive

Whilst COSHH assessments can be of value in assessing the environmental impact of fugitive releases, they should be interpreted with care. The high temperatures involved make the emitted fume very buoyant, and significant environmental releases can occur even though the work areas remain clear of fume.

Where moulds are water-cooled or no sand/binder systems are used, casting fume production is small and extraction, other than for hygiene requirements, may not be needed. Concentrations of pollutants will be low and arrestment will probably not be needed.

Table 3 of IPR 2/2 identifies the type of fume that will be released into the atmosphere from mould and core preparation, curing and subsequent casting.

### **Casting Practices**

### Static Sand Casting

Summary of the

activities

This is the simplest technique whereby the sand moulds are arranged on the shop floor and filled from a ladle. The castings are then left to solidify. It is generally impracticable to have fixed extraction hoods and ductwork in the casting area. In more automated foundries the moulds may be moved by conveyor into the pouring position where local extraction can be employed.

### Pipe Casting

Pipe casting is carried out by pouring hot metal from a ladle, via a tundish and runner into the rapidly rotating horizontal mould. The metal is forced out on to the inner cooling surface where it solidifies. The mould is stopped and the casting withdrawn.

### **Roll Casting**

This can be carried out centrifugally or statically. In the centrifugal system the vertical mould is placed in a machine and spun at high speed while the hot metal is bottom poured into it. Some time after filling, the mould is lifted from the machine, and is left for several days before stripping. Fume extraction is normal on the machine although not in the cooling pit.

In the static system the mould is mounted vertically and bottom poured from the ladle. It remains stationary while it cools and solidifies. Typically no fume extraction is required.

Environmental	Water:	Not significant		
mpaci	Land:	Not significant		
	Air:	Fume.		
	Waste:	Slag		
	Energy:	Not significant		
	Accidents:	Not significant		
	Noise:	Not significant		

**IPPC** 

INTRODUCTION		N TEC	TECHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues



### With the Application the Operator should:

BAT for casting, pouring or moulding 1. supply the general application requirements for Section 2.3 listed on page 20 for this aspect of the activities;

### Indicative BAT Requirements

- For certain types of casting operation, e.g. automotive castings and greensand foundries, and where a relatively large number of similar castings are being manufactured, BAT will normally include the use of a fixed pouring station with the moulds moving past on a conveyor belt system. Where possible, moulds should be totally enclosed or fully enclosed casting machines used. This will enable all the casting fume to be extracted efficiently and treated. Where there are significant fumes emitted after pouring then the conveyor should be enclosed and extracted.
- For large items such as rolls, machine tool beds, etc., it may be necessary to carry the hot metal in a ladle to a mould in a casting pit or casting bay. Here a movable or extendable extraction hood connected to fixed arrestment plant installed in the most advantageous position to collect casting fume should be considered.
- Where suction hoods are used, these should be placed as close to the sources of fume as possible to reduce dilution of the fumes caused by large volumes of air being drawn into the hoods. Suction hoods should not hinder process operations or compromise safety; considerations should be given to push-pull systems to improve efficiency.
- The effectiveness of hoods and extraction systems can be assessed using photographic systems to record the movement of emitted fume. Where optical photography is not practicable, infrared imaging may be an effective alternative.
- Part of the improvement programme to update operations should include a quantitative assessment of the emissions so that a further improvement programme can be established to reduce them if possible within BAT considerations. This is particularly important where an odour problem has occurred.

INTRODUCTION TEC		HNIQUES		EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.3.7 Knockout and reclamation

Knockout and reclamation	Knocking out be removed from the box operations is	t or stripping is the practice of removing the casting from the mould. Smaller castings may from the casting box manually or by use of a vibrating table which dislodges the casting and allows any sand to fall through the open surface of the table. Dust from these usually collected and cleaned by bag filters.							
	Large mould excavator. A impractical. knocking-out landfill.	s will be broken up and castings removed in-situ, using an overhead crane or bucket Ithough large quantities of dust may be generated, local extraction and filtration is Pipes are removed from their rotary moulds by mechanical means. The sand from the area is excavated, normally by mechanical digger and is either recycled or disposed of to							
Environmental	Water:	Not significant							
Impact	Land:	Contaminated sand							
	Air: During knockout the mould is broken open and the surface area from which compounds may be liberated is significantly increased. Pyrolysis products a the resin coated sand, such as phenols for example, volatilise, and are emit foundry atmosphere. Formaldehyde will be present in the knockout section the resin binder. Dust is emitted on which organic compounds may be adsord								
	Waste:	Contaminated sand							
	Energy:	Not significant							
	Accidents:	Not significant							
	Noise:	Many parts of the machine are very noisy and require acoustic shelters for worker protection. Standard noise protection measures should be taken to minimise disturbance in the local neighbourhood.							
	Application Question 2.3	Form 3 (cont.) Knock out and reclamation							
	With the Ap	plication the Operator should:							
	1. supply th activities	ne general application requirements for Section 2.3 listed on page 20 for this aspect of the s;							
	Indicative B	AT Requirements							
BAT for knock out and reclamation	Cooling of th <ul> <li>Knockou</li> <li>sufficien</li> </ul>	e mould before knockout reduces the mass of organics released. It area should be enclosed and connected to arrestment plant. (cyclones not considered t to be BAT).							
	<ul> <li>Knockoι</li> </ul>	it should not be done by hand.							
	<ul> <li>Fixed va</li> </ul>	cuum cleaning system around knockout area.							

INTROD	NTRODUCTION TEC		HNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Sand reclamation	2.3.8 Sa	nd reclamation							
	Used moulds castings and	can be broken down in ball mills and impeller or vibrating crushers. In some cases the sand are put into vibratory crushers together in order to separate the castings from the sand.							
	An alternative mixture in attr dried and reg	method of reclaiming mould sand is by mixing with water and scrubbing the sand/water ition cells using abrasion resistant high speed paddles. The sand must then be de-watered, raded. This method leads to an aqueous stream, which is avoided by other methods.							
Summary of the activities	Thermal recla mechanical re 800EC. Units ensures that bag filtration I	mation is used primarily for organic binder systems. It is used as a second stage, after eclamation, and achieves a high reclamation rate. The waste gases are burnt at 750EC to s of up to 5 tonnes/hr are in use usually employing a fluidised bed. An after-burning zone combustible gases are completely oxidised. After cooling, the gases should be cleaned by before release to atmosphere.							
	Two other teo first involves p 3,000EC for a the sand. Th	Two other techniques are currently at the pilot plant stage and may be commercially available soon. The first involves plasma technology whereby the sand passes through a plasma arc with temperatures up to $3,000EC$ for a few milliseconds which volatilises the organic binders without altering the composition of the sand. The VOCs are ionised and, as the gas cools, react with the air to form CO <sub>2</sub> and water.							
	In a second d turbulent mixi reduced ener	evelopment the use of a toroidal fluidised bed reactor which achieves a high degree of ng and high processing rates and is also expected to yield a lower capital cost and gy requirements compared with a conventional fluidised bed unit.							
Environmental	Water:	Not significant;							
Impact	Land:	Contaminated sand, fugitive dust;							
	Air:	Fume, products of combustion;							
	Waste:	Contaminated sand;							
	Energy:	Afterburners are significant energy users, and temperature settings should be the minimum compatible with acceptable emission control;							
	Accidents:	Not significant;							
	Noise:	Many parts of the machine are very noisy and require acoustic shelters for worker protection. Standard noise protection measures should be taken to minimise disturbance in the local neighbourhood.							
	Application I Question 2.3	Form Sand reclamation 3 (cont.)							
	With the App	olication the Operator should:							
	1. supply t the acti	the general application requirements for Section 2.3 listed on page 20 for this aspect of vities;							
BAT for sand reclamation	Indicative B The main cor • Use of aff	<b>AT Requirements</b> htrol issues are: erburner followed by bag filtration of the waste gases when thermal reclamation used							

• Effective and reliable temperature control systems on the afterburner
INTRODU	CTION	TEC	HNIQL	JES	EN	IISSION	IS		MPAC	Т
Management <sup>N</sup>	laterials Action inputs aba	vities & tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Fettling,	2.3.9 Fe	ettling,	dressir	ng or fi	nishin	g of cast	ings			
dressing or finishing of castings	After the ca the final fini dressing ind	stings hav sh require clude:-	ve cooled t d for the p	hey are s roduct. 1	ubjected hese ope	to a number erations, son	r of finis ne of wi	hing process nich are refe	ses in ord rred to as	er to obtain fettling or
	• Cl	eaning by	shotblastii	ng or othe	er means	to remove c	ore and	mould mate	rials and	scale.
Summary of the activities	• Re su	emoval of o perfluous	excess me metal.	tal such a	as feeder	heads, runn	ier or ga	ating systems	s and any	/ other
	• Re	emoval of l	blemishes	and defe	cts.					
	<ul> <li>Sn are</li> </ul>	noothing c eas on the	over of well surface o	dments, a f the cast	areas fror ing, gene	n which meta rally be grine	al has b ding.	een cut, and	any othe	er rough
	Fettling is g generation ceramic chi pipes the in pipe. Anoth	enerally a of dust an ps. This is iternal surf ner metho	chieved by d fume. S s usually c face is dre d uses an	y flame cu mall item arried ou ssed by e electric a	utting, gri s may be t in water extending rc to sele	nding or chis finished by to which su a rotating gi ctively reme	elling, a grinding rfactant rinding It unwar	and usually r g in tumbling s may be ado wheel or burn nted small ar	esults in drums to ded. In the r the full I reas of th	the gether with he case of ength of the e casting.
Environmental	Water:	Sludge	e where we	et techniq	ues are ι	ised.				
mpaci	Land:	Fugitive	e dust							
	Air:	Dust ar	nd Fume.							
	Waste:	Collect	ed dust							
	Energy:	Not sig	nificant							
	Accidents:	Not sig	nificant							
	Noise:	Many of protectio in the loc	these proc n. Standa al neighbo	esses are rd noise p ourhood.	e very no protectior	isy and requ n measures s	ire acou should b	ustic shelters be taken to m	for work ninimise o	er listurbance
	Application Question 2	n Form 2.3 (cont.)	>	Fettling castings	, dressing S	g or finishing	of			
BAT for fettling	With the A 1. suppl the ac	<b>pplicatior</b> y the gene ctivities;	n the Oper eral applica	r <b>ator sho</b> ation requ	<i>uld:</i> irements	for Section 2	2.3 liste	d on page 20	) for this	aspect of
dressing or finishing of castings	Indicative Contain Effective	<b>BAT Requ</b> ment and e means c	<i>uirements</i> extraction f detectior	n for filter	failure					

INTRODU	JCTION	CTION TECHNIQUES EMISSIONS IMPACT											
Management	Materials Activ inputs abat	ities & ( ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
Waste handling and recycling facilities Summary of the activities	2.3.10 Wa The most im • Slag from • Collected • Collected • Refractor	aste han portant issu n ladles d dust d sludge ry waste	ues from	and re waste ha	andling ar	g facilitie	<b>es</b> facilities	s are as follo	ws:				
Environmental impact	Water: Land: Air: Waste: Energy: Accidents: Noise:	Not signi Slag, con Fume. Slag Not signif Not signif	ificant Itaminate ficant ficant ficant	ed sand									
BAT for waste	Application Question 2. <i>With the Ap</i> 1. supply the act	Form 3 (cont.) pplication t the genera tivities; BAT Requir	the Oper al applica	Waste h facilities rator sho tion requ	andling a uld: irements	and recycling	2.3 liste	d on page 20	) for this	aspect of			
handling and recycling facilities	The main co   Water ef   Waste sh No further is	ntrol issues ficiency tec nould be red sues are id	s are: hniques covered entified.	should be - see Sec	e employ ction 2.6.	ed - see Sec	ction 2.3	3.14.					
Emissions to air	<ul> <li><b>2.3.11 Co</b></li> <li>The nature a will be confir</li> <li>SO<sub>X</sub>, NO</li> <li>particulation</li> </ul>	and source med in deta x, and CO <sub>x</sub> tes from sa	of the en ail in the from mo nd handl	SOURC nissions e Operator ost found ling, knoc	e emis expected 's respon ry operati king out,	sions to from each a se to Sectio ons; finishing op	<b>air</b> activity is on 3.1. I erations	s given in pre In general the s and sand re	evious se ey compr eclamatio	ctions and ise: n;			
Sources	<ul> <li>fume from from mix and sand</li> <li>solvents</li> <li>odorous casting, p</li> </ul>	om launders a, hydrogen ing and cur d reclamatic e.g. from ca compounds pouring and ral guidance	s and ref sulphide ing of bo on. arriers in s from re d mouldir e on aba	ining ope e, hydrog nding sys formulat fining ope ng, from k tement te	rations; en cyanic stems in i ed chemi erations, cnocking echniques	de, phenol, a mould and c cals (e.g. re from mould a out and from s for point sc	amines, ore pre lease ag and cor n sand r ource er	other VOCs, paration; also gents); e preparatior eclamation. nissions to a	, and acic o from kn n and sto ir can be	l vapours ocking out rage, from found in			

INTROD	UCTIC	N TEC	HNIQL	JES	EN	IISSION	IS		MPAC	Т				
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
Abatement to air <i>cont.</i>	App Que	blication Form estion 2.3 (cont.)	Х	Control c	of Point S	ource Emiss	ions to	Air						
	With	the Application	the Ope	rator sho	uld:									
	1.	supply the gene equipment; and	eral applica in additio	ation requ n	iirements	for Section	2.3 on p	bage 20 for c	control an	d abatement				
	2.	2. describe measures taken to increase the reliability with which the required control and abatement performance is achieved.												
	3.	<ol> <li>where VOCs are released, identify the main chemical constituents of the emissions and provide an assessment of the fate of these chemicals in the environment.</li> </ol>												
	Indi	cative BAT Requ	uirements	;										
	1.	The Operator s (see item 2 in S Regulator but ir	hould com section 2.3 n any case	plete any ) as an in within th	detailed nproveme e timesca	studies requent condition ale given in S	uired into to a tim Section	o abatement nescale to be 1.1;	t or contro e agreed v	l options with the				
	2.	<ol> <li>Steam plume elimination. Releases from wet scrubber vents should be hot enough to avoid visible plume formation in the vicinity of the vent. This is to prevent the condensation or adsorption of environmentally harmful substances by the condensing water vapour. Exhaust gases from a wet scrubber can be heated by the use of waste heat to raise the temperature of the exhaust gases and prevent immediate condensation on the exit from the vent. This procedure also aids the thermal buoyancy of the plume. Where there is no available waste heat and the vent contains no significant environmentally harmful substances, the Applicant may be able to demonstrate that the BAT criteria have nonetheless been met.</li> </ol>												

# 2.3.12 Abatement of point source emissions to surface water and sewer

Effluent treatment

The nature and source of the emissions expected from each activity is given in previous sections and will be confirmed in detail in the Operator's response to Section 3.1. In general, wastewater can arise from the process activity, from storm water, from cooling water, from accidental emissions of raw materials, products or waste materials, and from fire-fighting. In addition to the BREF and the techniques below, guidance on cost-effective effluent treatment techniques can be found in ETBPP Guides (Ref. 9).

Summary of the activities

The nature and source of the effluent from each activity in foundries is given in the preceding sections of 2.3. The amounts of process water used in foundries is small and comprises principally:-

- Discharge and blowdown from wet scrubbers;
- Cooling water, often containing biocides and anti-oxidants;
- Discharge from wet sand reclamation plant;
- Drum tumbling discharge containing metals and surfactants;
- Site drainage and stormwater;
- Leachate from slag and waste tips.

A wide variety of chemicals are also used in the core and mould preparation process and the effluent from reclamation will be a complex mixture.

INTRODUC	CTIO	N TEC	HNIQL	JES	E٨	<b>/ISSIO</b>	٧S	11	MPAC	Т
Management Ma	terials puts	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Effluent treatment	Appli Ques	ication Form stion 2.3 (cont.)	$\geq$	Effluent	Treatme	nt				
	With	the Application	the Ope	rator sho	uld:					
	1.	supply the gene	eral applications to wate	ation requ	irements	for Section	2.3 on	bage 20 to pi	revent or	reduce point
	2.	include, where a system for the a	appropriat activity;	e, off site	treatmer	nt in the des	cription	of the waste	water trea	atment
	3.	provide, where which it can be	effluent is reused (e	discharge .g. by ultra	ed, a justi afiltration	ification for where app	not clea ropriate)	ning the efflu ;	ent to a l	evel at
	4.	describe measu performance is ensure that they	ires taken delivered y are conti	to increa (heavy m rolled all t	se the rel etals are he time?	liability with measured etc.);	which th only occ	ne required c asionally – w	ontrol and hat techr	d abatement iiques
	5.	identify the main and assessmen out as in respor demonstrate that	n chemica It of the fa Ise to Sec at the cont	I constitue te of these tions 3.1 trols are a	ents of th e chemic and 4.1 b idequate.	e treated et als in the et out need to This appli	fluent (ir nvironme be under es wheth	ncluding the react of the react. These s rstood here in the reatment	make-up teps will t n order to t is on- or	of the COD) be carried off-site;
	6.	identify the toxic guidance is ava improvement pr	city of the ilable, this ogramme	treated ef s should, t ;	fluent <mark>(se</mark> unless alr	ee Section 2 ready in har	2.10). U nd, norm	ntil the Regu ally be carrie	lator's tox d out as	kicity part of an
	7.	where there are toxicity and the	harmful s technique	substance s propose	s or level ed to redu	ls of residua	al toxicity ential imp	/, identify the bacts;	causes	of the
	8.	consider whethe Water Treatmer	er the efflu nt Directive	uent flow i e.	s sufficie	nt to fall wit	hin the r	equirements	of the Ur	ban Waste
	Indic	ative BAT Requ	uirements	;						
BAT for effluent	1.	The Operator sl (see item 2 in S Regulator but in	hould com ection 2.3 any case	plete any ) as an in within th	detailed nproveme e timesca	studies req ent condition ale given in	uired int n to a tim Section	o abatement nescale to be 1.1;	or contro agreed v	l options with the
	2.	The following ge	eneral prir	nciples sh	ould be a	pplied in se	quence	to control en	nissions t	o water:
		<ul> <li>water use sh</li> <li>contamination</li> </ul>	ould be m	ninimised	and wast	ewater reus	sed or re	cycled (see simised (see	Section 2 Section 2	.2.3); 2.3.14):
		<ul> <li>ultimately, su statutory and as treatment should be us and alkaline streams, whi</li> </ul>	urplus wat d non-stati will be mo ed where streams. ile dilution	er is likely utory obje ore efficie possible Also, biol , by mixin	v to need ctives). ( nt. Howe to avoid a ogical tre g stream	treatment t Generally, e ever, the pro- adding furth eatment car s, can assis	o meet ti effluent s operties er chem occasio at treatm	he requireme treams shou of dissimilar icals, e.g. ne nally be inhil ent;	ents of BA Id be kep waste str eutralising bited by c	AT (and t separate eams waste acid concentrated
		<ul> <li>systems sho</li> </ul>	uld be eng	gineered f	to avoid e	effluent by-p	assing t	he treatment	plant.	
	3.	All emissions sh (see Sections 3 reasonable cost this will be carried	nould be c .2 and 5.1 t it should ed out in r	ontrolled, ) but notin do so (se response	as a min ng that w e Section to Section	imum, to av here BAT c n 1.1). Calc n 5.1.	void a br an delive sulations	each of wate er prevention and/or mode	r quality s or reduc elling to d	standards tion at emonstrate
	4.	With regard to E IPPC the prever be made at reas water, the adeq substances mus found in Refere	BOD, the r ntion or re sonable co uacy of th st also be nces (see	nature of t duction o ost should e plant to considere Ref. 12).	the receiv f BOD is l be carrie minimise ed. Guida	ving water s also subjec ed out. Fur e the emissi ance on trea	hould be t to BAT thermore on of sp atment o	e taken into a and further i e, irrespective ecific persist f persistent s	reduction e of the re ent harmf substance	However, in s which can eceiving ful es can be
										Cont.

Management         Materials inputs         Activities & abatement         Ground water         Waste         Energy         Accidents         Noise         Monitoring         Closure         Installation issues           Effluent treatment Cont.         5.         Where effluent is treated off-site at a sewage treatment works, the above factors apply in particular demonstrating that:         •         the treatment provided at the sewage treatment works is as good as would be achieved if the emission was treated on-site, based on reduction of load (not concentration) of each substance to the receiving water;         •         the probability of sewer bypass, via storm/emergency overflows or at intermediate sewage pumping stations, is acceptably low;         •         action plans in the event of bypass, e.g. knowing when bypass is occurring;         •         a suitable monitoring programme is in place for emissions to sever, taking into consideration the potential inhibition of any downstream biological processes and actions plan for any such event.           6.         Minimising the use of water and minimising the level of pollutants in each water stream are the primary aims followed by the recycling of waste water streams wherever possible. In this industr cleaning the water to 35 mg/l of suspended solids is likely to ensure that most of the insoluble pollutants will be within their normal limits. Such cleaned water would generally be of good quality and should be considered for recycling - however it is also important to consider the level of dissolved chemicals before concluding that recycling of the water is viable.           7.         For furnace gas cleaning, dry filtration systems eliminate potential waste	INTROD	UCTIO	ON TEC	HNIQU	JES	EM	IISSION	IS	I	MPAC	Т
<ol> <li>Effluent treatment Cont.</li> <li>5. Where effluent is treated off-site at a sewage treatment works, the above factors apply in particular demonstrating that:         <ul> <li>the treatment provided at the sewage treatment works is as good as would be achieved if the emission was treated on-site, based on reduction of load (not concentration) of each substance to the receiving water;</li> <li>the probability of sewer bypass, via storm/emergency overflows or at intermediate sewage pumping stations, is acceptably low;</li> <li>action plans in the event of bypass, e.g. knowing when bypass is occurring; rescheduling activities such as cleaning or even shutting down when bypass is occurring;</li> <li>a suitable monitoring programme is in place for emissions to sewer, taking into consideration the potential inhibition of any downstream biological processes and actions plan for any such event.</li> </ul> </li> <li>6. Minimising the use of water and minimising the level of pollutants in each water stream are the primary aims followed by the recycling of waste water streams wherever possible. In this industric cleaning the water to 35 mg/l of suspended solids is likely to ensure that most of the insoluble pollutants will be within their normal limits. Such cleaned water would generally be of good quality and should be considered for recycling - however it is also important to consider the level of dissolved chemicals before concluding that recycling of the water is viable.</li> <li>7. For furnace gas cleaning, dry filtration system seliminate potential waste wate streams, howeve with a wet cleaning system there is no need for an aqueous discharge if appropriate measures are taken to clean the water and recycle it. As long as solids are removed to a level acceptable to the scrubbing device solubles can normally be allowed to reach saturation, hydrocyclones, sand filtration, filter pressing.</li> <li>8. Having taken all steps to minim</li></ol>	Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
<ul> <li>environment, the liquor being cleaned and recycled. For an existing plant every effort should be made to prevent and minimise the waste water stream.</li> <li>9. Run-off from all open areas, but in particular from raw materials stocking areas, will contain suspended solids which will have to be removed by settlement or other techniques. Oil interceptors may be necessary in drainage from scrap handling areas. Drainage sumps should be of sufficient size to handle storm water and should be designed to accommodate storm surge in order to prevent carry-over of unsettled material.</li> <li>10. Bunding is a sensible precaution in all but the most trivial cases eg. dilute non-hazardous aqueous solutions. It is essential in many cases where there is a risk to controlled waters, sewers and drains, and on-site effluent treatment plants. Shared bunds are possible in cases where the materials stored are not incompatible. Bund capacities should always exceed the volume of the largest storage by a minimum of 10%.</li> <li>11. High level alarms and trips on storage tanks and bunds should also be regularly inspected, particularly where corrosive substances are involved, requiring well managed and documented inspections. Procedures for preventing the unauthorised discharges or leakages from bunds should be in place. Where it is considered inappropriate to bund a particular storage tank or process vessel then the applicant must justify this approach.</li> </ul>	Effluent Cont.	<ol> <li>5.</li> <li>6.</li> <li>7.</li> <li>8.</li> <li>9.</li> <li>10.</li> <li>11.</li> </ol>	<ul> <li>Where effluent particular demo</li> <li>the treatment emission was substance to</li> <li>the probability pumping station activities such activities and activities are activities and activities are taken to clear agents, VOCs or would also be a environment, the made to prevent and agents are to prevent at agents are to prevent at a are to prevent and the prevent at a are to prevent at a are the are are to prevent at a are the are are are to prevent at a are the are are are are to prevent at a are the are are are are are are are are are ar</li></ul>	is treated instrating to the provided is treated of the recein- ty of seweet tions, is action in the every thas clear ponitoring p inhibition use of wate llowed by ter to 35 r e within that and be con- emicals be a cleaning, ning system and the ware device seare a varied lids - e.g. sand filtra Il steps to can be dess cceptable e liquor be t and mini- open area ds which v y be nece size to hare ent carry-co- nsible pre- ins, and o rials store are some the trans- trans the trans- trans the trans- trans the trans- trans the trans- trans the trans- trans- trans the trans-	off-site at hat: I at the secon-site, b ving wate r bypass cceptably nt of byp ning or ev orogramm of any do er and m the recyc ng/l of su eir norma sidered f offore cond dry filtra multista tion, filter multista tion, filter multista tion, filter stroyed by to use so eing clear mise the as, but in vill have t ssary in o ndle storr over of ur caution ir ssential i n-site effl d are not age by a s on stor grity of si ve substa for preve e it is cor	a sewage ewage tre ased on re r; , via storm low; ass, e.g. I ven shuttin be is in pla ownstream inimising ling of wa spended a limits. S or recyclin cluding the spended a limits. S or recyclin cluding the cor recyclin cluding the spended and re ge separa r pressing e the use of y incineral crubbing a need and re waste wa particular o be remo drainage f n water an o be remo drainage f n water an settled m n all but the n many ca uent treat incompal minimum age tanks torage tar sidered in must justi	e treatment i atment work eduction of I n/emergency (nowing whe age for emiss in biological p the level of p aste water st solids is like Such cleane at recycling of ms eliminate for an aque As long as s ly be allowe or combination ation, floccul of organic so tation, and thi as long as the ecycled. Fo ter stream. from raw m by d by settl rom scrap h ad should be aterial. e most trivia ases where f ment plants tible. Bund of of 10%. should be of happropriate fy this appro	works, t s is as g oad (no y overflo en bypa en bypa sions to process oplutan reams v ly to en d water r it is als of the w e potent ous disk of the w e potent ous disk ad to rea on there atorn c olvents, is would here is n r an exis here is n r an exis atterials capacitic d shou d discha e to bun ach.	the above face good as wou to concentration ows or at inter- ss is occurring sewer, taking es and action ts in each was wherever pos- sure that mo- would gener so important vater is viable ial waste was charge if app te removed to ch saturation to release of sting plant ev- binders and l be BAT for stocking are for other tech areas. Drain hed to accom eg. dilute m- a risk to con ed bunds are es should alw d to an appro- d also be re- wall analge arges or leak d a particular	ctors app Id be ach on) of ea ermediate ng, reschang; Ig into col as plan for as plan for as plan for as of the i rally be of to consid e. ter strear of a level a n without uld adeq ipitation, chlorinate a new pro- liquor to f very effor eas, will c nage sum modate s on-hazan trolled wa possible ways exco opriate infi guarly in: d and do ages from	ly in lieved if the ch sewage eduling nsideration or any such m are the this industry nsoluble f good ler the levels ns, however measures acceptable any adverse uately ed release ocess. It the t should be ontain Dil nps should storm surge dous aters, in cases eed the tegrity and spected, cumented n bunds tank or

INTROD	UCTIO	N TEC	HNIQU	JES	EN	IISSION	IS	I	MPAC	Т
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Fugitives	2.3. On m emis • s' • tr • tr • c • p e • p	<b>13 Control o</b> nany installations sions. Common torage areas (e.g ne loading and ur ransferring mater onveyor systems ipework and duc tc.); oor building cont	of <u>fugiti</u> fugitive, c examples g. bays, sto nloading o ial from or s; twork syst ainment a	ve em or diffuse, of the sc ockpiles, f transpo ne vessel ems (e.g nd extrac	emissions ources of lagoons e rt contain to anothe . pumps, ction;	s to air ns may be m fugitive emis etc.); ers; er (e.g. furna valves, flang	ore sigr ssions a ace, ladl jes, cato	nificant than pre: e, reactors, s chpots, drain	point sou silos); s, inspec	rce tion hatches
	• p • a a	ccidental loss of nd core preparat	containme ion plant;.	ent from f	ailed plar	it and equipr	ment inc	cluding leaka	ge eg. fro	om mould
	App Que	Dication Form estion 2.3 (cont.)	>	Fugitiv	e emissio	ns to air				
	<b>With</b> 1. 2.	the Application supply the gene emissions to air identify, and wh sources, includi	the Oper eral applica ; and in ac ere possit ng those t	rator sho ation requ ddition, ble, quant below, es	uirements tify signific timating t	for Section cant fugitive he proportio	2.3 on p emissic n of tota	bage 20 for c	ontrol of n all relev which are	fugitive vant e attributable
	- H - H - P fu - F	3.1 but need to landling of molter landling of dusty process sources s umes inishing (dust)	be unders n metals materials such as re	fining, po	uring and	to demonstr	- Waste - Handl Bag f - Stora	t the controls ewater treatm ing odorous nouse dust ige silos	are adeo nent (odo raw mate	ur) erials
	<i>Indic</i> 1. 2.	cative BAT Requ The Operator sl (see item 2 in S Regulator but in Where there are	uirements hould com fection 2.3 h any case e opportur	plete any ) as an in within th hities for r	v detailed nproveme e timesca reductions	studies requent condition ale given in S s, the Permit	uired into to a tim Section may re	o abatement lescale to be 1.1. quire the upo	or contro agreed v	ol options with the entory of
	<ol> <li>Where there are opportunities for reductions, the Permit may require the updated inventory fugitive emissions to be submitted on a regular basis.</li> <li>Dust - The following general techniques should be employed where appropriate:         <ul> <li>covering of skips and vessels;</li> <li>avoidance of outdoor or uncovered stockpiles (where possible);</li> <li>where unavoidable, use of sprays, binders, stockpile management techniques, windbrea etc.;</li> <li>wheel and road cleaning (avoiding transfer of pollution to water and wind blow);</li> <li>closed conveyors, pneumatic conveying (noting the higher energy needs), minimising discussional discussiona discussiona discussional discussional discussional discussion</li></ul></li></ol>									
	4.	<ul> <li>VOCs</li> <li>When transferfilling via filling that transference enclosed system</li> <li>Vent system</li> </ul>	erring volang pipes e the vapou stem with o s should b	atile liquid xtended f ir from the extraction be choser	s, the foll to the bot e containe to suitat	owing techn tom of the co er being filleo ble abatemer hise breathin	iques sh ontainer d to the nt plant. g emiss	nould be emp , the use of v one being er ions (e.g. pro	oloyed – s /apour ba mptied, o essure/va	subsurface alance lines r an acuum
		vaives) and, equipment.	where rele	evant, sh	oula de fi	lied with Kho	OCK-OUT	bots and app	oropriate a	apatement

5. **Odour -** See Section 2.3.15.

INTRODU	CTIC	ON TECH	INIQU	JES	EM	IISSIC	NS	11	MPAC	т
Management Ma	aterials	Activities &	Ground	Waste	Energy	Accident	s Noise	Monitorina	Closure	Installation
ir	nputs	abatement	water	Waste	Energy	Recident	140130	Monitoring	Closure	issues
Fugitives	2.3	.14 Control o and grou	f <u>fugiti</u> ndwate	i <u>ve</u> emi er	issions	s to sur	face w	ater, sew	er	
	Ap Qu	plication Form lestion 2.3 (cont.)	$\ge$	Fugitive Water	Emissior	ns to				
	With	h the Application	the Ope	rator sho	uld:					
	1.	supply the gene emissions to wa	ral applica ter; and ir	ation requ	iirements	for Sectio	on 2.3 on p	bage 20 for c	ontrol of	fugitive
	2.	identify, and whe sources, estimat each substance, understood here	ere possib ting the pl These st in order	ole, quant roportion teps will b to demon	ify signifie of total er be carried strate tha	cant fugitiv nissions v out as in it the cont	ve emissic vhich are a response rols are ac	ons to water t attributable t to Section 3 dequate.	from all re o fugitive .1 but nee	elevant releases for ed to be
	Indi	cative BAT Requ	irements	;						
	1.	Where there are fugitive emissior	opportur s to be s	nities for r	eductions on a regu	, the Perr lar basis.	nit may re	quire the upo	dated inve	entory of
	2.	Subsurface str	uctures -	- the Ope	erator she	ould:				
	<u> </u>	<ul> <li>establish and</li> </ul>	record t	he routing	g of all ins	tallation c	Irains and	subsurface	pipework	,
		<ul> <li>identify all su</li> </ul>	bsurface	sumps ar	nd storage	e vessels;				
		engineer syst	tems to e	nsure lea	kages fro	m pipes e	tc are min	imised and v	where the	se occur,
		can be readil	y detecte	d, particul	larly wher	e hazardo	bus (e.g. li	sted) substa	nces are	involved;
		<ul> <li>provide in pa pipework, sui</li> </ul>	nps and s	econdary storage v	containm essels:	ient and/o	r leakage	detection for	r such sul	osurface
		<ul> <li>establish an i pressure test</li> </ul>	nspectior s, leak te	n and mai sts, mater	ntenance rial thickn	programr ess check	me for all s ເຣ or CCT	subsurface s V.	tructures,	, e.g.
	3.	Surfacing – the	Operator	should:						
		describe the	design(#)	,and cond	dition of th	ne surfacir	ng of all op	perational are	eas;	
		<ul> <li>have an inspective kerbs;</li> </ul>	ection and	d mainten	ance pro	gramme c	of impervic	ous surfaces	and conta	ainment
		• justify where	operation	al areas l	have <u>not</u>	been equi	pped with	:		
		- an imperv	vious surfa	ace;						
		- spill conta	ainment k	erbs;						
		- sealed co	nstruction	n joints;						
		- connectio	n to a sea	aled drain	age syste	em.				
		(# Relevant infor permeability; stre procedures; and	mation m ength/rein quality as	ay include forcemen ssurance	e as appr it; resistai procedure	opriate: ca nce to che es.)	apacities; mical atta	thicknesses; ick; inspectic	falls; mai on and ma	terial; aintenance
	4.	Bunds All tanks contain For further inform • be impermea	ing liquids nation on ble and re	s whose s bund sizi esistant to	pillage connections of the store	ould be ha esign, <mark>see</mark> ed materia	rmful to th Ref. 12. als;	ne environme Bunds shoul	ent should ld:	1 be bunded.
		have no outle	et (i.e. no	drains or	taps) and	l drain to a	a blind col	lection point;	;	
		have pipewor	k routed	within bu	nded area	as with no	penetratio	on of contain	ed surfac	es;
		be designed	to catch le	eaks from	i tanks or	fittings;				
		<ul> <li>have a capac</li> </ul>	ity which	is the gre	eater of 1	10% of the	e largest ta	ank or 25% c	of the tota	I tankage;
	1	<ul> <li>be subject to under manual</li> </ul>	regular v	isual insp	ection an	d any con	tents pum	ped out or o	therwise	removed
	1	where not free	auently ir	ispected	be fitted	with a high	h-level nro	be and an a	larm as a	appropriate.
	1	<ul> <li>have fill point</li> </ul>	s within th	he bund v	vhere pos	sible or o	therwise n	provide adea	uate cont	ainment:
		<ul> <li>have a routin testing where</li> </ul>	e progran structura	nmed ins al integrity	pection of	<sup>-</sup> bunds, (r bt).	normally v	isual but exte	ending to	water

IPPC



INTROD	UCTIO	N TE			EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

Groundwater protection legislation

Groundwater

# 2.4 Emissions to Groundwater

The Groundwater Regulations for the UK came into force on 1 April 1999. An IPPC Permit will be subject to the following requirements under these Regulations.

- i. It shall not be granted at all if it would permit the direct discharge of a List I substance (Regulation 4(1)) (except in limited circumstances see note 1 below).
- ii. If the Permit allows the disposal of a List I substance or any other activity which might lead to an <u>indirect discharge</u> (see note 2 below) of a List I substance then **prior investigation** (as defined in Regulation 7) is required and the Permit shall not be granted if this reveals that indirect discharges of List I substances would occur and in any event conditions to secure prevention of such discharges must be imposed (Regulation 4(2) and (3)).
- iii. In the case of List II substances, Permits allowing direct discharges or possible indirect discharges cannot be granted unless there has been a prior investigation and conditions must be imposed to prevent groundwater pollution (Regulation 5).
- iv. The Regulations contain further detailed provisions covering surveillance of groundwater (Regulation 8); conditions required when direct discharges are permitted (Regulation 9); when indirect discharges are permitted (Regulation 10); and review periods and compliance (Regulation 11).

The principles, powers and responsibilities for groundwater protection in England and Wales, together with the Agency's policies in this regard, are outlined in the Environment Agency's document "*Policy and Practice for the Protection of Groundwater*" (PPPG) (see Ref. 24). This outlines the concepts of vulnerability and risk and the likely acceptability from the Agency's viewpoint of certain activities within groundwater protection zones.

- A **Prior investigation** of the potential effect on groundwater of on-site disposal activities or discharges to groundwater. Such investigations will vary from case to case, but the Regulator is likely to require a map of the proposed disposal area; a description of the underlying geology, hydrogeology and soil type, including the depth of saturated zone and quality of groundwater; the proximity of the site to any surface waters and abstraction points, and the relationship between ground and surface waters; the composition and volume of waste to be disposed of; and the rate of planned disposal.
- **B** Surveillance this will also vary from case to case, but will include monitoring of groundwater quality and ensuring the necessary precautions to prevent groundwater pollution are being undertaken.
- *Note* 1 The Regulations state that, subject to certain conditions, the discharges of List I substances to groundwater may be authorised if the groundwater is "permanently unsuitable for other uses". Advice must be sought from the Regulator where this is being considered as a justification for such discharges.
- *Note* 2 List I and List II refer to the list in the Groundwater Regulations and should not be confused with the similar lists in the Dangerous Substances Directive.

Application Form Question 2.4 Identify if there may be a discharge of any List I or List II substances and if any are identified, explain how the requirements of the Groundwater Regulations 1998 have been addressed.

### With the Application the Operator should:

- 1. confirm that there are no direct or indirect emissions to groundwater of List I or List II substances from the installation, or
- 2. where there are such releases, provide the information and surveillance arrangements described in A and B above.

Under these Regulations the Permit may not be granted if the situation is not satisfactory, therefore, with the application, the Operator should supply information on list I and list II substances and if necessary, prior investigation and surveillance information:

Meeting the requirements of the Groundwater Regulations

INTRODUC	OTION	٧	TE	CHNIQU	JES	EN	<b>MISSIO</b>	NS		MPAC	Т
Management Ma	terials	Activit	ies &	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation
ir	nputs	abate	ment	water	Wasie	Lifeigy	Accidents	NUISE	wontoning	Closure	issues
	_										
Groundwater	List I	[									
	1(1)	Subi	ect to	sub-paragrai	ph (2) bel	low. a sul	ostance is ir	n list I if i	t belonas to	one of th	e followina
	. ,	fami	lies or	groups of su	bstances	;-			0		0
List I and List II		(a)	orgar	ohalogen co	mpounds	s and sub	stances whi	ich may	form such co	ompound	s in the
substances			aquat	tic environme	ent;						
		(b)	orgar	ophosphoru	s compou	unds;					
		(c)	orgar	otin compou	inds;						
		(d)	subst	ances which	possess	carcinog	enic, mutag	enic or t	eratogenic p	properties	in or via the
			other	wise be in lis	t II);	ung subs		innaven	lilose piopei		i woulu
		(e)	merci	ury and its co	ompound	s;					
		(f)	cadm	ium and its c	compound	ds;					
		(g)	miner	al oils and h	, ydrocarbo	ons;					
		(h)	cyani	des.	-						
	2.	A su	bstand	e is not in lis	st I if it ha	s been d	etermined b	v the An	ency to be i	nappropri	ate to list I
		on th	ne basi	is of a low ris	sk of toxic	ity, persi	stence and l	bioaccur	mulation.		
	List I	Π									
	1(1)	Asu	bstand	e is in list II	if it could	have a h	armful effec	t on gro	undwater an	id it belon	igs to one of
		the f	amilies	s or groups o	of substar	nces:					
		(a)	the fo	llowing meta	alloids and	d metals	and their co	mpound	S:		
			Zi	nc	Tin		(	Copper			
			Ba	arium	NIC	kel	E	Beryllium	1		
				nromium	BOI	onium	L	.ead /opodiur	~		
				rsenic	Col	halt	ν Δ	Antimon	11 /		
			Tł	nallium	Mo	lvbdenum	, ד ו	ellurium	, I		
			Ti	tanium	Silv	/er		0			
		(b)	biocic	les and their	derivativ	es not ap	pearing in li	st I;			
		(C)	subst	ances which	have a d	leleteriou	s effect on t	he taste	or odour of	groundwa	ater, and
			comp	ounds liable	to cause	the form	ation of sucl	n substa	nces in such	n water ar	nd to render
		(-1)		t for human (	consump	tion;	-l		h = 4 = =		
		(a)	forma	or persistent	compour	compoun ids in wat	as of silicon ter excludin	, and su a those	which are bi	lich may o	harmless
			or are	e rapidly conv	verted in	water into	b harmless s	substance	ces;	ologiouily	namiece
		(e)	inorga	anic compou	nds of ph	losphorus	and eleme	ntal pho	sphorus;		
		(f)	fluorio	des;							
		(g)	ammo	onia and nitri	tes						
	(2)	A su	bstand	ce is also in l	ist II if-						
		(a)	it belo	ongs to one o	of the fam	nilies or g	roups of sub	stances	set out in p	aragraph	1(1) above;
		(b)	it has	been detern	nined by	the Agen	cy to be ina	opropria	te to list I un	der parag	graph 1(2);
		$(\mathbf{c})$	it has	heen detern	nined by	the Agen	cv to be ann	ronriate	to list II hav	vina reaar	d to toxicity
		(0)	persis	stence and b	ioaccum	lation.	cy to be app	. opnale		g regar	a to tonioity,
	3(1)	) The powe	Secret	tary of State der paragrap	may revie h 1(2) or	ew any de 2 (2).	ecision of th	e Agenc	y in relation	to the exe	ercise of its
	3(2)	) The para	Secret graph	tary of State (1) above ar	shall noti nd it shall	fy the Ag be the du	ency of his out of the Ac	decision jency to	following a i give effect to	review un o that dec	der sub- cision.
	4	The this	Agenc Sched	cy shall from ule in such m	time to tir nanner as	ne publis it consic	h a summar lers appropr	y of the iate and	effect of its I shall make	determina copies of	ations under any such
		Sam									

INTROD	UCTIOI	N TEC	TECHNIQUES			<b>IISSION</b>	IS	IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

aste	2.5 Waste Handling
	The normal nature and source of the waste from each activity is given in Section 2.3 and will be confirmed in detail in the Operator's response to Section 3.1. In general the waste streams comprise:
	<ul> <li>raw material dusts collected in bag or cartridge filters;</li> </ul>
	launder and ladle gas cleaning dust and sludge;
	slag from ladles;
	casting dust and tume collected in filter plant;
	fettiing waste;     scrubber liquers and sludges and the ETP (Section 2.3.12);
	<ul> <li>scrubbel liquois and sludges and life ETP (Section 2.3.12),</li> <li>refractory waste from launders and ladles:</li> </ul>
	<ul> <li>sand</li> </ul>
	chemical containers and general inert industrial waste.
	Application Form Question 2.5 Characterise and quantify each waste stream and describe the proposed measures for waste management storage and handling.
	With the Application the Operator should:
	1. identify and quantify the waste streams;
	2. identify the current or proposed handling arrangements;
	<ol> <li>describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;</li> </ol>
	<ol> <li>demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures.</li> </ol>
	Indicative BAT Requirements
	<ol> <li>A system should be maintained to record the quantity, nature, origin and where relevant, the destination, frequency of collection, mode of transport and treatment method of any waste which is disposed of or recovered.</li> </ol>
	<ol> <li>Wherever practicable, waste should be segregated and the disposal route identified which shoul be as close to the point of production as possible.</li> </ol>
	3. Records should be maintained of any waste that is sent off-site (Duty of Care).
	<ol> <li>Storage areas should be located away from watercourses and sensitive boundaries e.g. adjacer to areas of public use and protected against vandalism.</li> </ol>
	5. Storage areas should be clearly marked and signed and containers should be clearly labelled.
	<ol> <li>The maximum storage capacity of storage areas should be stated and not exceeded. The maximum storage period for containers should be specified.</li> </ol>
	<ol> <li>Appropriate storage facilities should be provided for special requirements such as for substance that are flammable, sensitive to heat or light etc., and incompatible waste types should be kept separate.</li> </ol>
	<ol> <li>Containers should be stored with lids, caps and valves secured and in place. This also applies t emptied containers.</li> </ol>
	9. Storage containers, drums etc. should be regularly inspected.
	10. Procedures should be in place to deal with damaged or leaking containers.
	11. All appropriate steps to prevent emissions (e.g. liquids, dust, VOCs and odour) from storage or handling should be taken (see Sections 2.3.13, 2.3.14 and 2.3.15).

V

INTROD	UCTIOI	N TEC	TECHNIQUES			IISSION	IS	IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

# 2.6 Waste Recovery or Disposal

Waste Recovery or Disposal

The Regulations require the Regulator, in setting Permit conditions, to take account of certain general principles including that the installation in question should be operated in such a way that "waste production is avoided in accordance with Council Directive 75/442/EEC on waste; and where waste is produced it is recovered, or where this is technically or economically impossible it is disposed of, while avoiding or reducing the impact on the environment". The objectives of the National Waste Strategies should also be considered.

In order to meet this requirement the Regulator needs Operators to provide the information below.



# With the Application the Operator should:

- describe, in respect of each waste stream produced by the installation, whether the waste in question is to be recovered or disposed of, and if a disposal option is planned, to justify why recovery is "technically and economically impossible" together with "the measures planned to avoid or reduce any impact on the environment";
- 2. include in the description, the Operator's view as to whether waste disposal is likely to be restricted by the implementation of the Landfill Directive;
- 3. describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;
- 4. demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures;

#### Indicative BAT Requirements

- 1. Unless agreed with the Regulator to be inappropriate, the Operator should provide a detailed assessment identifying the best practicable environmental options for waste disposal. For existing activities, this may be carried out as an improvement condition to a timescale to be approved by the Regulator.
- 2. Where landfill is the only option it should be noted that, particularly when high in fillers, sludge does not readily de-water and can cause serious problems in landfill sites.
- 3. Table 2-2 summarises the routes of the various waste streams from a typical foundry site. Whether recycling is possible at a particular site will depend on the particular fuels and raw materials being used, the products being made and the methods of operation employed. The table reflects where recycling can be achieved when the appropriate combination of these factors can be arranged.
- 4. Operators should identify all of the dusts, sludges and slags arising from the process, identify their content, whether they are recycled and, if not, the steps which would need to be taken in order to recycle the wastes, including any economic factors which need to be taken into account.
- 5. Where sludge has to be landfilled, consideration should be given to what will happen to the water. In many plants the sludge is de-watered to produce a cake with about 20% moisture content.
- 6. The application should identify the species likely to be present in releases to land from a knowledge of the process and process metallurgy, validated as necessary by the appropriate analytical techniques.

INTROD		TECHNIQUES			EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

# Table 2-2 Solid waste stream routes

PROCESS	WASTE STREAM
Raw material bag filters	Recycled
Desulphurisation: collected dusts	Landfill - not suitable for recycling
slag	Landfill - not suitable for recycling
Sand	Recycled wherever possible
Nodularisation dust - primarily magnesium oxide	Recycling to be investigated - else to landfill
Filter casting dust and fume a) leaded	To landfill as special precautions waste
b) unleaded	Low tramp content permits recycling
Fettling waste	Recycled to furnace, where significant
Refractory waste	Landfill
Scrubber liquors and sludges	Minimised then to licensed waste disposal contractors

### Energy

# 2.7 Energy

BAT for energy efficiency under the PPC Regulations will be satisfied provided the Operator meets the following conditions:

#### either

the Operator meets the basic energy requirements in sections 2.7.1 and 2.7.2 below and is a
participant to a Climate Change Agreement (CCA) or Trading Agreement with the government

#### or

 the Operator meets the basic energy requirements in sections 2.7.1 and 2.7.2 below and the further sector-specific energy requirements in section 2.7.3 below.

Note that even where a Climate Change Agreement or Trading Agreement is in place, this does not preclude the consideration of energy efficiency as part of an integrated assessment of Best Available Techniques in which it may be balanced against other emissions.

Further guidance is given in the Energy Efficiency Guidance Note (Ref. 14).

# 2.7.1 Basic energy requirements (1)

Application Form Question 2.7 (part 1) *Provide a breakdown of the energy consumption and generation by source and the associated environmental emissions.* 

The requirements of this section are basic, low cost, energy standards which apply whether or not a Climate Change Agreement or Trading Agreement is in force for the installation.

With the Application the Operator should:

#### 1. provide the following Energy consumption information:

Energy consumption information should be provided in terms of delivered energy and also, in the case of electricity, converted to primary energy consumption. For the public electricity supply, a conversion factor of 2.6 should be used. Where applicable, the use of factors derived from onsite heat and/or power generation, or from direct (non-grid) suppliers should be used. In the latter cases, the Applicant shall provide details of such factors. Where energy is exported from the installation, the Applicant should also provide this information. An example of the format in which this information should be presented is given in Table 2.1 below. The Operator should also supplement this information with energy flow diagrams (e.g. "Sankey" diagrams or energy balances) showing how the energy is used throughout the process.

(Note that the Permit will require energy consumption information to be submitted annually)

Enorgy source	Energy consumption									
Energy source	Delivered, MWh	Primary, MWh	% of total							
Electricity*										
Gas										
Oil										
Other (Operator to specify)										

\* specify source.

#### 2. provide the following Specific Energy Consumption information

The Operator should define and calculate the specific energy consumption of the activity (or activities) based on primary energy consumption for the products or raw material inputs which most closely match the main purpose or production capacity of the installation. The Operator should provide a comparison of Specific Energy Consumption against any relevant benchmarks available for the sector.

### 3. provide associated environmental emissions

This is dealt with in the Operator's response to Section 3.1.

Table 2.1 -Example breakdown of delivered and primary energy consumption

INTRODU	JCTIC	N TEC	HNIQ	JES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Energy Continued	2.7	.2 Basic e	nergy r	equire	ments (	2)				100000
	Ap Qu	plication Form estion 2.7 (par	t 2)	Desc effici The I	cribe the p iency. requireme	roposed me	easures	are basic, le	ement of	<sup>r</sup> energy energy
				stand Agre	dards which ement or <sup>-</sup>	ch apply wh Trading Agi	ether c reemen	or not Clima t is in force	te Chang for the ii	e Istallation.
	With	the Applicati	on the Op	erator sh	nould:					
	1.	describe the or requirements	current or p <b>s</b> below, ar	proposed nd provide	position wi e justificatio	th regard to ons for not u	the <b>bas</b> sing an	sic, low cost y of the tech	energy niques de	escribed;
	2.	provide an er options as de	ergy efficie scribed be	ency plan low.	which app	raises the co	osts and	d benefits of	different	energy
	Bas	ic Energy Req	uirements	;						
	1.	<b>Operating, m</b> areas, accord Guidance No	<b>naintenand</b> ling to the o te, where r	ce and ho checklists elevant:	p <b>usekeepi</b> s provided i	<b>ng measure</b> in Appendix	e <b>s</b> shoul 2 of the	ld be in place IPPC Energ	e in the fo y Efficien	llowing cy
		<ul> <li>air condition evaporator</li> </ul>	oning, proc r/condense	ess refrig er mainter	eration and nance);	d cooling sys	stems (I	eaks, seals,	temperat	ure control,
		operation	of motors a	and drives	s;					
		compresse	ed gas syst	tems (lea	ks, proced	ures for use)	);			
		<ul> <li>steam dist</li> </ul>	ribution sys	stems (le	aks, traps,	insulation);				
		space hea	ting and ho	ot water s	systems;					
		Iubrication	to avoid h	igh frictio	n losses;					
		boiler main	ntenance e	.g. optimi	Ising exces	s air;				
		• other main	itenance re	elevant to	the activiti	es within the	Installa	ation.		
	2.	Basic, low co insulation, co unnecessary	<b>ost, physic</b> ntainment discharge	<i>cal techn</i> methods, of heated	<b>iques</b> sho (e.g. seals I water or a	uld be in pla and self-clo iir (e.g. by fit	ce to av osing do ting sim	void gross ine oors) and avo uple control s	efficiencie pidance o ystems).	s; to include f
	3.	Building ser of the Building industries the energy issues constitute mo	vices ener g Services se issues r s. They sh re than 5%	<b>gy effici</b> Section of may be of ould none of the to	ency techi of the Ener f minor imp etheless fin tal energy	niques shou gy Efficiency pact and sho id a place in consumption	Id be in Guidar uld not the pro n.	place to delince Note. Fo distract effori gramme, par	ver the re or energy t <b>from</b> the ticularly v	equirements -intensive e major where they
	4.	Provide an e	energy effi	ciency pl	<b>lan</b> which:					
		<ul> <li>identifies a 2 7 3<sup>1</sup></li> </ul>	all techniqu	es releva	int to the in	stallation, in	cluding	those listed	below an	d in Section
		<ul> <li>identifies t</li> </ul>	he extent t	o which tl	hese have	been emplo	ved:			
		<ul> <li>prioritises</li> </ul>	the applica	ble techr	niques acco	ording to the	apprais	al method p	rovided ir	the Energy
		<ul> <li>Efficiency</li> <li>identifies a requiring finance</li> </ul>	Guidance I any techniq urther asse	Note whic ues that essment (	ch includes could lead e.g. accord	advice on a to other adv	ppropria erse en odology	ate discount vironmental v, see Ref. 6)	rates, pla impacts, <sup>-</sup>	int life etc.; thereby
		Where other a that appropria	appraisal m te discount	iethodolo t rates, as	gies have l sset life an	been used, s d expenditur	state the	e method, an criteria have l	d provide been emp	evidence bloyed.
		This should be supporting infe that the Opera Agreement o implementati	e submitted ormation fr ator has co r Trading a ion of thos	d in a sun om any a nsidered <b>Agreeme</b> se measu	nmary form ppraisal pr all relevan ent is in pla ires in cat	at similar to ocedure car t techniques ace the Reg egories 1-3	the exa ried out . Howe julator above.	imple below, . The plan is ever, where a will only enf	together required a Climate orce	with to ensure e Change
										Cont.

INTRODUCTION TECHNIQUES					EMISSIONS			IMPACT			
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

		Energy efficiency	NPV	CO <sub>2</sub> saving	gs (tonnes)	NPV/CO <sub>2</sub> saved	Priority* for					
Table 2-3		option	£k	annual	lifetime	£/tonne	implementation					
for Energy		7MW CHP plant	1,372	13,500	135,000	10	high					
Efficiency		High efficiency motor	0.5	2	14	35	medium					
Measures		Compressed air	n/a	5	n/a	n/a	immediate					
		* Indicative only, based o	n cost/be	nefit apprais	al:							
		Where a Climate Change Plan should be submitted Regulator but in any case	Agreeme as an im within th	ent or Trading provement o e timescale g	g Agreement condition to a given in Sect	t is in place, the Ene timescale to be agr ion 1.1.	rgy Efficiency eed with the					
	5.	<i>Energy management te</i> 2.1 noting, in particular, th reductions.	<i>chnique</i> : he need f	<b>s</b> should be i for monitoring	n place, acco g of energy f	ording to the require lows and targeting o	ments of Section of areas for					
	Indic	ative BAT Requirements	5									
BAT for energy	1.	<ul> <li>Operating, maintenance checklists provided in App as applicable:</li> <li>air conditioning, proce evaporator/condenser</li> <li>operation of motors ar</li> <li>compressed gas syste</li> <li>steam distribution syste</li> <li>space heating and hot</li> <li>lubrication to avoid hig</li> <li>boiler maintenance relevante</li> </ul>	<ul> <li>Operating, maintenance and housekeeping measures should be in place, according to the checklists provided in Appendix 3 of the Energy Efficiency Guidance Note, in the following areas as applicable:</li> <li>air conditioning, process refrigeration and cooling systems (leaks, seals, temperature control, evaporator/condenser maintenance);</li> <li>operation of motors and drives;</li> <li>compressed gas systems (leaks, procedures for use);</li> <li>steam distribution systems (leaks, traps, insulation);</li> <li>space heating and hot water systems;</li> <li>lubrication to avoid high friction losses;</li> <li>boiler maintenance e.g. optimising excess air;</li> <li>other meintenance endower to the activities within the installation.</li> </ul>									
	2.	Basic, low cost, physical insulation, containment m unnecessary discharge o	<b>al techni</b> nethods, ( f heated	<b>ques</b> should (e.g. seals ar water or air (	be in place in nd self-closin e.g. by fitting	to avoid gross ineffic g doors) and avoida g simple control syst	ciencies; to include ance of ems).					
	3.	<b>Building services</b> energy the Building Services Sec industries these issues m energy issues. They sho constitute more than 5%	y efficien ction of th ay be of uld none of the tot	cy technique le Energy Eff minor impac theless find a al energy col	es should be ficiency Guid t and should a place in the nsumption.	in place to deliver th ance Note. For ene not distract effort <b>f</b> r programme, partic	ie requirements of irgy-intensive <i>om</i> the major ularly where they					
	4.	<b>Energy management te</b> 2.1 noting, in particular, the reductions.	<i>chnique</i> he need f	<b>s</b> should be i for monitoring	n place, acco g of energy f	ording to the require lows and targeting c	ments of Section of areas for					

INTRODUC	CTIOI	N TEC	HNIQ	JES	EM	ISSION	S	IMPACT						
Management Ma	terials puts	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
	2.7.3 Appli Ques	2.7.3 Further energy efficiency requirements         Application Form         Question 2.7 (part 3)         Describe the proposed measures for improvement of energy efficiency(only where the installation is not the subject of a Climate Change Agreement or Trading Agreement).												
				Where Agree to whi impler consid	e there is n ment in pla ch the furt nentation deration fo	o Climate C ace, the Op her energy plan, includ r this secto	Change erator s efficier ling tho r and ju	Agreement should demo ncy measure ose below, h ustify where	or Tradii onstrate es identif ave beer they hav	ng the degree ïed in the n taken into ve not.				
	With a	the Application	on the Op	erator sl	hould:									
	1.	identify which appraisal for t	of the mea he energy	asures be efficienc	elow are ap y plan in se	plicable to the ction 2.7.2.	ne activ	ities, and inc	lude then	n in the				
	2.	describe the c which are per	current or p tinent to th	oroposed e installa	position wi ition;	th regard to	the tecl	nniques belo	w, or any	others				
	3.	<ul> <li>demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures;</li> </ul>												
	Indica	ndicative BAT Requirements												
BAT for energy	The fo cost/b Guida	The following techniques should be implemented where they are judged to be BAT based on a cost/benefit appraisal according to the methodology provided in Appendix 4 of the Energy Efficiency Guidance note (Ref. 14).												
BREF Sections: 4.3.9, 5.3.7, 6.3.8	1. \ \ Q	<b>Energy effici</b> Within IPPC it generation) ar options for ene	ency tech is valid to id indirect ergy efficie	<i>niques</i> consider (emissior ncy.	both the end of the end of the strength of the	mission of di emote power	rect (he station	at and emise ) pollution wl	sions fron nen cons	n on-site idering				
	1	The following t	techniques	are app	licable in th	iis sector. F	urther ir	nformation wi	ill be four	nd in (Ref.				
	•	heat recov	ery from di	ifferent p	arts of the	processes;								
	•	high efficie	ncy dewat	ering tec	hniques to	minimise dry	ying ene	ergy;						
		minimisatio	on of water	use and	closed cire	culating wate	er syste	ms;						
		plant lavou	it to reduce	e pumpin	a distances	5:								
		phase opti	misation of	felectron	ic control r	notors;								
	:	belt convey	it cooling w ying instea releases);	vater (wh d of pneu	ich is raise umatic (alth	d in tempera lough this m	iture) in ust be b	order to reco balanced aga	over the h inst high	neat; er potential				
	•	optimised etc.;	efficiency r	neasures	s for combu	istion plant e	e.g. air/f	eedwater pre	eheating,	excess air				
	•	continuous	processin	ig instead	d of batch p	processes.								
	Note a examp	to Authors: C bles)	Continue w	ith sector	specific te	chniques (se	ee pulp	and paper ar	nd other i	notes for				
	2.	Energy supp	ly techniq	ues										
	• use of CHP;													
		use of less	s pollutina f	uels.	Ξ,									
			Pendungi											
										Cont.				

INTROD	UCTIO	N TEC	HNIQ	JES	EM	ISSION	S	I	MPAC	Т
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
BAT for energ (cont,)	V	<ul> <li>Irrespective of there are othe of the choice</li> <li>where the conflicts w</li> <li>the Operat BAT.</li> <li>Where there is than 50MW, C</li> <li>1.01 and supp Authority Air F applicable to p consulted.</li> </ul>	f whether a or BAT cons of fuel imp potential n ith energy tor should s an on-site Derators s Derators s Dement S3 Pollution Co Dlant under	Climate sideration pacts upo ninimisation efficiency provide ju combus hould cor 1.01) and ontrol guid	Change Ages is involved, in emission on of waste requirements istification to tion plant of sult the IP the Opera dance. On lso. For inv	reement or such as: s other than e emissions t ents; that the prop ther guidance tors of plant IPPC install cineration pla	Trading carbon by recov osed or ce is als on pov of 20-5 ations t ant S2.8	Agreement e.g. sulphur very of energ current situa or relevant. wer generatio 0MW should his guidance 501 Waste Ir	is in place in fuel; gy from wa ation repr For plants on (refere consult is will be g ncineratio	e, where aste esents s greater ence S2 the Local enerally n should be

INTRODUCTION T			TECHNIQUES			MISSION	S	IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.8 Accidents and their Consequences

#### Guidance

IPPC requires as a general principle that necessary measures should be taken to prevent accidents which may have environmental consequences, and to limit those consequences. This section covers general areas of any installation operations which have the potential for accidental emission.

Some installations will also be subject to the Control of Major Accident Hazards Regulations 1999 (COMAH) (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland). There is an element of overlap between IPPC and COMAH and it is recognised that some systems and information for both regimes may be interchangeable.

The COMAH regime applies to major hazards. For accident aspects covered by COMAH, reference should be made to any reports already held by the Regulator. However, the accident provisions under IPPC may fall beneath the threshold for major accident classification under COMAH and therefore consideration should be given to smaller accidents and incidents as well. Guidance (see Ref. 19), prepared in support of the COMAH Regulations may also be of help to IPPC Operators (whether or not they are covered by the COMAH regime), in considering ways to reduce the risks and consequences of accident.

General management requirements are covered in Section 2.1. For accident management, there are three particular components:

- identification of the hazards posed by the installation/activity;
- assessment of the risks (hazard x probability) of accidents and their possible consequences;
- implementation of **measures to reduce the risks** of accidents, and contingency plans for any accidents that occur.



Describe your documented system that you proposed to be used to identify, assess and minimise the environmental risks and hazards of accidents and their consequences.

# With the Application the Operator should:

- provide the accident management plan described in the indicative BAT requirements below describing the current or proposed position with regard to the techniques listed below or any others which are pertinent to the installation;
- demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures;
- 3. identify any issues which may be critical.

### Indicative BAT Requirements

- 1. A structured accident management plan should be submitted to the Regulator which should:
  - *a. identify the hazards* to the environment posed by the installation. Particular areas to consider may include, but should not be limited to, the following:
    - transfer of substances (e.g. loading or unloading from or to silos or storage tanks);
    - overfilling of silos or tanks;
    - failure of plant and/or equipment (e.g. extraction fans or pumps, over-pressure of storage silos and pipework, blocked drains);
    - failure of containment (e.g. bund and/or overfilling of drainage sumps);
    - fires and problems arising from fighting fires such as failure to contain firewaters;
    - making the wrong connections in drains or other systems;
    - preventing incompatible substances coming into contact;
    - unwanted reactions and/or runaway reactions;
    - emission of an effluent before adequate checking of its composition has taken place;
    - steam main issues;
    - vandalism.

BAT for control of accidents

INTRODUC	CTION	TEC	HNIQ	UES	E	MISSION	S	I	MPAC	Т
Management Ma	iterials Action	vities & tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
BAT for control of accidents	b.	Assess viewed a	the risks as addres	<b>s</b> - having ssing six l	identifie basic que	d the hazards, estions:	the proce	ess of asses	sing the r	isks can be
(cont.)		1. what	t is the es	stimated p	probability	/ of their occuri	rence? (S	Source freque	ency);	
		3. when	re does it	get to? (	Prediction	ns for the emise	sion – wł	nat are the pa	athways a	and
		4. what	t are the o	conseque	ences? (0	Consequence a	assessme	ent – the effe	cts on the	e receptors);
		5. what envi	t are the or ronment);	overall ris ;	ks? (Det	ermination of t	he overa	ll risk and its	significa	nce to the
		6. what accid	t can prev dents and	/ent or re l/or reduc	duce the their er	risk? (Risk ma nvironmental co	nagemer onsequer	nt – measure nces).	s to preve	ent
		The dep its locati	oth and ty ion. The	pe of ass main fact	essment ors which	will depend on n should be tak	the char en into a	acteristics of ccount are:	f the insta	llation and
		<ul> <li>the sactive</li> </ul>	scale and /ities;	nature o	f the acci	dent hazard pro	esented	by the install	ation and	the
		• the r	risks to ar	eas of po	pulation	and the enviror	nment (re	eceptors);		
		<ul> <li>the r diffic</li> </ul>	nature of total total culty in de	the instal	lation and d justifyir	d complexity or ng the adequac	otherwis by of the i	isk control te	vities and echniques	the relative 3.
	с.	identify	the tech	niques r	necessar	y to reduce th	e risks i	ncluding:		
		<b>c1.</b> the	e following	g techniq	ues, whic	h are relevant	to most i	nstallations:		
		•	an inver which co forgotter damagir destroy changes	ntory shou build have in that ma ing if they its ecosysts to the in	uld be ma environr ny appar escape ( stem). Th ventory;	intained of sub nental consequ ently innocuous e.g. a tanker of ne Permit will ro	ostances, uences if s substar f milk spil equire the	present or li they escape nces can be led into a wa e Regulator t	kely to be . It shoul environm itercourse to be noti	<ul> <li>⇒ present,</li> <li>d not be</li> <li>entally</li> <li>⇒ could</li> <li>fied of any</li> </ul>
		•	procedu compatil contact;	res shoul bility with	d be in p other sul	lace for checkir ostances with v	ng raw m which the	aterials and y may accide	wastes to entally co	ensure me into
		•	adequat provideo	e storage l;	e arrange	ments for raw r	materials	, products ar	nd wastes	should be
		•	to ensur be giver systems readings process	e that con to proce based of s such as paramete	ntrol is m ss desigr n micropr ultrasoni ers;	aintained in em alarms, trips a ocessor contro c gauges, high	nergency and other of and pase -level wa	situations, c r control aspo ssing valve c rnings and p	onsiderat ects, e.g. control, ta process in	ion should automatic nk level terlocks and
		•	preventa from the	ative tech moveme	niques, s ent of veh	uch as suitable icles, should be	e barriers e include	to prevent d d as appropr	lamage to riate;	equipment
		•	appropri containn	ate conta nent;	inment s	hould be provid	ded, e.g.	bunds and c	atchpots,	building
		•	techniqu tanks (lic high-leve	ies and p quid or po el cut-off,	rocedure owder), e and bato	s should be im .g. level measu h metering;	plemente irement,	ed to preven independent	t overfillin high-leve	g of storage श alarms,
		•	installati appropri	on securi ate and s	ty system should inc	ns to prevent un clude maintena	nauthoris nce arrai	ed access s ngements wh	hould be here nece	provided as ssary;
		•	there sh changes	ould be a to proce	n installa dures, at	tion log/diary to phormal events	o record	all incidents, ings of main	near-mis tenance i	ses, nspections;
		•	procedu incidents	res shoul s;	d be esta	blished to iden	ntify, resp	ond to and le	earn from	such
		•	the roles be ident	s and res ified;	ponsibiliti	es of personne	l involve	d in accident	managei	nent should

INTROD	JCTION	TE	CHNIQ	UES	E	MISSION	IS	I	MPACT		
Management	Materials Ac inputs ab	tivities & atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
BAT for control of accidents (cont.)		•	<ul> <li>clear gui manage</li> <li>procedu commur other en</li> <li>safe shu</li> <li>commur emerger procedu redress</li> <li>appropri accident authoritii</li> <li>personn</li> <li>the syste (Section</li> <li>proc a bu treat</li> <li>drair auto</li> </ul>	idance sh d, e.g. co res shoul lication a gineering tdown pr lication ro ncy servic res shoul this; ate contro , such as es and ev el training ems for th s 2.3.13 a edures sh nd sump, ment or co nage sum	nould be a intainmer d be in p mong op y work; ocedures butes sho ces both l d include ol technic col technic col technic ol technic col t	available on ho at or dispersion lace to avoid in erations staff d should be in p ould be establis before and in the the assessme ques should be ge equipment, procedures; ments should b ation of fugitive 4) and in addit in place to ens connected to a d be equipped rage (not to dis	w each a to extin acidents c uring shift blace; hed with he event in place isolation be identifi emissior ion, for d ure that t a drainag with a hig charge);	accident scer guish fires or occurring as a ft changes ar relevant auti of an acciden m caused an to limit the c of drains, ale ed and provi- ns are genera rainage syste he compositi je system, ar gh-level alarr there should	hario shou r let them a result o nd mainte horities a nt. Post-a d steps n onsequer erting of r ded; ally releva ems: on of the e checke n or sens l be a sys	JId be burn; f poor nance or nd accident leeded to nces of an relevant ant contents of d before sor with	
		c2. t	<ul> <li>high- level</li> <li>he following risks as ider</li> <li>adequat testing to process containe contai</li></ul>	level alar control; plus any tified in e redunda o the sam waters, s hated wat d and wh surges ar r sewer. d. There ental emist their entrol e account ney stora controlle ation sho al emissi nadvisab	y other sp 1 and 2 a ancy or s ancy or s be standa site draina ters and s bere nece nd storm- Sufficier should a ssion of r y into wa t of the au ge lagool ed waters build be gi ons from ale on saf emission	becific techniques becific techniques bove tandby plant sl ards as the mai age waters, em spillages of che essary, routed t water flows, ar it storage shou lso be spill con aw materials, p ter. Any emerge ditional firewa ans may be need to (see Refs. 15) ven to the pose vents and safe ety grounds, at ;	outinely u ues identi hould be n plant; hergency emicals s to the effl nd treated ld be pro tringency products gency fire ter flows ded to pr and 16); sibility of ety relief	fied as neces provided with firewater, ch hould, where uent system, d before emis vided to ensi procedures and waste m ewater collec or fire-fightir event contar containment valves/bursti hould be foc	es, primary m ssary to r h mainter emically e appropr with pro- ssion to c ure that fl to minimi- aterials a tion syste- ng foams. ninated fi or abate ng discs. used on r	ethod of ninimise the nance and iate, be vision to ontrolled his could be se the risk ind to em should rewater ment for Where this reducing the	

INTROD	N TEC	<b>TECHNIQUES</b>			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.9 Noise and Vibration

Within this section "*noise*" should be taken to refer to "*noise and/or vibration*" as appropriate, detectable beyond the site boundary.

The PPC Regulations require installations to be operated in such a way that "all the appropriate preventative measures are taken against pollution, in particular through the application of BAT". The definition of pollution includes "emissions which may be harmful to human health or the quality of the environment, cause offence to human senses or impair or interfere with amenities and other legitimate uses of the environment". BAT is therefore likely to be similar, in practice, to the requirements of the statutory nuisance legislation, which requires the use of "best practicable means" to prevent or minimise noise nuisance.

In the case of noise, "offence to any human senses" can normally be judged by the likelihood of complaints, but in some cases it may be possible to reduce noise emissions still further at reasonable costs, and this may exceptionally therefore be BAT for noise emissions.

For advice on how noise and/or vibration related limits and conditions will be determined see "IPPC Noise – Part 1 Regulation and Permitting", (see Ref. 20).



Describe the main sources of noise and vibration (including infrequent sources); the nearest noise-sensitive locations and relevant environmental surveys which have been undertaken; and the proposed techniques and measures for the control of noise.

#### Information needed to determine BAT 1. provide the following information for eac

for noise and

vibration

- provide the following information for **each main source of noise and vibration** that fall within the IPPC installation:
  - the source and its location on a scaled plan of the site;
  - whether continuous/ intermittent, fixed or mobile;
  - the hours of operation;
  - its description, (e.g. clatter, whine, hiss, screech, hum, bangs, clicks, thumps or tonal elements);
  - its contribution to overall site noise emission (categorise each as high, medium or low unless supporting data is available).

A common sense approach needs to be adopted in determining which sources to include. The ones which need to be considered are those which may have environmental nuisance impact; e.g. a small unit could cause an occupational noise issue in an enclosed space but would be unlikely to cause an environmental issue. Conversely a large unit or a number of smaller units enclosed within a building could, for example, cause a nuisance if doors are left open. It must also be remembered that noise, which is not particularly noticeable during the day, may become more noticeable at night.

- Provide the information required in (1) for each source plus its times of operation for *Infrequent* sources of noise and vibration, not listed above that fall within the IPPC installation: (such as infrequently operated/ seasonal operations, cleaning/maintenance activities, on-site deliveries/collections/transport or out-of-hours activities, emergency generators or pumps and alarm testing),
- 3. identify the nearest noise-sensitive sites (typically dwellings, parkland and open spaces schools, hospitals and commercial premises <u>may</u> be, depending upon the activities undertaken there) and any other points/boundary where conditions have been applied by Local Authority officers or as part of a planning consent, relating to:
  - (a) the local environment:
    - provide an accurate map or scaled plan showing grid reference, nature of the receiving site, distance and direction from site boundary;
  - (b) conditions/limits imposed which relate to other locations (i.e. boundary fence or surrogate for nearest sensitive receptor):
    - any planning conditions imposed by the Local Authority;
    - other conditions imposed by agreements, e.g. limits on operating times, technologies etc;
    - any requirements of any legal notices etc.

INTROD	UCTIO	N TEO	CHNIQ	UES	E١	<b>AISSION</b>	IS	IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Information needed to determine BA1 for noise and	-	(c) the noi • bac	se environ kground n	ment: oise level	l, if known	(day/night/e	vening) L	A,90,T;		

- specific noise level (day/evening/night) L<sub>A eq,T</sub>; and/or
- ambient noise level (day/evening/night) LA eq,T, as appropriate;
- vibration data which may be expressed in terms of the peak particle velocity (ppv) in mm s<sup>-1</sup> or the vibration dose value (VDV) in m s<sup>-1.75</sup>.

For noise these are given the meaning as defined in BS4142:1997 "Method for rating industrial noise affecting mixed residential and industrial areas", and to which reference should be made for a full description. For vibration, the appropriate standard is BS6472:1992 "Evaluation of human exposure to vibration in buildings1 to 80 Hz". In very general terms "background" is taken to be the equivalent continuous A-weighted noise remaining when the source under investigation is not operational averaged over a representative time period, T. The "ambient" level is the equivalent continuous A-weighted combination of all noise sources far and distant, including the source under investigation and "specific noise" is the equivalent continuous A-weighted noise remainer at a selected assessment point. Both are averaged over a time period, T. BS4142 gives advice on the appropriate reference periods. "Worst case" situations and impulsive or tonal noise should be accounted for separately and not "averaged out" over the measurement period.

- 4. provide *details of any environmental noise measurement surveys,* modelling or any other noise measurements undertaken relevant to the environmental impact of the site, identifying:
  - the purpose/context of the survey;
  - the locations where measurements were taken;
  - the source(s) investigated or identified;
  - the outcomes.

vibration (cont.)

- 5. Identify any specific local issues and proposals for improvements.
- 6. describe the current or proposed position with regard to the techniques below, any in Ref. 20 or any others which are pertinent to the installation.
- 7. demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures.

### Indicative BAT Requirements

- 1. The Operator should employ basic good practice measures for the control of noise, including adequate maintenance of any parts of plant or equipment whose deterioration may give rise to increases in noise (eg maintenance of bearings, air handling plant, the building fabric as well as specific noise attenuation measures associated with plant, equipment or machinery).
- 2. In addition the Operator should employ such other noise control techniques to ensure that the noise from the installation does not give rise to reasonable cause for annoyance, in the view of the Regulator and, in particular should justify where either Rating Levels (L<sub>Aeq,T</sub>) from the installation exceed the numerical value of the Background Sound Level (L<sub>A90,T</sub>), or the absolute levels of 50dB L<sub>Aeq</sub> by day or 45 by night are exceeded. Reasons why these levels may be exceeded in certain circumstances are given in Ref. 20.
- 3. In some circumstances "creeping background" see Ref. 20 may be an issue. Where this has been identified in pre-application discussions or in previous discussions with the Local Authority, the Operator should employ such noise control techniques as are considered to be appropriate to minimise problems of to an acceptable level within the BAT criteria.
- 4. Noise surveys, measurement, investigation (which can involve detailed assessment of sound power levels for individual items of plant) or modelling may be necessary for either new or existing installations depending upon the potential for noise problems. Operators may have a noise management plan as part of their management system. More information on such techniques is given in Part 2 of Ref. 20

INTRODUCTION TEC			CHNIQ	UES	ES EMISSIONS IMP					
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.10 Monitoring

This section describes monitoring and reporting requirements for emissions to all environmental media. Guidance is provided for the selection of the appropriate monitoring methodologies, frequency of monitoring, compliance assessment criteria and environmental monitoring.



Describe the proposed measures for monitoring emissions, including any environmental monitoring, and the frequency, measurement methodology and evaluation procedure proposed.

### With the Application the Operator should:

- describe the current or proposed position with regard to the monitoring requirements below or any others which are pertinent to the installation for "Emissions monitoring", "Environmental monitoring", "Process monitoring" (where environmentally relevant) and "Monitoring standards" employed;
- 2. provide, in particular, the information described in requirement 13 below;
- 3. provide justifications for not using any of the monitoring requirements described;
- 4. Identify shortfalls in the above information which the Operator believes require longer term studies to establish.

### **Emissions monitoring**

The following monitoring parameters and frequency are normally appropriate in this sector. Generally, monitoring should be undertaken during commissioning, start-up, normal operation and shut-down unless the Regulator agrees that it would be inappropriate to do so.

Where effective surrogates are available they may be used to minimise monitoring costs.

Where monitoring shows that substances are not emitted in significant quantities, consideration can be given to a reduced monitoring frequency.

#### Monitoring and reporting of emissions to water and sewer

1. Monitoring of process effluents released to controlled waters and sewers should include at least:

Parameter	Monitoring frequency
Flow rate	Continuous and integrated daily flow rate
рН	Continuous
Temperature	Continuous
COD/BOD	Flow weighted sample or composite samples, weekly analysis, reported as flow weighted monthly averages
TOC	Continuous
Turbidity	Continuous
Dissolved oxygen	Continuous

**NB** - other parameters specifically limited in the Permit should be monitored. The appropriateness of the above frequencies will vary depending upon the sensitivity of the receiving water and should be proportionate to the scale of the operations.

BOD/ADt and COD/ADt should be established annually as an annual average.

2. In addition, the Operator should have a fuller analysis carried out covering a broad spectrum of substances to establish that all relevant substances have been taken into account when setting the release limits. This should cover the substances listed in Schedule 5 of the Regulations unless it is agreed with the Regulator that they are not applicable. This should normally be done at least annually.

JIIO		JHNIQ	UES	E	MISSIO	NS	IMPACT			
terials	Activities &	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation	
puts	abatement	water		- 0,					ISSUES	
3. Any substances found to be of concern, or any other individual substances to which the local environment may be susceptible and upon which the operations may impact, should also be monitored more regularly. This would particularly apply to the common pesticides and heavy metals. Using composite samples is the technique most likely to be appropriate where the concentration does not vary excessively.										
4.	In some sec whose capa "Whole efflu measureme available (R special circu	ctors there licity for har lient toxicity ints of harm lef. 21) and umstances	may be r rm is unc /" monito n, e.g. dir d the Reg toxicity t	releases c ertain, pa ring techn rect toxicit julator will esting sho	f substances rticularly who iques can th y assessme be providing ould await th	s which a en in cor herefore l ent. Som g further at guidar	are more difficu nbination with o be appropriate e guidance on guidance in du nce.	It to mea other sub to provide toxicity te le course	sure and stances. e direct esting is . Except in	
Moni	itoring and r	reporting of	of emiss	ions to a	ir					
5.	Continuous needed to n	monitoring naintain go	would b	e expecte ol;	d where the	releases	s are significan	t and whe	ere it is	
6.	Gas flow sh releases;	ould be me	easured,	or otherw	ise determin	ied, to re	late concentrat	ions to m	lass	
7.	To relate me recorded:	easuremer	nts to refe	erence co	nditions, the	following	g will need to be	e determi	ined and	
	<ul> <li>temperat</li> </ul>	ure and pr	essure;							
	<ul> <li>oxygen,</li> </ul>	where the	emission	s are the	result of a co	ombustic	on process;			
<ul> <li>water vapour content, where the emissions are the result of a combustion pro other wet gas stream. It would not be needed where the water vapour conten exceed 3% v/v or where the measuring technique measures the other pollutar removing the water.</li> </ul>										
8.	Where appr ensure that mist or fume	opriate, pe all final rel e and free f	eriodic vis eases to from drop	ual and o air should plets.	lfactory asse l be essentia	essment ally colou	of releases sho irless, free from	ould be ui n persiste	ndertaken to nt trailing	
Moni	itoring and r	reporting	of waste	emissior	ıs					
9.	For waste e	missions tl	he follow	ing should	be monitor	ed and re	ecorded:			
	• the physi	ical and ch	emical c	ompositio	n of the was	te;				
	<ul> <li>its hazar</li> </ul>	d characte	ristics;							
	<ul> <li>handling</li> </ul>	precautior	ns and su	Ibstances	with which i	t cannot	be mixed;			
	<ul> <li>where wa a program potential the food</li> </ul>	aste is disp mme of mo contamina chain.	oosed of onitoring ants and	directly to should be potential p	land, for exa established oathways fro	ample slu that take m the lau	udge spreading es into account nd to groundwa	or an on the mate ter surfac	n-site landfill, erials, ce water or	
Envi	ronmental n	nonitoring	(beyond	d the inst	allation)					
10.	The Operate emissions to	or should controlled	consider f d water, g	the need f groundwa	or environm ter, air or lar	ental mo nd or emi	nitoring to asse ssions of noise	ess the ef	fects of	
	<ul> <li>Environment</li> <li>there are</li> <li>the emissimay be a</li> <li>the Oper environm</li> <li>there is a</li> </ul>	tal monitor e vulnerable sions are a at risk; ator is loof nent; a need to v	ing may l e recepto a significa king for d alidate m	be require ors; ant contrib epartures nodelling v	ed, e.g. wher utor to an E from standa vork.	n: nvironme ards base	ental Quality St ed on lack of ef	andard (E fect on th	EQS) which Ie	
	Image: straig	<ol> <li>Activities &amp; abatement</li> <li>In some sector monitored non-internet and the sector and the sector</li></ol>	<ol> <li>Activities &amp; Ground water</li> <li>Activities &amp; Ground water</li> <li>Activities &amp; Ground water</li> <li>abatement</li> <li>Ground water</li> <li>Any substances found environment may be smonitored more regula metals. Using composiconcentration does not concentration does not concentration does not concentration does not special circumstances</li> <li>In some sectors there whose capacity for harmavailable (Ref. 21) and special circumstances</li> <li>Monitoring and reporting of the special circumstances</li> <li>Continuous monitoring needed to maintain go</li> <li>Gas flow should be more releases;</li> <li>To relate measuremer recorded:         <ul> <li>temperature and pr</li> <li>oxygen, where the</li> <li>water vapour conteres other wet gas streat exceed 3% v/v or waremoving the water</li> </ul> </li> <li>Where appropriate, performing and reporting of the physical and characte its hazard characte</li> <li>handling precaution</li> <li>where waste is disp a programme of more potential contamination the food chain.</li> <li>Environmental monitoring</li> <li>The Operator should contamination of the emissions to controlled contamination its or further are vulnerable.</li> <li>the operator is look environment;</li> <li>there is a need to variable is a need to variable.</li> </ol>	<ul> <li><b>110N IECENTIQUES</b></li> <li><b>terials</b> Activities &amp; Ground water Waste</li> <li><b>3.</b> Any substances found to be of environment may be susceptible monitored more regularly. This metals. Using composite samp concentration does not vary exited. In some sectors there may be rewhose capacity for harm is unco "Whole effluent toxicity" monitor measurements of harm, e.g. dia available (Ref. 21) and the Reg special circumstances toxicity t</li> <li><i>Monitoring and reporting of emisss</i></li> <li>5. Continuous monitoring would be needed to maintain good control.</li> <li>6. Gas flow should be measured, releases;</li> <li>7. To relate measurements to referecorded: <ul> <li>temperature and pressure;</li> <li>oxygen, where the emission</li> <li>water vapour content, where other wet gas stream. It wo exceed 3% v/v or where the removing the water.</li> </ul> </li> <li>8. Where appropriate, periodic vise ensure that all final releases to mist or fume and free from drop <i>Monitoring and reporting of waste</i></li> <li>9. For waste emissions the follow? <ul> <li>the physical and chemical comist or fume and free from drop <i>Monitoring and reporting of waste</i></li> <li>the physical and chemical comist or fume and free from drop <i>Monitoring and reporting of waste</i></li> </ul> </li> <li>10. The Operator should consider the food chain.</li> <li>Environmental monitoring may be the emissions are a signification and support is looking for denvironment; <ul> <li>there is a need to validate monitoring may be at risk;</li> <li>there is a need to validate monitoring for denvironment;</li> </ul> </li> </ul>	<ul> <li>Activities &amp; Ground water Waste Energy</li> <li>Activities &amp; Ground water Waste Energy</li> <li>Any substances found to be of concern, environment may be susceptible and upp monitored more regularly. This would pa metals. Using composite samples is the concentration does not vary excessively.</li> <li>In some sectors there may be releases or whose capacity for harm is uncertain, pa "Whole effluent toxicity" monitoring techn measurements of harm, e.g. direct toxicit available (Ref. 21) and the Regulator will special circumstances toxicity testing shot Monitoring and reporting of emissions to ai</li> <li>Continuous monitoring would be expected needed to maintain good control;</li> <li>Gas flow should be measured, or otherw releases;</li> <li>To relate measurements to reference con recorded: <ul> <li>temperature and pressure;</li> <li>oxygen, where the emissions are the</li> <li>water vapour content, where the measuring removing the water.</li> </ul> </li> <li>Where appropriate, periodic visual and o ensure that all final releases to air should mist or fume and free from droplets.</li> <li>Monitoring and reporting of waste emissions are the of the physical and chemical composition is hazard characteristics; <ul> <li>handling precautions and substances</li> <li>where waste is disposed of directly to a programme of monitoring should be potential contaminants and potential potential potential contaminants and potential potential contaminants and potential potential contaminants and potential potential potential contaminants and potential potential potential potential receptors;</li> <li>the Operator is looking for departures environment;</li> <li>there is a need to validate modelli</li></ul></li></ul>	<ul> <li>Activities &amp; Ground water Waste Energy Accidents</li> <li>Activities &amp; Ground water Waste Energy Accidents</li> <li>Any substances found to be of concern, or any other environment may be susceptible and upon which the monitored more regularly. This would particularly ap metals. Using composite samples is the technique n concentration does not vary excessively.</li> <li>In some sectors there may be releases of substance whose capacity for harm is uncertain, particularly ap metals. Using composite samples is the techniques can the measurements of harm, e.g. direct toxicity assessme available (Ref. 21) and the Regulator will be providin special circumstances toxicity testing should await the Monitoring and reporting of emissions to air</li> <li>Continuous monitoring would be expected where the needed to maintain good control;</li> <li>Gas flow should be measured, or otherwise determin releases;</li> <li>To relate measurements to reference conditions, the recorded:         <ul> <li>water vapour content, where the emissions are the result of a content water water and pressure;</li> <li>oxygen, where the emissions are the result of a content water apour content, where the emissions are the other wet gas stream. It would not be needed wh exceed 3% v/v or where the measuring technique removing the water.</li> </ul> </li> <li>Where appropriate, periodic visual and olfactory assee ensure that all final releases to air should be essentiamist or fume and free from droplets.</li> <li>Monitoring and reporting of waste emissions</li> <li>For waste emissions the following should be monitor in the physical and chemical composition of the was a programme of monitoring should be established potential contaminants and potential pathways from the food chain.</li> <li>The Operator should consider the need for environment emissions to controlled water, groundwater, air or lar may be at risk;</li> <li>the Operator is looking for departures fro</li></ul>	<ul> <li>Internals Activities &amp; Ground water Waste Energy Accidents Noise</li> <li>Activities &amp; Ground to be of concern, or any other individual environment may be susceptible and upon which the operatic monitored more regularly. This would particularly apply to the metals. Using composite samples is the technique most likely concentration does not vary excessively.</li> <li>In some sectors there may be releases of substances which a whose capacity for harm is uncertain, particularly apply to the metals. Using composite samples is the technique most likely concentration does not vary excessively.</li> <li>In some sectors there may be releases of substances which a whose capacity for harm is uncertain, particularly when in com "Whole effluent toxicity" monitoring techniques can therefore measurements for harm, e.g. direct toxicity assessment. Som available (Ref. 21) and the Regulator will be providing further special circumstances toxicity testing should await that guidat Monitoring and reporting of emissions to air</li> <li>Continuous monitoring would be expected where the releases: needed to maintain good control:</li> <li>Gas flow should be measured, or otherwise determined, to re releases;</li> <li>To relate measurements to reference conditions, the following recorded:</li> <li>temperature and pressure;</li> <li>oxygen, where the emissions are the result of a combustic other wet gas stream. It would not be needed where the v exceed 3% viv or where the measuring technique measur removing the water.</li> <li>Where appropriate, periodic visual and olfactory assessment ensure that all final releases to air should be essentially colour mist or fume and free from droplets.</li> <li>Monitoring and reporting of waste emissions</li> <li>For waste emissions the following should be monitored and noit or fume and free from droplets.</li> <li>Monitoring and reporting of waste emission of the waste;</li> <li>its hazard characteristics;</li> <li>handling precautio</li></ul>	<ul> <li>Internals ACtivations &amp; Group Waste Energy Accidents Noise Monitoring batement water water and puts advanters of the environment may be susceptible and upon which the operations may impact monitored more regularly. This would particularly apply to the common pest metals. Using composite samples is the technique most likely to be appropriate on available (Ref. 21) and the Regulator will be providing further guidance in due whose capacity for harm is uncertain, particularly when in combination with a "water" whole effluent toxicity "monitoring techniques can therefore be appropriate measurements of harm, e.g. direct toxicity assessment. Some guidance in du special circumstances toxicity testing should await that guidance.</li> <li>Monitoring and reporting of emissions to air</li> <li>Continuous monitoring would be expected where the releases are significan needed to maintain good control;</li> <li>Gas flow should be measured, or otherwise determined, to relate concentrat releases;</li> <li>To relate measurements to reference conditions, the following will need to b recorded:         <ul> <li>temperature and pressure;</li> <li>water vapour content, where the releasing are the result of a combustion process;</li> <li>water vapour content, where the emissions are the result of a combustion process;</li> </ul> </li> <li>Where appropriate, periodic visual and olfactory assessment of releases she ensure that all final releases to air should be monitored and recorded:             <ul> <li>the appropriate of monitoring to wate emission of the waste;</li> <li>thandling precautions and substances with which it cannot be mixed;</li> <li>water vapour content, where the emissions are the result of a combustion roter we gas stream. It would not be needed where the water vapour content water.</li> </ul> </li> <li>Where appropriate, periodic visual and olfactory assessment of releases she ensure th</li></ul>	<ul> <li>Interview and the provided pro</li></ul>	

INTRODUC	TION T	CHNIQ	UES	E	MISSIO	NS	IN	IMPACT					
Management Ma	terials Activities	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation				
in	puts abatemer	t water		- 35					issues				
Environmental monitoring (cont.)	The need • ground accoun gradie	should be of water, when it short and l it and down	consider e it should long-term -gradient	ed for: d be desi variation of the site	gned to char is in both.  M e;	acterise t onitoring	ooth quality an will need to ta	d flow an ke place	d take into both up-				
	<ul> <li>surface upstrea</li> </ul>	e water, whe am and dow	re consid nstream c	eration w quality of	ill be needeo the controlle	d for sam d water;	pling, analysis	and repo	orting for				
	<ul> <li>air, inc</li> </ul>	air, including odour;											
	<ul> <li>land co</li> </ul>	<ul> <li>land contamination, including vegetation, and agricultural products;</li> </ul>											
	assess     noiso	<ul> <li>assessment of health impacts;</li> <li>noise</li> </ul>											
	• noise.	• noise.											
	Where env up propos	Where environmental monitoring is needed the following should be considered in drawing up proposals:											
	determ	inands to be	emonitore	ed, stand	ard reference	e method	s, sampling pro	otocols;					
	monito     determ	ring strategy	, selectio	n of mon	itoring points	, optimisa , othor oo	ation of monito	ring appi	oach;				
	<ul> <li>uncerta measu</li> </ul>	ainty for the rement;	employed	d method	ologies and t	he resulta	ant overall unc	ertainty o	of				
	<ul> <li>quality mainter</li> </ul>	assurance ( nance, sam	QA) and ole storag	quality co je and ch	ontrol (QC) pr ain of custod	rotocols, ly/audit tr	equipment cali ail;	bration a	ınd				
	<ul> <li>reporti the pro</li> </ul>	ng procedure	es, data s prmation f	torage, in for the Re	nterpretation egulator.	and revie	ew of results, re	eporting f	format for				
	Guidance Guidance	Guidance on air quality monitoring strategies and methodologies can be found in Technical Guidance Notes M8 and M9 (see Ref. 21), for noise (see Ref. 20) and for odour (see Ref. 23).											
	Monitoring of	process var	iables										
Monitoring process	11. Some pro and moni	cess variabl ored as app	es will ha ropriate.	ve potent Example	tial environmes might be:	ental imp	act and these	should b	e identified				
variables	<ul> <li>raw mainadeo</li> </ul>	aterials moni uate supplie	toring for r informa	contamir tion (see	Section 2.2.	contamin 1);	ants are likely	and there	e is				
	<ul> <li>plant e</li> </ul>	fficiency whe	ere it has	an enviro	onmental rele	evance;							
	energy     energy	plan. Frequ	ency – no	ormally co	ontinuous an	dual poin d recorde	ed;	cordance					
	<ul> <li>fresh v part of</li> </ul>	the water ef	ross the a ficiency p	ictivities a Ilan <mark>(see</mark>	Section 2.2.3	ual points 3). Frequ	s of use should lency – continu	be mon lous and	recorded.				
	Monitoring sta	ndards (Sta	ndard R	eference	Methods)								
	Equipment sta	ndards											
Equipment standards MCERTS	The Environme quality of monit monitoring are monitoring syst stack emissions water monitorin installation, cali provision of saf	nt Agency ha pring data ar it for purpos ems (CEMs) monitoring, g instrument pration and r e access for	as introdu nd to ensu e. Perfor , and othe portable ation, dat maintenau manual s	iced its M ure that the mance star er MCER emission ta acquision ta acquision ta acquision tack mor	lonitoring Ce ne instrumen andards hav TS standards is monitoring ition and Ope ponitoring equi nitoring.	rtification tation and e been p s are und equipme erators' or ipment, p	Scheme (MCI d methodologie ublished for co ler developmer ent, ambient air wn arrangeme position of sam	ERTS) to es emplo ontinuous to cove quality r nts, such pling por	improve the yed for emissions r manual nonitors, as for ts and				
	12. As far as requireme using a re accordan the monit MCERTS	possible, Op ents of MCEF gistered state ce with MCE oring provision equipment	erators sl RTS wher ck testing RTS requ ons in det	hould ens re availat organisa uirements ail. See	sure their mo ole, e.g. using ation etc. Wh a, the Operato Environment	nitoring a g certified here the n or should Agency	arrangements of I instruments a nonitoring arra provide justifio Website (Ref.	comply w nd equip ngement cation an 21) for lis	ith the ment, and s are not in d describe sting of				

INTRODU	CTIC	DN TEC	CHNIQ	UES	E	MISSIO	NS	IMPAC	Т			
Management <sup>N</sup>	laterials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring Closure	Installation issues			
Monitoring standards (cont.)	13.	<ul> <li>13. The following should be described in the application indicating which monitoring provisions comply with MCERTS requirements or for which other arrangements have been made:</li> <li>monitoring methods and procedures (selection of Standard Reference Methods);</li> <li>justification for continuous monitoring or spot sampling;</li> <li>reference conditions and averaging periods;</li> <li>measurement uncertainty of the proposed methods and the resultant overall uncertainty;</li> <li>criteria for the assessment of non-compliance with Permit limits and details of monitoring strategy aimed at demonstration of compliance;</li> <li>reporting procedures and data storage of monitoring results, record keeping and reporting intervals for the provision of information to the Regulator;</li> <li>procedures for monitoring during start-up and shut-down and abnormal process conditions;</li> <li>drift correction calibration intervals and methods;</li> <li>the accreditation held by samplers and laboratories or details of the people used and the training/competencies.</li> </ul>										
Standards for sampling and analysis BREF: Monitoring REF document in preparation.	Sam 14. Furth Tech this s anal Furth Com If in c	The analytic needing to b Comité E British St Internatio United S Americar Deutches Verein D Association Subject is curr ysis which will her guidance mittee of Ana doubt the Ope	alysis sta al method be monitore uropéen d andards Ir onal Stand tates Envir n Society fe s Institute f eutcher Ing on França on standar ce Note 4 rently in pr I also be s relevant to ilysts.	ndards s given ir ed, stand le Norma natitution ardisation or Testing für Normu genieure ise de No rds for me (Monitorii eparatior uitable fo water ar	Append ards shor lisation (( (BSI); n Organis l Protecti g and Ma ung (DIN) (VDI); ormalisati onitoring ng) (see l n. This gu or calibrat nd waste	ix 1 should b uld be used i CEN); ation (ISO); on Agency (I terials (ASTN ; on (AFNOR) gaseous rele Ref. 21). A s uidance spec ion of continu is available f gulator.	ue used. n the foll JS EPA) M); aseries of cifies mai uous em from the	In the event of other su lowing order of priority: ; ; evant to IPC/IPPC is giv updated Guidance Note nual methods of samplir ission monitoring instrur publications of the Stand	bstances ven in the es covering ng and nents. ding			

INTROD	N TE	TECHNIQUES			MISSIO	NS		IMPACT		
Management	Materials inputs	Main activities	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.11 De-commissioning

The IPPC application requires the preparation of a site report whose purpose, as described in more detail in Refs. 4 and 5 is to provide a point of reference against which later determinations can be made of whether there has been any deterioration of the site and information on the vulnerability of the site.



### With the Application the Operator should:

- 1. supply the site report;
- describe the current or proposed position with regard to the techniques below or any others which are pertinent to the installation;
- 3. for existing activities, identify shortfalls in the above information which the Operator believes require longer term studies to establish.

# Indicative BAT Requirements

# 1. Operations during the IPPC Permit

Operations during the life of the IPPC Permit should not lead to any deterioration of the site if the requirements of the other sections of this and the specific sector notes are adhered to. Should any instances arise which have, or might have, impacted on the state of the site the Operator should record them along with any further investigation or ameliorating work carried out. This will ensure that there is a coherent record of the state of the site throughout the period of the IPPC Permit. This is as important for the protection of the Operator as it is for the protection of the environment. Any changes to this record should be submitted to the Regulator.

# 2. Steps to be taken at the design and build stage of the activities

Care should be taken at the design stage to minimise risks during decommissioning. For existing installations, where potential problems are identified, a programme of improvements should be put in place to a timescale agreed with the Regulator. Designs should ensure that:

- underground tanks and pipework are avoided where possible (unless protected by secondary containment or a suitable monitoring programme);
- there is provision for the draining and clean-out of vessels and pipework prior to dismantling;
- lagoons and landfills are designed with a view to their eventual clean-up or surrender;
- insulation is provided which is readily dismantled without dust or hazard;
- materials used are recyclable (having regard for operational or other environmental objectives).

# 3. The site closure plan

A site closure plan should be maintained to demonstrate that, in its current state, the installation can be decommissioned to avoid any pollution risk and return the site of operation to a satisfactory state. The plan should be kept updated as material changes occur. Common sense should be used in the level of detail, since the circumstances at closure will affect the final plans. However, even at an early stage, the closure plan should include:

- either the removal or the flushing out of pipelines and vessels where appropriate and their complete emptying of any potentially harmful contents;
- plans of all underground pipes and vessels;
- · the method and resource necessary for the clearing of lagoons;
- the method of ensuring that any on-site landfills can meet the equivalent of surrender conditions;
- the removal of asbestos or other potentially harmful materials unless agreed that it is reasonable to leave such liabilities to future owners;

BAT for

decommissioning

INTROD	N TE	CHNIQ	UES	E	MISSIO	NS		IMPACT		
Management	Materials inputs	Main activities	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

BAT for decommissioning (cont.)	<ul> <li>methods of dismantling buildings and other structures, see Ref. 25 which gives guidance on the protection of surface and groundwater at construction and demolition-sites;</li> </ul>
	<ul> <li>testing of the soil to ascertain the degree of any pollution caused by the activities and the need for any remediation to return the site to a satisfactory state as defined by the initial site report.</li> </ul>
	(Note that radioactive sources are not covered by this legislation, but decommissioning plans should be co-ordinated with responsibilities under the Radioactive Substances Act 1993.)

For existing activities, the site closure plan may, if agreed with the Regulator, be submitted as an improvement condition.

INTRODUCTION T			CHNIQ	UES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

# 2.12 Installation Wide Issues

In some cases it is possible that actions which benefit the environmental performance of the overall installation will increase the emissions from one Permit holder's activities. For example, taking treated effluent as a raw water supply will probably slightly increase emissions from that activity but could dramatically cut the total emissions from the whole installation.



Where you are not the only Operator of the installation, describe the proposed techniques and measures (including those to be taken jointly by yourself and other Operators) for ensuring the satisfactory operation of the whole installation.

# With the Application the Operator should:

- 1. where there are a number of separate Permits for the installation (particularly where there are different Operators), **identify** any installation wide issues and opportunities for further interactions between the Permit holders whereby the performance of the overall installation may be improved; and in particular
- 2. describe the current or proposed position with regard to the techniques below, or any others which are pertinent to the installation;

# Indicative BAT Requirements

The possibilities will be both sector and site-specific, and include:

- 1. communication procedures between the various Permit holders; in particular those needed to ensure that the risk of environmental incidents is minimised;
- 2. benefiting from the economies of scale to justify the installation of a CHP plant;
- 3. the combining of combustible wastes to justify a combined waste-to-energy/CHP plant;
- 4. the waste from one activity being a possible feedstock for another;
- 5. the treated effluent from one activity being of adequate quality to be the raw water feed for another activity;
- 6. the combining of effluent to justify a combined or upgraded effluent treatment plant;
- 7. the avoidance of accidents from one activity which may have a detrimental knock-on effect on the neighbouring activity;
- 8. land contamination from one activity affecting another or the possibility that one Operator owns the land on which the other is situated.

BAT across the whole installation

INTRODU	TI	ECHI	VIQUES		EMISSI	ONS	IMPACT			
Benchmark comparison	Benchmark status	BOD	COD	Halogens	Heavy metals	Nitrogen oxides	Nutrients	Particulate	Sulphur dioxide	VOCs

# **3 EMISSION BENCHMARKS**

# 3.1 Emissions Inventory and Benchmark Comparison

Describe the nature, quantities and sources of foreseeable emissions into each medium (which will result from the techniques proposed in Section 2).

With the Application the Operator should:

- 1. provide a table of significant emissions of substances (except noise, vibration, odour or heat which are covered in their respective sections) that will result from the proposals in Section 2 and should include, preferably in order of significance:
  - substance (where the substance is a mixture, e.g. VOCs or COD, separate identification of the main constituents or inclusion of an improvement proposal to identify them);
  - source, including height, location and efflux velocity;
  - media to which it is released;
  - any relevant EQS or other obligations;
  - benchmark;

Application Form

Question 3.1

- proposed emissions normal/max expressed, as appropriate (see Section 3.2), for:
  - mass/unit time;
  - concentration;
  - annual mass emissions.
- statistical basis (average, percentile etc.);
- notes covering the confidence in the ability to meet the benchmark values;
- if intermittent, the appropriate frequencies;
- plant loads at which the data is applicable;
- whether measured or calculated (the method of calculation should be provided).

The response should clearly state whether the emissions are current emission rates or those planned following improvements, and should cover emissions under both normal and abnormal conditions for:

- · point source emissions to surface water, groundwater and sewer;
- waste emissions (refer to Sections 2.5 and 2.6 Waste Management);
- point source emissions to air;
- significant fugitive emissions to all media, identifying the proportion of each substance released which is due to fugitives rather than point source releases;
- abnormal emissions from emergency relief vents, flares etc.;
- indirect and direct emission of carbon dioxide associated with energy consumed or generated.

Emissions of carbon dioxide associated with energy use should be broken down by energy type and, in the case of electricity, by source e.g. public supply, direct supply or on site generation. Where energy is generated on site, or from a direct (non-public) supplier, the Operator should specify and use the appropriate factor. Standard factors for carbon dioxide emissions are provided in the Energy Efficiency Guidance Note.

Where VOCs are released, the main chemical constituents of the emissions should be identified. The assessment of the impact of these chemicals in the environment will be carried out as in response to Section 4.1.

For waste, emissions relate to any wastes removed from the installation, or disposed of at the installation under the conditions of the Permit, e.g. landfill. Each waste should have its composition determined and the amounts expressed in terms of cubic metres or tonnes per month.

A suitable table on which to record this information is provided in the electronic version of this Guidance Note.

- 2. compare the emissions with the benchmark values given in the remainder of this Section;
- 3. where the benchmarks are not met, revisit the responses made in Section 2 as appropriate (see Section 1.2) and make proposals for improvements or justify not doing so.

**IPPC** 

INTRODUCTION	TECHNIQUES				EMISS	ONS	IMPACT		
Benchmark Benchmark comparison status	BOD	COD	Halogens	Heavy metals	Nitrogen oxides	Nutrients	Particulate	Sulphur dioxide	VOCs

# 3.2 The Emission Benchmarks

Introduction to emission benchmarks Guidance is given below on release concentrations or mass release rates achievable for key substances using the best combination of techniques. These BAT-based benchmarks are not mandatory release limits and reference should be made to Section 1 and the *Guide for Applicants* regarding their use.

# 3.2.1 Emissions to air associated with the use of BAT

The emissions quoted in Table 3.1 are as daily averages based upon continuous monitoring during the period of operation. Standard conditions of 273 K and 101.3 kPa with no correction is applied for water vapour or oxygen content of the emission. Care should always be taken to convert benchmark and proposed releases to the same reference conditions for comparison. To convert measured values to reference conditions, see Technical Guidance Note M2 (Ref.21) for more information

Limits in Permits may be set for mean or median values over long or short periods. The periods and limits selected should reflect:

- the manner in which the emission may impact upon the environment;
- likely variations which will arise during operation within BAT;
- possible failure modes and their consequences;
- the capabilities of the monitoring and testing system employed.

Where emissions are expressed in terms of concentrations and where continuous monitors are employed, it is recommended that limits are defined such that:

- not more than one calendar monthly average during any rolling twelvemonth period shall exceed the benchmark value by more than 10%;
- not more than one half hour period\* during any rolling 24 hour period shall exceed the benchmark value by more than 50%.

\* for the purpose of this limit half hourly periods commence on the hour and the half hour.

Where spot tests are employed:

- the half hour limit above shall be applied over the period of the test;
- the mean of three consecutive tests taken during a calendar year shall not exceed the benchmark value by more than 10%.

# 3.2.2 Emissions to water associated with the use of BAT

Wastewater treatment systems can maximise the removal of metals using precipitation, sedimentation and filtration. The reagents used for precipitation will be defined by the mix of metals present, and may include hydroxide, sulphide or a combination of both. Concentrated effluents should be pre treated before discharge into the final effluent treatment system, and techniques such as electrolysis, reverse osmosis and metal removal using ion exchange systems may need to be employed. Water discharges should be kept to a minimum by using closed cycle cooling systems and by maximising the reuse of treated process water.

Where automatic sampling systems are employed, limits may be defined such that:

not more than 5% of samples shall exceed the benchmark value.

Where spot samples are taken:

• no spot sample shall exceed the benchmark value by more than 50%.

Examples of emissions to water associated with the use of BAT:

INTRODUCTION		TECHNIQUES				EMISSIONS		IMPACT		
Benchmark comparison	Benchmark status	BOD	COD	Halogens	Heavy metals	Nitrogen oxides	Nutrients	Particulate	Sulphur dioxide	VOCs

# Table 3-1 Emissions to air associated with the use of BAT

Process	Emission	New Processes - Abated Release					
		Average Concentration mg/m <sup>3</sup>	Peak Concentratio n mg/m <sup>3</sup>	Mass Emission g/tonne of product			
EAF Secondary	Particulate	10	20	50			
EAF Combined Extraction	Particulate	15	30				
Roof Extraction	Particulate	10	20	20			
Ladle Treatment	Particulate	10	20				
General Local Extraction	Particulate	10	-				
Leaded Steel Processes	Lead	3	-				
Stainless Steel & Special Alloy Processes	Chromium	4	-				
	Nickel	2	-				
All Iron & Steel Processes	Dioxins (ITEQ)	1ng/m <sup>3</sup>	-				
	Cadmium	0.2	-				
	Lead	1	-				
	Chromium	2	-				
	Nickel	1	-				
	VOC <sup>*</sup> as total carbon	50	-				
	Fluorides	5	-				

### NOTES:

- 1. Figures are based on measurements taken at the point of discharge.
- 2. The average concentration figures refer to measurements taken over a complete process cycle for batch operations. For continuous processes, releases are based on an hourly average figure over a rolling 24 hour period taking into account only the hours when the plant is in actual operation, including start-up and shut-down. The maximum hourly average value should not exceed the release level indicated in the table.
- 3. The term "Volatile Organic Compounds" includes all organic compounds released to air in the gas phase. For details of the classification scheme see Appendix 1.
- 4. Operators should be aware of the potential for the release of polychlorinated dibenzo-p-dioxins and polychlorinated dibenzofurans ("dioxins") from metals processes particularly in secondary recovery/refining processes. An achievable release of 1ng/m<sup>3</sup> ITEQ is appropriate.

All releases must be controlled and minimised to ensure that ambient air quality beyond the process boundary complies, as a minimum, with air quality standards (see Ref 11).

INTRODUCTION		TECHNIQUES			5	EMISSIONS		IMPACT		
Benchmark	Benchmark	BOD	COD	Halogens	Heavy	Nitrogen	Nutrients	Particulate	Sulphur	
comparison	status	DOD	COD	rialogens	metals	oxides	Nutricitio	i articulate	dioxide	V003

# 3.2.3 Standards and obligations

In addition to meeting the requirements of BAT, there are other national and international standards and obligations which must either be safeguarded through the IPPC Permit or, at least, taken into account in setting Permit conditions. This is particularly the case for any EC based EQSs.

### EC based EQ standards

*IPPC: A Practical Guide* (see Ref. 4) explains how these should be taken into account and contains an annex listing the relevant standards. See Appendix 2 for equivalent legislation in Scotland and Northern Ireland). They can be summarised as follows.

### Air Quality

- Statutory Instrument 1989 No 317, Clean Air, The Air Quality Standards Regulations 1989.
- Statutory Instrument 1997 No 3043, Environmental Protection, The Air Quality Regulations 1997.

### Water Quality

- Directive 76/464/EEC on Pollution Caused by Dangerous Substances Discharged to Water contains two lists of substances. List I relates to the most dangerous, and standards are set out in various daughter Directives. List II substances must also be controlled. Annual mean concentration limits for receiving waters for List I substances can be found in SI 1989/2286 and SI 1992/337 the Surface Water (Dangerous Substances Classification) Regulations. Values for List II substances are contained in SI 1997/2560 and SI 1998/389. Daughter Directives cover EQS values for mercury, cadmium, hexachlorocyclohexane, DDT, carbon tetrachloride, pentachlorophenol, aldrin, dieldrin, endrin, isodrin, hexachlorobenzene, hexachlorobutadiene, chloroform, 1,2-dichloroethane, trichloroethane, perchloroethane and trichlorobenzene.
- Other waters with specific uses have water quality concentration limits for certain substances. These are covered by the following Regulations:
  - -SI 1991/1597 Bathing Waters (Classification) Regulations;
  - -SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regulations;
  - SI 1997/1332 Surface Waters (Shellfish) (Classification) Regulations;
  - -SI 1996/3001 The Surface Waters (Abstraction and Drinking Water) (Classification) Regulations.

#### Future likely changes include:

- Some air and water quality standards may be replaced by new standards in the near future.
- The (Draft) Solvents Directive on the limitation of emissions of VOCs due to the use of organic solvents in certain activities and installations.

#### Other standards and obligations

Those most applicable to this sector are:

- Hazardous Waste Incineration Directive;
- Waste Incineration Directive;
- Large Combustion Plant Directive;
- Reducing Emissions of VOCs and Levels of Ground Level Ozone: a UK Strategy;
- Water Quality Objectives assigned water quality objectives to inland rivers and water courses (ref. Surface (Rivers Ecosystem) Classification);
- The UNECE convention on long-range transboundary air pollution;
- The Montreal Protocol;
- The Habitats Directive (see Section 5.3).

# 3.2.4 Units for benchmarks and setting limits in Permits

Releases can be expressed in terms of:

- "concentration" (e.g. mg/l or mg/m<sup>3</sup>) which is a useful day-to-day measure of the effectiveness of any abatement plant and is usually measurable and enforceable The total flow must be measured/controlled as well;
- "specific mass release" (e.g. kg/ tproduct or input or other appropriate parameter) which is a
  measure of the overall environmental performance of the plant (including the abatement plant)
  compared with similar plants elsewhere;

INTRODUCTION		TECHNIQUES			;	EMISSIONS		IMPACT		
Benchmark comparison	Benchmark status	BOD	COD	Halogens	Heavy metals	Nitrogen oxides	Nutrients	Particulate	Sulphur dioxide	VOCs

• "absolute mass release" (e.g. kg/hr, t/yr) which relates directly to environmental impact.

When endeavouring to reduce the environmental impact of an installation, its performance against each of these levels should be considered, as appropriate to the circumstances, in assessing where improvements can best be made.

When setting limits in Permits the most appropriate measure will depend on the purpose of the limit. It may also be appropriate to use surrogate parameters which reflect optimum environmental performance of plant as the routine measurement, supported by less frequent check-analyses on the final concentration. Examples of surrogate measures would be the continuous measurement of conductivity (after ion-exchange treatment) or total carbon (before a guard-column in activated carbon treatment) to indicate when regeneration or replacement is required.

# 3.2.5 Statistical basis for benchmarks and limits in Permits

Conditions in Permits can be set with percentile, mean or median values over yearly, monthly or daily periods, which reflect probable variation in performance. In addition absolute maxima can be set.

Where there are known failure modes, which will occur even when applying BAT, limits in Permits may be specifically disapplied but with commensurate requirements to notify the Regulator and to take specific remedial action.

*For Water:* UK benchmarks or limits are most frequently 95 percentile concentrations or absolute concentrations, (with flow limited on a daily average or maximum basis).

*For Air:* benchmarks or limits are most frequently expressed as daily averages or, typically 95% of hourly averages.

# 3.2.6 Reference conditions for releases to air

The reference conditions of substances in releases to air from point sources are: temperature 273 K (0° C), pressure 101.3 kPa (1 atmosphere), no correction for water vapour or oxygen.

The reference conditions for combustion or incineration processes are as given in the appropriate guidance note.

These reference conditions relate to the benchmark release levels given in this Note and care should always be taken to convert benchmark and proposed releases to the same reference conditions for comparison. The Permit may employ different reference conditions if they are more suitable for the process in question.

To convert measured values to reference conditions, see Technical Guidance Note M2 (Ref. 21) for more information.

# **4 BENCHMARK RELEASE LEVELS**

The release levels given are achievable by all new processes using the best available techniques described in Sections 2 and 3. They are applicable to each non-combustion point source. They should not be applied as uniform release limits, but should be assessed to take account of site-specific conditions so as to comply with the requirement to use BAT and any relevant environmental quality standards.

The benchmark release levels to air are shown by substance and process sector in Table 4.1.

# 4.1 Benchmark releases to air

# Table 4-1 Benchmark levels for releases to air<sup>a</sup>

Source	Emission	Release concentration (mg/m <sup>3)</sup>	Release rate
Storage and transport of sand, including reclaimed sand	Sand and dust	10	
Storage, handling and	Volatile organic	20	Total Class A 100 g/h
use of chemical binders	compounds (VOCs) <sup>b</sup>	80	Class B 2 kg/h
Knock-out and sand	Sand and dust	10	
Tecovery	VOCs		As for storage and handling
Shot blasting, fettling and other finishing operations	Sand, dust, metallic particles	10	
	Metallurgical fume		
Casting station	Metallurgical fume	10	
	VOCs		As for storage and handling

*a)* All releases should be essentially colourless, free from persistent trailing mist or fume and free from droplets.

Releases from the processes should not give rise to an offensive odour noticeable outside the site where the process is carried on.

Releases below these mass emission rates may not be trivial, and so may still require controls and the setting of appropriate release limits.

When defining release limits for specific operations, the Agency will take into consideration the accuracy, precision and reliability of the monitoring or testing methods that are to be employed to demonstrate compliance.

*b)* The term 'volatile organic compounds' includes all organic compounds released to air in the gas phase.

Where it is practicable to do so, the VOCs being released should be identified individually using nomenclature compatible with that of the *Inventory of Sources and Releases* (ISR) reporting form devised by the Agency. The VOC concentration levels apply where the total mass release rates are exceeded.
## 4.2 Benchmark releases to surface water

Releases to surface waters from the ancillary activity are unlikely to represent more than a small fraction of the total discharge from the overall process. Furthermore, it is unlikely that they will be discharged separately from releases arising from the primary process. The benchmark levels for the main iron and steel process should therefore apply. Refer to Section 2.3.12 for indicative BAT requirements for abatement of point sources to surface water and sewer.

# **5 IMPACT**

## 5.1 Assessment of the Impact of Emissions on the Environment

The Operator should assess that the emissions resulting from the proposals for the activities/installation will provide a high level of protection for the environment as a whole, in particular having regard to EQSs etc, revisiting the techniques in Section 2 as necessary (see Section 1.2).



## 5.2 The Waste Management Licensing Regulations



In relation to activities involving the disposal or recovery of waste, the Regulators are required to exercise their functions for the purpose of achieving the relevant objectives as set out in Schedule 4 of the Waste Management Licensing Regulations 1994. (For the equivalent Regulations in Scotland, see Appendix 2. In Northern Ireland there are no equivalent regulations at the time of writing. Contact EHS for further information.)

The relevant objectives, contained in paragraph 4, Schedule 4 of the Waste Management Licensing Regulations 1994 (*SI 1994/1056 as amended*) are extensive, but will only require attention for activities which involve the recovery or disposal of waste. Paragraph 4 (1) is as follows:

a) "ensuring the waste is recovered or disposed of without endangering human health and without using process or methods which could harm the environment and in particular without:

risk to water, air, soil, plants or animals; or

causing nuisance through noise or odours; or

adversely affecting the countryside or places of special interest;

b) implementing, as far as material, any plan made under the plan-making provisions".

The application of BAT is likely to already address risks to water, air, soil, plants or animals, odour nuisance and some aspects of effects on the countryside. It will, however, be necessary for the Operator to briefly to consider each of these objectives individually and provide a comment on how they are being addressed by the proposals. It is also necessary to ensure that any places of special concern which could be affected, such as SSSIs, are identified and commented upon although, again, these may have been addressed in the assessment for BAT, in which case a cross-reference may suffice.

Operators should identify any development plans made by the local planning authority, including any waste local plan, and comment on the extent to which the proposals accord with the contents of any such plan (see Section 2.6).

## 5.3 The Habitats Regulations



An application for an IPPC Permit will be regarded as a new plan or project for the purposes of the Habitats Regulations (for the equivalent Regulations in Scotland and Northern Ireland see Appendix 2). Therefore, Operators should provide an initial assessment of whether the installation is likely to have a significant effect on any European site in the UK (either alone or in combination with other relevant plans or projects) and, if so, an initial assessment of the implications of the installation for any such site. The application of BAT is likely to have gone some way towards addressing the potential impact of the installation on European sites and putting into place techniques to avoid any significant effects. The Operator should provide a description of how the BAT assessment has specifically taken these matters into account, bearing in mind the conservation objectives of any such site.

European sites are defined in Regulation 10 of the Habitats Regulations to include Special Areas of Conservation (SACs); sites of community importance (sites that have been selected as candidate SACs by member states and adopted by the European Commission but which are not yet formally classified); and Special Protection Areas (SPAs). It is also Government policy (set out in PPG 9 on nature conservation) that potential SPAs and candidate SACs should be considered to be European sites for the purposes of Regulation 10.

Information on the location of European Sites and their conservation objectives is available from

- English Nature (01733 455000), http://www.english-nature.org.uk
- Countryside Council for Wales (01248 385620), http://www.ccw.gov.uk
- Scottish Natural Heritage (0131 447 4784), http://www.snh.org.uk
- Joint Nature Conservation Committee (01733 866852), http://www.jncc.gov.uk
- Environment and Heritage Service, Northern Ireland, http://www.ehsni.gov.uk

The Regulator will need to consider the Operator's initial assessment. If it concludes that the installation is likely to have a significant effect on a European site, then the Regulator will need to carry out an "appropriate assessment" of the implications of the installation in view of that site's conservation objectives. The Regulations impose a duty on the Regulator to carry out these assessments so it cannot rely on the Operator's initial assessments. Therefore the Regulator must be provided with any relevant information upon which the Operator's assessment is based.

Note that in many cases the impact of the Habitats Regulations will have been considered at the planning application stage, in which case the Regulator should be advised of the details.

# REFERENCES

For a full list of available Technical Guidance see Appendix A of the *Guide for Applicants* or visit the Environment Agency Website http://www.environment-agency.gov.uk. Many of the references below are being made available free of charge for viewing or download on the Website. The same information can also be accessed via the SEPA web site http://www.sepa.org.uk, or the NIEHS web site www.ehsni.gov.uk. Most titles will also be available in hard copy from The Stationery Office (TSO). Some existing titles are not yet available on the Website but can be obtained from TSO.

- 1. IPPC Reference Document on Best Available Techniques in the Ferrous Foundry Industry European Commission http://eippcb.jrc.es FOUNDRY NOTE NOT YET PUBLISHED – DUE 2002
- 2. The Pollution Prevention and Control Act (1999) (www.legislation.hmso.gov.uk).
- 3. The Pollution Prevention and Control Regulations (SI 1973 2000) (www.legislation.hmso.gov.uk).
- 4. IPPC: A Practical Guide (for England and Wales) (or equivalents in Scotland and Northern Ireland) (www.environment.detr.gov.uk).
- IPPC Part A(1) Installations: Guide for Applicants (includes Preparation of a Site Report in a Permit Application) (EA Website).
- 6. Assessment methodologies:
- E1 BPEO Assessment Methodology for IPC
- IPPC Environmental Assessments for BAT (in preparation as H1)
- 7. Management system references:
- Sector specific NB: Reference to be included
- 8. Waste minimisation support references:
- Environment Agency web site. Waste minimisation information accessible via:
- http://www.environment-agency.gov.uk/subjects/waste/131528/
- Waste Minimisation an environmental good practice guide for industry (helps industry to minimise waste and achieve national environmental goals). Available free to companies who intend to undertake a waste reduction programme (tel 0345 33 77 00)
- Profiting from Pollution Prevention 3Es methodology (emissions, efficiency, economics). Video and A4 guide aimed at process industries. Available from Environment Agency, North East region (tel 0113 244 0191, ask for regional PIR)
- Waste Minimisation Interactive Tools (WIMIT). Produced in association with the ETBPP and the BOC Foundation (a software tool designed for small and medium businesses.). Available free from The Environmental Helpline (tel 0800 585794)
- Environmental Technology Best Practice Programme ETBPP. A joint DTI/DETR programme, with over 200 separate case studies, good practice guides, leaflets, flyers, software tools and videos covering 12 industry sectors, packaging, solvents and the generic areas of waste minimisation and cleaner technology. The ETBPP is accessible via a FREE and confidential helpline (tel 0800 585794) or via the web site www.etsu.com/etbpp/
- ETBPP, Increased Profit Through Improved Materials Additions: Management/Technical Guide, GG194/195
- Waste Management Information Bureau. The UK's national referral centre for help on the full range of waste management issues. It produces a database called Waste Info, which is available for online searching and on CD-ROM. Short enquiries are free (tel 01235 463162)
- Institution of Chemical Engineers Training Package E07 Waste Minimisation. Basic course which contains guide, video, slides, OHPs etc. (tel 01788 578214)
- 9. Water efficiency references:
- ETBPP, Simple measures restrict water costs, GC22
- ETBPP, Effluent costs eliminated by water treatment, GC24
- ETBPP, Saving money through waste minimisation: Reducing water use, GG26
- ETBPP Helpline 0800 585794
- Environment Agency (1998) Optimum use of water for industry and agriculture dependent on direct abstraction: Best practice manual. R&D technical report W157, WRc Dissemination Centre, Swindon (tel 01793 865012)
- 11. Releases to air references:
- BREF on Waste Water and Waste Gas Treatment.
- A1 Guidance on effective flaring in the gas, petroleum etc industries, 1993, ISBN 0-11-752916-8
- A2 Pollution abatement technology for the reduction of solvent vapour emissions, 1994, £5.00, ISBN 0-11-752925-7
- A3 Pollution abatement technology for particulate and trace gas removal, 1994, £5.00, ISBN 0-11-752983-4
- Landfill gas flaring
- Part B PG1/3 Boilers and Furnaces 20-50 MW net thermal input ISBN 0-11-753146-4-7
- Part B PG1/4 Gas Turbines 20-50 MW net thermal input ISBN 0-11-753147-2
- 12. Releases to water references:
  - BREF on Waste Water and Waste Gas Treatment
- A4 Effluent Treatment Techniques, TGN A4, Environment Agency, ISBN 0-11-310127-9 (EA website)
- Environment Agency, Pollution Prevention Guidance Note Above-ground oil storage tanks, PPG 2, gives information on tanks and bunding which have general relevance beyond just oil (EA website)

- Mason, P. A, Amies, H. J, Sangarapillai, G. Rose, Construction of bunds for oil storage tanks, Construction Industry Research and Information Association (CIRIA), Report 163, 1997, CIRIA, 6 Storey's Gate, Westminster, London SW1P 3AU. Abbreviated versions are also available for masonry and concrete bunds (www.ciria.org.uk online purchase)
- 13. Dispersion Methodology Guide D1 (EA website summary only)
- 14. IPPC Energy Efficiency Guidance Note (the consultation version, available on the website should be used until the final version is published)
- 15. BS 5908: Code of Practice for Fire Precautions in the Chemical and Allied Industries
- 16. Environment Agency, Pollution Prevention Guidance Note Pollution prevention measures for the control of spillages and fire-fighting run-off, PPG 18, gives information on sizing firewater containment systems (EA website)
- 17. Investigation of the criteria for, and guidance on, the landspreading of industrial wastes final report to the DETR, the Environment Agency and MAFF, May 1998
- 18. Agency guidance on the exemption 7 activity (proposed)
- 19. COMAH guides
- A Guide to the Control of Major Accident Hazards Regulations 1999, Health and Safety Executive (HSE) Books L111, 1999, ISBN 0 07176 1604 5
- Preparing Safety Reports: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)190, 1999
- Emergency Planning for Major Accidents: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)191, 1999
- Guidance on the Environmental Risk Assessment Aspects of COMAH Safety Reports, Environment Agency, 1999 (EA website)
- Guidance on the Interpretation of Major Accidents to the Environment for the Purposes of the COMAH Regulations, DETR, 1999, ISBN 753501 X, available from the Stationery Office
- 20. Assessment and Control of Environmental Noise and Vibration from Industrial Activities (joint Regulator's guidance in preparation)
- 21. Monitoring Guidance (EA website)
- M1 Sampling facility requirements for the monitoring of particulates in gaseous releases to atmosphere, March 1993, £5.00, ISBN 0-11-752777-7
- M2 Monitoring emissions of pollutants at source January 1994, £10.00, ISBN 0-11-752922-2
- M3 Standards for IPC Monitoring Part 1: Standards, organisations and the measurement infrastructure, August 1995, £11.00, ISBN 0-11-753133-2
- M4 Standards for IPC Monitoring Part 2 : Standards in support of IPC Monitoring, revised 1998
- MCERTS approved equipment link via http://www.environment-agency.gov.uk/epns "Guidance for Business and Industry";
- Direct Toxicity Assessment for Effluent Control: Technical Guidance (2000), UKWIR 00/TX/02/07.
- 22. The Categorisation of Volatile Organic Compounds, DOE Research Report No DOE/HMIP/RR/95/009 (EA website)
- 23. Odour Assessment and Control Guidance for Regulators and Industry (joint agencies guidance in preparation)
- 24. "Policy and Practice for the Protection of Groundwater" (PPPG) (EA website)
- 25. Working at Construction and Demolition-sites (PPG 6) (EA website)
- 26. The Environmental Technology Best Practice Programme, ETSU, Harwell, Oxfordshire OX11 0RA. Helpline 0800 585794 Good Practice Guides:
- GG104 Cost effective management of chemical binders in foundries.
- GG71 Cost-effective reduction of fugitive solvent emissions.
- GG119 Optimising sand use in foundries.
- 27. Beneficial Re-use for Managers. The Castings Development Centre. June 1998
- 28. A Review of the Industrial Uses of Continuous Monitoring Systems: Metals Industries Processes. Environment Agency Report No NCAS/TR/98/003 March 1998.

# DEFINITIONS

BAT	Best Available Techniques – see IPPC A Practical Guide or the Regulations for further definition
BAT Criteria	The criteria to be taken into account when assessing BAT, given in Schedule 2 of the PPC Regulations
BOD	Biological Oxygen Demand
BREF	BAT Reference Document
CEM	Continuous Emissions Monitoring
CHP	Combined heat and power plant
COD	Chemical Oxygen Demand
EMS	Environmental Management System
ETP	Effluent treatment plant
ITEQ	International Toxicity Equivalents
MCERTS	Monitoring Certification Scheme
NIEHS	Northern Ireland Environment and Heritage Service
SAC	Special Areas of Conservation
SECp	Specific Energy consumption
SEPA	Scottish Environment Protection Agency
SPA	Special Protection Area
TSS	Total Suspended Solids
TOC	Total Organic Carbon
VOC	Volatile organic compounds

# APPENDIX 1 - SOME COMMON MONITORING AND SAMPLING METHODS

Determinand	Method	Detection limit Uncertainty	Valid for range mg/l	Standard
Suspended solids	Filtration through glass fibre filters	1 mg/l 20%	10-40	ISO 11929:1997 EN872 Determination of suspended solids
COD	Oxidation with di- chromate	12 mg/l 20%	50-400	ISO 6060: 1989 Water Quality- Determination of chemical oxygen demand
BOD₅	Seeding with microorganisms and measurement of oxygen content	2 mg/l 20%	5-30	ISO 5815: 1989 Water Quality Determination of BOD after 5 days, dilution and seeding method <b>EN 1899</b> (BOD 2Parts)
AOX	Adsorption on activated carbon and combustion	 20%	0.4 – 1.0	ISO 9562: 1998 EN1485 – Determination of adsorbable organically bound halogens.
Tot P				BS 6068: Section 2.28 1997 Determination of phosphorus –ammonium molybdate spectrometric method
Tot N				BS 6068: Section 2.62 1998 – Determination of nitrogen Part 1 Method using oxidative digestion with peroxydisulphate BS EN ISO 11905
рН				SCA The measurement of electric conductivity and the determination of pH ISBN 0117514284
Turbidity				SCA Colour and turbidity of waters 1981 ISBN 0117519553 EN 27027:1999
Flow rate	Mechanical ultrasonic or electromagnetic gauges			SCA Estimation of Flow and Load ISBN 011752364X
Temperature				
тос				SCA The Instrumental Determination of Total Organic Carbon and Related Determinants 1995 ISNB 0117529796 <b>EN 1484:1997</b>
Fatty Acids				Determination of Volatile Fatty Acids in Sewage Sludge 1979 ISBN 0117514624
Metals				BS 6068: Section 2.60 1998 – Determination of 33 elements by inductively coupled plasma atomic emission spectroscopy
Chlorine				BS6068: Section 2.27 1990 – Method for the determination of total chlorine: iodometric titration method
Chloroform Bromoform				BS 6068: Section 2.58 Determination of highly volatile halogenetaed hydrocarbons – Gas chromatographic methods
Dispersants Surfactants Anionic Cationic Non-ionic				SCA Analysis of Surfactants in Waters, Wastewaters and Sludges ISBN 01176058 EN 903:1993 (Used for anionic surfactants)
Pentachloro- Phenol				BS5666 Part 6 1983 – Wood preservative and treated timber quantitative analysis of wood preservatives containing pentachlorophenol EN 12673:1997 (used for chlorophenol and polychlorinated phenols)
Formaldehyde				SCA The determination of formaldehyde, other volatile aldehydes and alcohols in water
Phosphates and Nitrates				BS 6068: Section 2.53 1997 Determination of dissolved ions by liquid chromatography
Sulphites and sulphates				BS 6068: Section 2.53 1997 Determination of dissolved ions by liquid chromatography
Ammonia				BS 6068: Section 2.11 1987 – Method for the determination of ammonium: automated spectrometric method
Grease and oils	IR absorption	0.06 mg/kg		SCA The determination of hydrocarbon oils in waters by solvent extraction IR absorption and gravimetry ISBN 011751 7283

#### Table A1.1: Measurement methods for common substances to water

#### APPENDIX 1 - MONITORING AND SAMPLING METHODS

Determinand	Method	Av'ging time Detection limit Uncertainty	Compliance criterion	Standard
Formaldehyde	Impingement In 2,4 dinitro-phenyl- Hydrazine HPLC	1 hour 1 mg/m <sup>3</sup> 30%	Two samples taken. Each result below limit after subtraction of measurement	NIOSH
Ammonia	Ion Chromato- graphy	1 hour 0.5mg/m <sup>3</sup> 25%	uncertainty	US EPA Method 26
VOCs Speciated	Adsorption Thermal Desorption GCMS	1 hour 0.1 mg/m <sup>3</sup> 30%		BS EN 1076:Workplace atmospheres. Pumped sorbent tubes for the determination of gases and vapours. Requirements and test methods.
Chloroform	Absorption on activated carbon solvent extraction. GC analysis	1 hour 1 mg/m <sup>3</sup> 20%		MDHS 28 Chlorinated hydrocarbon solvent vapours in air (modified)
Oxides of Sulphur	UV fluoresence automatic analyser	1 hour 1 ppm 10%	95% of hourly averages over a year below specified limit	ISO 7935 (BS6069 Section 4.4) Stationary source emissions- determination of mass concentrations of sulphur dioxide CEN Standard in preparation
	Wet sampling train Ion chromatography	1 hour 1 mg/m <sup>3</sup> 25%	Two samples taken. Each result below limit after subtraction of measurement uncertainty	ISO 7934 (BS6069 Section 4.1) Method for the determination of the mass concentration of sulphur dioxide- hydrogen peroxide/barium perchlorate method

#### Table A1.2: Measurement methods for air emissions

Measurement uncertainty is defined as total expanded uncertainty at 95% confidence limit calculated in accordance with the Guide to the Expression of Uncertainty in Measurement, ISBN 92-67-10188-9, 1<sup>st</sup> Ed., Geneva, Switzerland, ISO 1993.

See also Monitoring Guidance (Ref. 21).

# APPENDIX 2 - EQUIVALENT LEGISLATION IN SCOTLAND & NORTHERN IRELAND

The legislation referred to in the text is that for England and Wales. The following are the equivalents for Scotland and Northern Ireland.

Table A.2.1	-
Equivalent	
Legislation	

England and Wales	Scotland	Northern Ireland
PPC Regulations (England and Wales) 2000	PPC (Scotland) Regulations 2000; SI 200/323	
Waste Management Licensing Regulations SI:1994 1056	Waste Management Licensing Regulations SI:1994 1056	No NI equivalent
The Water Resources Act 1991	COPA 1974 (S30A-30E equiv to Part III WRA91) Natural Heritage (Scotland) Act 1991(Part II equiv to Part I WRA91)	The Water (NI) Order 1999
SI 1989 No 317: Clean Air, The Air Quality Standards Regulations 1989	SI 1989/317: Clean Air, The Air Quality Standards Regulations 1989	The Air Quality Standards Regulations (Northern Ireland) 1990. Statutory Rules of Northern Ireland 1990 No 145
SI 1997 No 3043: Environmental Protection, The Air Quality Regulations 1997	SSI 2000/97 The Air Quality (Scotland) Regs	No NI equivalent
SI 1989 No 2286 and 1998 No 389 the Surface Water (Dangerous Substances Classification) Regulations. (Values for List II substances are contained in SI 1997/2560 and SI 1998/389)	SI 1990/126 Surface Water (Dangerous Substances) (Classification) (Scotland) Regs	Surface Waters (Dangerous Substances) (Classification) Regulations 1998. Statutory Rules of Northern Ireland 1998 No 397 SI1991/1597:
SI 1991/1597: Bathing Waters (Classification) Regs.	SI 1991/1609 Bathing Waters (Classification) (Scotland) Regs	The Quality of Bathing Water Regulations (NI) 1993
SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regs.	SI 1997/2471 Surface Waters (Fishlife) (Classification) Regs	The Surface Water (Fishlife) (Classification) Regulations (NI) 1997
SI1997/1332 Surface Waters (Shellfish) (Classification) Regs.	SI 1997/2470 Surface Waters (Shellfish) (Classification) Regs	The Surface Water (Shellfish) (Classification) Regulations (NI) 1997
SI1994/2716 Conservation (Natural Habitats etc) Regulations 1994	SI 1994/2716 Conservation (Natural Habitats etc) Regs	Conservation (Natural Habitats etc) Regulations (Northern Ireland) 1995
Control of Major Accident Hazards Regulations 1999 (COMAH)	SI 1999/743 Control of Major Accident Hazards Regs	Control of Major Accident Hazard Regulations (Northern Ireland) 2000